SECURITIES AND EXCHANGE COMMISSION

FORM 10-Q

Quarterly report pursuant to sections 13 or 15(d)

Filing Date: 2011-05-12 | Period of Report: 2011-03-31 SEC Accession No. 0001193125-11-137036

(HTML Version on secdatabase.com)

FILER

NewPage Holdin	a CORP
----------------	--------

CIK:1329068| IRS No.: 050616158 | State of Incorp.:DE | Fiscal Year End: 1231

Type: 10-Q | Act: 34 | File No.: 001-32956 | Film No.: 11834279

SIC: 2621 Paper mills

Mailing Address MIAMISBURG OH 45342

Business Address 8540 GANDER CREEK DRIVE 8540 GANDER CREEK DRIVE MIAMISBURG OH 45342 (877) 855-7243

NewPage CORP

CIK:1329069| IRS No.: 050616156 | State of Incorp.:DE | Fiscal Year End: 1231

Type: 10-Q | Act: 34 | File No.: 333-125952 | Film No.: 11834278

SIC: 2621 Paper mills

Mailing Address MIAMISBURG OH 45342

Business Address 8540 GANDER CREEK DRIVE 8540 GANDER CREEK DRIVE MIAMISBURG OH 45342 (877) 855-7243

NewPage Corporation

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the Quarterly Period Ended March 31, 2011



8540 Gander Creek Drive Miamisburg, Ohio 45342 877.855.7243

		07.10			
Commission				IRS Employer	State of
File Number	R	l egistrant		Identification Number	Incorporation
001-32956	NEWPAGE HOL	DING CORPORA	ATION	05-0616158	Delaware
333-125952	NEWPAGE	CORPORATION	N	05-0616156	Delaware
Indicate by check mark wh	ether the registrant (1) h	as filed all reports	required to b	be filed by Section 13 or 15(d)	of the Securities Exchange
Act of 1934 during the pred	ceding 12 months (or for	such shorter perio	d that the re	gistrant was required to file suc	ch reports), and (2) has been
subject to such filing requir	rements for the past 90 d	ays.			
NewPage Holding Co	orporation	Yes ⊠	No □		
NewPage Corporation	1	Yes ⊠	No □		
Indicate by check mark who	ether the registrant has s	ubmitted electronic	cally and po	sted on its corporate Web site,	if any, every Interactive Data
File required to be submitted	ed and posted pursuant to	Rule 405 of Regu	ılation S-T (§232.405 of this chapter) during	ng the preceding 12 months
(or for such shorter period	that the registrant was re	quired to submit ar	nd post such	files).	
NewPage Holding Co	orporation	Yes □	No □		
NewPage Corporation	1	Yes □	No □		
	=	=		erated filer, a non-accelerated fismaller reporting company" in	
NewPage Holding Co	orporation	Large accelerate	ed filer □	Accelerated fil	er □
		Non-accelerated	d filer ⊠	Smaller reports	ing company □

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act).

Accelerated filer □

Smaller reporting company □

Large accelerated filer □

Non-accelerated filer ⊠

NewPage Holding Corporation	Yes □	No ⊠	
NewPage Corporation	Yes □	No ⊠	

There were 10 Common Shares, \$0.01 per share par value, of NewPage Holding Corporation and 100 Common Shares, \$0.01 per share par value, of NewPage Corporation outstanding as of May 1, 2011.

This Form 10-Q is a combined quarterly report being filed separately by two registrants: NewPage Holding Corporation and NewPage Corporation. NewPage Corporation meets the conditions set forth in general instruction H(1)(a) and (b) of Form 10-Q and is therefore filing this form with the reduced disclosure format.

References to "NewPage Holding" refer to NewPage Holding Corporation, a Delaware corporation; references to "NewPage" refer to NewPage Corporation, a Delaware corporation and a wholly-owned subsidiary of NewPage Holding. References to "NewPage Group" refer to NewPage Group Inc., a Delaware corporation and the direct parent of NewPage Holding. Unless the context provides otherwise, references to "we," "us" and "our" refer to NewPage Holding and its subsidiaries. All assets, liabilities, income, expenses and cash flows presented for all periods represent those of NewPage Holding's wholly-owned subsidiary, NewPage, except for activity related to NewPage Holding's debt and equity and income tax effects. Unless otherwise noted, the information provided pertains to both NewPage Holding and NewPage.

FORWARD-LOOKING STATEMENTS

This quarterly report contains "forward-looking statements" within the meaning of the Securities Exchange Act of 1934, the Private Securities Litigation Reform Act of 1995 and other related laws. All statements other than statements of historical fact are "forward-looking statements" for purposes of federal and state securities laws. Forward-looking statements may include the words "may," "plans," "estimates," "anticipates," "believes," "expects," "intends" and similar expressions. Although we believe that these forward-looking statements are based on reasonable assumptions, they are subject to numerous factors, risks and uncertainties that could cause actual outcomes and results to be materially different from those projected or assumed in our forward-looking statements. These factors, risks and uncertainties include, among others, the following:

our substantial level of indebtedness

our ability to obtain additional financing or refinance our indebtedness may be limited

changes in the supply of, demand for, or prices of our products

general economic and business conditions in the United States and Canada and elsewhere

the ability of our customers to continue as a going concern, including our ability to collect accounts receivable according to customary business terms

the activities of competitors, including those that may be engaged in unfair trade practices

changes in significant operating expenses, including raw material and energy costs

changes in currency exchange rates

changes in the availability of capital

changes in the regulatory environment, including requirements for enhanced environmental compliance

the other factors described under "Risk Factors" in our Annual Report on Form 10-K for the year ended December 31, 2010

Given these risks and uncertainties, we caution you not to place undue reliance on forward-looking statements. We undertake no obligation to update or revise any forward-looking statements, either to reflect new developments or for any other reason, except as required by law.

INDEX

		Pag
PART I	FINANCIAL INFORMATION	
Item 1.	Financial Statements:	
	Condensed Consolidated Balance Sheets as of March 31, 2011 and December 31, 2010	3
	Consolidated Statements of Operations for the first quarter ended March 31, 2011 and 2010	4
	Consolidated Statements of Equity (Deficit) for the first quarter ended March 31, 2011 and 2010	5
	Condensed Consolidated Statements of Cash Flows for the first quarter ended March 31, 2011 and 2010	7
	Notes to Condensed Consolidated Financial Statements	8
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	22
Item 3.	Quantitative and Qualitative Disclosures about Market Risk	28
Item 4.	Controls and Procedures	29
PART II	OTHER INFORMATION	
Item 1.	<u>Legal Proceedings</u>	29
Item 5.	Other Information	30
Item 6.	<u>Exhibits</u>	31
<u>Signatures</u>		32

PART I FINANCIAL INFORMATION

ITEM 1. FINANCIAL STATEMENTS

CONDENSED CONSOLIDATED BALANCE SHEETS (unaudited) MARCH 31, 2011 AND DECEMBER 31, 2010

Dollars in millions, except per share amounts

	NewPage	Holding	New	Page
	Mar. 31,	Dec. 31,	Mar. 31,	Dec. 31,
	2011	2010	2011	2010
ASSETS				
Cash and cash equivalents	\$9	\$8	\$9	\$8
Accounts receivable, net	294	292	294	292
Inventories (Note 3)	551	523	551	523
Other current assets	19	20	19	20
Total current assets	873	843	873	843
Property, plant and equipment, net of accumulated depreciation of \$1,210 as of March 31, 2011				
and \$1,159 as of December 31, 2010	2,509	2,558	2,509	2,558
Port Hawkesbury biomass project (Note 7)	19	5	19	5
Other assets	102	106	101	105
TOTAL ASSETS	\$3,503	\$3,512	\$3,502	\$3,511
LIABILITIES AND EQUITY (DEFICIT)				
Accounts payable	\$204	\$195	\$204	\$195
Other current liabilities	264	210	264	210
Current maturities of long-term debt (Note 6)	1,783		1,783	
Total current liabilities	2,251	405	2,251	405
Long-term debt (Note 6)	1,604	3,376	1,380	3,157
Proceeds from NSPI for Port Hawkesbury biomass project (Note 7)	89	80	89	80
Other long-term obligations	528	526	528	526
Commitments and contingencies (Note 14)				
EQUITY (DEFICIT)				
Common stock, NewPage Holding-10 shares authorized, issued and outstanding, \$0.01 per				
share par value; NewPage-100 shares authorized, issued and outstanding, \$0.01 per share				
par value	-	_	-	-
Additional paid-in capital	707	707	813	813
Accumulated deficit	(1,373)	(1,280)	(1,266)	(1,178)
Accumulated other comprehensive loss	(303)	(302)	(293)	(292)
Total equity (deficit)	(969)	(875)	(746)	(657)
TOTAL LIABILITIES AND EQUITY (DEFICIT)	\$3,503	\$3,512	\$3,502	\$3,511

CONSOLIDATED STATEMENTS OF OPERATIONS (unaudited) FIRST QUARTER ENDED MARCH 31, 2011 AND 2010

Dollars in millions

	NewPage	Holding	NewPage First Quarter		
	First C	Quarter			
	Ended l	Mar. 31,	Ended Mar. 3		
	2011	2010	2011	2010	
Net sales	\$904	\$817	\$904	\$817	
Cost of sales	858	849	858	849	
Selling, general and administrative expenses	40	49	40	49	
Interest expense (including non-cash interest expense of \$17, \$16, \$12 and \$12)	100	101	95	97	
Other (income) expense, net (Note 10)	(2)	(3)	(2)	(3)	
Income (loss) before income taxes	(92)	(179)	(87)	(175)	
Income tax (benefit)	1	_	1		
Net income (loss)	\$(93)	\$(179)	\$(88)	\$(175)	

NEWPAGE HOLDING CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF EQUITY (DEFICIT) (unaudited) FIRST QUARTER ENDED MARCH 31, 2011

Dollars in millions

						Accumulat	ed	
	Comm	on Stock				Other		
	Comm	on Stock	Additional			Comprehens	sive	
			Paid-in	Accumulat	ed	Income		
	Shares	Amount	Capital	Deficit		(Loss)		Total
Balance at December 31, 2010	10	\$ -	\$ 707	\$ (1,280)	\$ (302)	\$(875)
Comprehensive income (loss):								
Net income (loss)				(93)			(93)
Defined-benefit postretirement plans:								
Amortization of net actuarial loss on defined benefit								
plans						(1)	(1)
Cash-flow hedges:								
Reclassification adjustment to net income (loss)						1		1
Foreign currency translation adjustment						(1)	(1)
Comprehensive income (loss)								\$(94)
Balance at March 31, 2011	10	\$ -	\$ 707	\$ (1,373)	\$ (303)	\$(969)

FIRST QUARTER ENDED MARCH 31, 2010

Dollars in millions

	Comm	on Stock	Additional Paid-in	Accumulat	ted	Accumula Other Comprehen Income	isive	
	Shares	Amount	Capital	Deficit	····	(Loss)		Total
Balance at December 31, 2009	10	\$ -	\$ 685	\$ (606)	\$ (265)	\$(186)
Comprehensive income (loss):								
Net income (loss)				(179)			(179)
Amortization of net actuarial loss on defined benefit								
plans						(1)	(1)
Cash-flow hedges:								
Change in unrealized gains (losses)						(2)	(2)
Reclassification adjustment to net income								
(loss)						1		1
Foreign currency translation adjustment						1		1
Comprehensive income (loss)								\$(180)
Equity awards (Note 8)			8					8
Balance at March 31, 2010	10	\$ -	\$ 693	\$ (785)	\$ (266)	\$(358)

NEWPAGE CORPORATION AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF EQUITY (DEFICIT) (unaudited) FIRST QUARTER ENDED MARCH 31, 2011

Dollars in millions

						Accumulate	ed	
	Comm	on Stock				Other		
	Comm	on Stock	Additional			Comprehens	sive	
			Paid-in	Accumulate	ed	Income		
	Shares	Amount	Capital	Deficit		(Loss)		Total
Balance at December 31, 2010	100	\$ -	\$ 813	\$ (1,178)	\$ (292)	\$(657)
Comprehensive income (loss):								
Net income (loss)				(88))			(88)
Defined-benefit postretirement plans:								
Amortization of net actuarial loss on defined benefit								
plans						(1)	(1)
Cash-flow hedges:								
Reclassification adjustment to net income (loss)						1		1
Foreign currency translation adjustment						(1)	(1)
Comprehensive income (loss)								\$(89)
Balance at March 31, 2011	100	\$ -	\$ 813	\$ (1,266)	\$ (293)	\$(746)

FIRST QUARTER ENDED MARCH 31, 2010

Dollars in millions

	Comm	on Stock Amount	Additional Paid-in Capital	Accumula Deficit		Accumulate Other Comprehen Income (Loss)	sive	Total
Balance at December 31, 2009	100	\$ -	\$ 791	\$ (522)	\$ (255)	\$14
Comprehensive income (loss):								
Net income (loss)				(175)			(175)
Amortization of net actuarial loss on defined benefit plans						(1)	(1)
Cash-flow hedges:								
Change in unrealized gains (losses)						(2)	(2)
Reclassification adjustment to net income (loss)						1		1
Foreign currency translation adjustment						1		1
Comprehensive income (loss)								\$(176)
Equity awards (Note 8)			8					8
Balance at March 31, 2010	100	\$ -	\$ 799	\$ (697)	\$ (256)	<u>\$(154)</u>

CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (unaudited) FIRST QUARTER ENDED MARCH 31, 2011 AND 2010

Dollars in millions

	NewPage Holding		NewPage		
	First Q	uarter	First Q	uarter	
	Ended I	Mar. 31,	Ended Mar. 31,		
	2011	2010	2011	2010	
CASH FLOWS FROM OPERATING ACTIVITIES					
Net income (loss)	\$(93)	\$(179)	\$(88)	\$(175)	
Adjustments to reconcile net income (loss) to net cash provided by (used for) operating activities:					
Depreciation and amortization	62	68	62	68	
Non-cash interest expense	17	16	12	12	
(Gain) loss on disposal and impairment of assets	1	6	1	6	
Non-cash U.S. pension expense	6	9	6	9	
Equity award expense (Note 8)	-	8	-	8	
Change in operating assets and liabilities	27	68	27	68	
Net cash provided by (used for) operating activities	20	(4)	20	(4)	
CASH FLOWS FROM INVESTING ACTIVITIES					
Capital expenditures	(14)	(11)	(14)	(11)	
Proceeds from sales of assets	1	11	1	11	
Expenditures for Port Hawkesbury biomass project (Note 7)	(10)		(10)		
Net cash provided by (used for) investing activities	(23)	_	(23)	-	
CASH FLOWS FROM FINANCING ACTIVITIES					
Proceeds from issuance of long-term debt	-	67	-	67	
Payment of financing costs	(1)	(4)	(1)	(4)	
Borrowings on revolving credit facility	244	124	244	124	
Payments on revolving credit facility	(242)	(176)	(242)	(176)	
Proceeds from NSPI for Port Hawkesbury biomass project (Note 7)	2		2		
Net cash provided by (used for) financing activities	3	11	3	11	
Effect of exchange rate changes on cash and cash equivalents	1	1	1	1	
Net increase (decrease) in cash and cash equivalents	1	8	1	8	
Cash and cash equivalents at beginning of period	8	5	8	5	
Cash and cash equivalents at end of period	\$9	\$13	\$9	\$13	
SUPPLEMENTAL INFORMATION					
Cash paid for interest	\$10	\$13	\$10	\$13	

NEWPAGE HOLDING CORPORATION AND SUBSIDIARIES NEWPAGE CORPORATION AND SUBSIDIARIES NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (unaudited)

Dollars in millions, except per share amounts

1. BASIS OF PRESENTATION

NewPage Holding Corporation ("NewPage Holding") is a holding company that owns all of the outstanding capital stock of NewPage Corporation. NewPage Corporation and its subsidiaries are engaged in manufacturing, marketing and distributing printing papers used primarily for commercial printing, magazines, catalogs, textbooks and labels. Our products include coated, uncoated, supercalendered, newsprint and specialty papers and market pulp. Our products are manufactured at multiple mills in the United States and one mill in Canada and are supported by multiple distribution and converting locations. We operate within one operating segment. The condensed consolidated financial statements include the accounts of NewPage Holding and all entities it controls. All intercompany transactions and balances have been eliminated.

Unless the context provides otherwise, the terms "we," "our" and "us" refer to NewPage Holding and its consolidated subsidiaries, including NewPage Corporation, a separate public-reporting company. Unless otherwise noted, "NewPage" refers to NewPage Corporation and its consolidated subsidiaries. Other than NewPage Holding's debt obligation and related financing costs, interest expense and income tax effects, all other assets, liabilities, income, expenses, and cash flows presented for all periods represent those of its wholly-owned subsidiary, NewPage. Unless otherwise noted, the information provided pertains to both NewPage Holding and NewPage.

These interim condensed consolidated financial statements have not been audited. However, in the opinion of management, these financial statements include all normal recurring adjustments necessary for a fair statement in accordance with accounting principles generally accepted in the United States of America ("U.S. GAAP"). Certain information and footnote disclosures normally included in annual financial statements presented in accordance with U.S. GAAP have been condensed or omitted. The results of operations and cash flows for the interim periods presented are not necessarily indicative of the results to be expected for the full year. The accompanying financial statements should be read in conjunction with the annual financial statements and notes thereto included in our Annual Report on Form 10-K for the year ended December 31, 2010.

2. FINANCIAL INSTRUMENTS

Derivative Financial Instruments

We periodically use derivative financial instruments as part of our overall strategy to manage exposure to market risks associated with foreign currency exchange rate and natural gas price fluctuations. We do not hold or issue derivative financial instruments for trading purposes. We regularly monitor the credit-worthiness of the counterparties to our derivative instruments in order to manage our credit risk exposures under these agreements. Our risk of loss in the event of nonperformance by any counterparty under derivative financial instrument agreements is not considered material by management. These derivative instruments are measured at fair value and are classified as other assets or other long-term obligations on our balance sheets depending on the fair value of the instrument.

If the derivative is designated as a fair-value hedge, the changes in the fair value of the derivative and of the hedged item attributable to the hedged risk are recognized in earnings. If the derivative is designated as a cash-flow hedge, the effective portion of the change in the fair value of the derivative is recorded in

accumulated other comprehensive income (loss) and is recognized in the consolidated statements of operations when the hedged item affects earnings. Ineffective portions of changes in the fair value of cash-flow hedges and financial instruments not designated as hedges are recognized in earnings.

Interest Rates

We measure the fair values of our interest rate swaps using observable interest-rate yield curves for comparable assets and liabilities at commonly quoted intervals. As of March 31, 2011, we had no interest rate swaps outstanding. During the first quarter of 2010, we recognized in interest expense a loss of \$4 for interest rate swaps that did not qualify for hedge accounting. We paid cash of \$7 on our interest rate agreements for the first quarter ended March 31, 2010. Also, during the first quarter of 2010, we locked in the floating rate components of the remaining interest rate swaps and subsequently settled the remaining liability in the third quarter of 2010.

Natural Gas

In order to hedge the future cost of natural gas consumed at our mills, we engage in financial hedging of future gas purchase prices, designated as cash flow hedges. We hedge with financial instruments that are priced based on New York Mercantile Exchange natural gas futures contracts. We do not hedge basis (the effect of varying delivery points or locations) or transportation costs (the cost to transport the gas from the delivery point to a company location) under these transactions. As of March 31, 2011 and December 31, 2010, we were party to natural gas futures contracts for notional amounts aggregating 560,000 and 800,000 MMBTUs, which expire through October 2011. We measure the fair values of our natural gas contracts based on natural gas futures contracts priced on the New York Mercantile Exchange.

Assets and Liabilities Measured at Fair Value on a Recurring Basis

As of March 31, 2011 and December 31, 2010, the fair values and carrying amounts of our financial assets and liabilities measured on a recurring basis are as follows:

	Mar. 31,	Dec. 31,
Significant other observable inputs (Level 2)	2011	2010
Qualifying as hedges-		
Other long-term liabilities-natural gas contracts	\$ (1)	\$ (2)

The amount of gain (loss) on cash flow hedges recognized in accumulated other comprehensive income (loss) ("AOCI") and the amount reclassified to income (loss) during the first quarter ended March 31, 2011 and 2010 are as follows:

			Amount of
			gain (loss)
	Amount of	Location of gain	reclassified
	gain (loss)	(loss) reclassified	from AOCI
	recognized	from AOCI to	to income
Derivative Type	in OCI	income (loss)	(loss)
First Quarter Ended March 31, 2011			
Natural gas contracts	\$ -	Cost of sales	\$ (1)
First Quarter Ended March 31, 2010			
Natural gas contracts	(2)	Cost of sales	(1)

Assets and Liabilities Not Carried at Fair Value

The fair value of long-term debt is based upon quoted market prices for the same or similar issues or on the current interest rates available to us for debt of similar terms and maturities. At March 31, 2011 and December 31, 2010, the carrying amounts of all other assets and liabilities that qualify as financial instruments approximated their fair value. Details of our long-term debt are as follows:

	March 3	March 31, 2011		December 31, 2010	
	Fair	Carrying	Fair	Carrying	
	Value	Amount	Value	Amount	
Long-term debt:					
NewPage Holding	\$(2,668)	\$(3,236)	\$(2,456)	\$(3,224)	
NewPage	(2,636)	(3,012)	(2,425)	(3,005)	

Other Fair Value Disclosures

See Note 4 for fair value information of assets held for sale recorded at fair value.

3. INVENTORIES

Inventories as of March 31, 2011 and December 31, 2010 consist of:

	Mar. 31,	Dec. 31,
	2011	2010
Finished and in-process goods	\$ 306	\$ 298
Raw materials	110	91
Stores and supplies	135	134
	\$ 551	\$ 523

If inventories had been valued at current costs, they would have been valued at \$540 and \$496 at March 31, 2011 and December 31, 2010.

4. PROPERTY, PLANT AND EQUIPMENT

Property, plant and equipment includes approximately \$43 in net book value of assets held for sale as of March 31, 2011. The assets held for sale are classified as Level 3 for the use of inputs in determining fair value.

5. ACCOUNTS PAYABLE

Accounts payable as of March 31, 2011 and December 31, 2010 includes \$11 and \$31 of outstanding checks in excess of cash.

6. LONG-TERM DEBT

The balances of long-term debt as of March 31, 2011 and December 31, 2010 are as follows:

	Mar. 31, 2011	Dec. 31, 2010
NewPage:	2011	2010
Revolving credit facility	\$95	\$92
11.375% first-lien senior secured notes (face amount \$1,770)	1,688	1,684
Floating rate second-lien senior secured notes (LIBOR plus 6.25%)	225	225
10% second-lien senior secured notes (face amount \$806)	805	805
12% senior subordinated notes (face amount \$200)	199	199
Capital lease	151	152
Total long-term debt, including current portion	3,163	3,157
Current portion of long-term debt	1,783	_
Subtotal	1,380	3,157
NewPage Holding-		
Senior unsecured NewPage Holding PIK Notes (face amount \$228 and \$224;		
LIBOR plus 7.00%)	224	219
Long-term debt	\$1,604	\$3,376

In January 2011, we amended our revolving credit facility to extend its termination date. For those lenders who accepted this amendment (aggregating to \$470), the revolving credit facility matures on December 21, 2012, unless we do not repay or refinance our floating rate and 10% second-lien senior secured notes ("Second-Lien Notes") by December 2, 2011, in which case the revolving credit facility matures on March 1, 2012. For the remaining \$30 of commitment, the revolving credit facility matures on December 21, 2012 unless we do not repay or refinance our Second-Lien Notes by July 4, 2011, in which case the \$30 portion of the revolving credit facility matures on October 3, 2011.

The 11.375% senior secured first-lien notes (the "First-Lien Notes") mature on the earlier of (i) December 31, 2014 or (ii) the date that is 31 days prior to the maturity date of the Second-Lien Notes, the senior subordinated notes, the NewPage Holding PIK notes or any refinancing thereof. The Second-Lien Notes mature on May 1, 2012. However, if we do not repay or refinance our Second-Lien Notes by January 31, 2012, the First-Lien Notes would mature on March 31, 2012. The senior subordinated notes mature on May 1, 2013. The NewPage Holding PIK Notes mature on November 1, 2013.

As of March 31, 2011, our current liabilities exceeded our current assets by \$1,378. This deficit is driven by the classification of our First-Lien Notes and our revolving credit facility as current liabilities. Absent a refinancing of our Second-Lien Notes, an additional \$1,030 will become current during the second quarter of 2011. We are exploring various alternatives to address our ongoing capital needs. However, we cannot assure you that we will be able to refinance any of our indebtedness, or that we will be able to do so on commercially reasonable terms. Nor can we assure you that these measures will be sufficient to satisfy our scheduled debt service obligations, or that we will be able to fund our business and operations without additional capital. If we are unable to generate sufficient cash flow or refinance our debt, including our revolving credit facility which is a principal source of our liquidity, prior to the scheduled maturities, it would have a material adverse effect on our financial condition, the value of the outstanding debt and our ability to make required cash payments under our indebtedness.

7. PORT HAWKESBURY BIOMASS PROJECT

In November 2010, NewPage Port Hawkesbury Corp. ("NPPH"), an indirect wholly-owned subsidiary of NewPage, completed the sale of certain assets, including a boiler and land at the Port Hawkesbury, Nova Scotia mill, to Nova Scotia Power Inc. ("NSPI") for a cash sales price of \$79. The assets and proceeds of this sale will remain on our books until the completion of the construction phase (proceeds will be shown as a long-term liability).

In addition, NSPI and NPPH entered into a construction agreement for NPPH to construct for NSPI a biomass cogeneration utility plant for the generation of electricity by December 31, 2012 for approximately C\$93. All costs of construction up to C\$93 will be paid or reimbursed by NSPI. As part of the construction agreement, NPPH is subject to damages for failure to complete certain milestones on time, subject to limitations as provided in the agreement. In addition, NewPage issued a performance guarantee to NSPI, subject to a limitation of C\$29, if NPPH defaults on the agreement. NewPage also issued a letter of credit of \$10 to NSPI in November 2010 and an additional letter of credit of \$5 in January 2011 in accordance with the terms of the construction agreement. All construction costs incurred, both directly and indirectly, will become other assets and all reimbursements and direct payments by NSPI will become long-term liabilities, pending completion of the construction. Upon completion of the construction phase, all the previously described assets and liabilities will be de-recognized from the balance sheet and recognized as (gain) loss on sale of assets in the statement of operations.

NSPI and NPPH also entered into a management, operations and maintenance agreement related to NPPH operating the utility assets for and under the control of NSPI for an initial period of 25 years plus the period prior to completion of the turbine, with three automatic five-year renewal terms, unless notice of termination is given. In exchange for these services, NPPH will receive payment from NSPI based upon the amount of energy produced and NPPH will receive a portion of the steam generated from the utility plant for its needs. NPPH is subject to liquidated damages in the event that a minimum level of output is not achieved. In addition, NewPage issued a performance guarantee to NSPI, subject to a limitation of C\$15, if NPPH defaults on the agreement.

8. EQUITY

Included in selling, general and administrative expenses for the first quarter ended March 31, 2011 and 2010 is equity award expense of zero and \$8. In February 2010, the compensation committee amended the outstanding stock option awards of employees and directors to change the exercise price to \$2.00 per share and to vest the unearned performance-based options for 2008 and 2009. Included in the amount of equity award expense for 2010 is \$6 to reflect the effects of the modification on the vested time-based options and the vesting of the prior-year unearned performance-based options.

The following table summarizes activity in the plan:

		Weighted-
		average
		exercise
Shares of NewPage Group issuable under stock options, in thousands	Options	price
Outstanding at December 31, 2010	3,250	\$ 1.95
Granted	379	2.00
Outstanding at March 31, 2011	3,629	1.95
Exercisable at March 31, 2011	2,097	2.00

The outstanding options and the exercisable options at March 31, 2011 have a weighted-average remaining contractual life of 7.1 years.

We utilize a Black-Scholes pricing model to determine the fair value of options granted. The determination of the fair value of stock-based payment awards on the date of grant using an option-pricing model is affected by the stock price as well as assumptions regarding a number of other variables. These variables include expected stock price volatility over the term of the awards and projected employee stock option exercise behaviors (term of option). We estimate the expected term of options granted by incorporating the contractual term of the options and employees' expected exercise behaviors. We estimate the volatility of NewPage Group's common stock by considering volatility of appropriate peer companies and adjusting for factors unique to our stock, including the effect of debt leverage.

Assumptions used to determine the fair value of option grants are as follows:

	First Q	First Quarter	
	Ended Ma	arch 31,	
	2011	2010	
Weighted-average fair value of options granted	\$0.38	\$2.41	
Weighted-average assumptions used for grants:			
Expected volatility	90 %	90 %	
Risk-free interest rate	2.4 %	2.5 %	
Expected life of option (in years)	5	4	

9. RETIREMENT AND OTHER POSTRETIREMENT BENEFITS

A summary of the components of net periodic costs for the first quarter ended March 31, 2011 and 2010 is as follows:

Pension Plans

	U.S.	Plans	Canadian Plans First Quarter	
	First C	uarter		
	Ended I	Mar. 31,	1, Ended Mar. 31	
	2011	2010	2011	2010
Service cost	\$4	\$6	\$ 1	\$ 1
Interest cost	16	16	5	5
Expected return on plan assets	(17)	(16)	(5)	(5)
Amortization of net loss	_3	3	1	1
Net periodic cost	<u>\$6</u>	\$9	\$ 2	\$ 2

Other Postretirement Plans

	U.S. I	U.S. Plans First Quarter Ended Mar. 31,		Canadian Plans	
	First Q			Quarter	
	Ended M			Mar. 31,	
	2011	2010	2011	2010	
Service cost	\$ 1	\$ -	\$ -	\$ -	
Interest cost	1	2	_	_	
Amortization of net prior service cost (credit)	(3)	(3)			
Net periodic cost (income)	<u>\$ (1)</u>	<u>\$ (1)</u>	<u>\$ - </u>	<u>\$</u>	

10. OTHER (INCOME) EXPENSE

Other (income) expense during the first quarter ended March 31, 2011 and 2010 was \$(2) and \$(3). During the first quarter ended March 31, 2010, we recognized \$11 of net loss on disposal and impairment of assets in other (income) expense. In addition, we recognized \$(13) of (income) for alternative fuel mixture tax credits during the first quarter ended March 31, 2010, as income recognition criteria was met during the quarter.

11. INCOME TAXES

For the first quarter ended March 31, 2011 and 2010, we recorded a valuation allowance against our net deferred income tax benefit for federal income taxes and for certain states as it was more likely than not that we would not realize those benefits.

12. RESTRUCTURING

Whiting Mill Closure

During the fourth quarter of 2010, we announced the permanent closure of the Whiting, Wisconsin mill by the end of February 2011. Approximately 360 employees at the facility were affected. The decision to close the facility was made in order to better align capacity with market demand. As a result of this action, including the evaluation of capacity needs, during the fourth quarter of 2010, we recorded asset impairment charges for the Whiting mill and the previously-closed Kimberly mill of \$196 in other (income) expense and classified these assets as held for sale. During the fourth quarter of 2010 and first quarter of 2011, we recognized \$2 and \$5 of employee-related costs recorded in cost of sales. In addition, during the first quarter of 2011, we recognized \$2 in cost of sales for contract termination and other closure related costs and we expect to recognize additional charges in cost of sales of approximately \$15 to \$20 for contract termination and other closure related costs during 2011 and subsequent years as contract terms expire. The facility closure and related activities are expected to be substantially completed during the first half of 2011. The severance and other closure cost require the outlay of cash while the asset impairment charge represents a non-cash charge. We expect no material effect on future earnings or cash flows as a result of the Whiting mill closure.

The activity in the accrued restructuring liability relating to these actions for the first quarter ended March 31, 2011 was as follows:

	Closure	Employee
	Costs	Costs
Balance accrued at December 31, 2010	\$ -	\$ 2
Current charges	2	5
Payments	(2)	(2)
Balance accrued at March 31, 2011	<u>\$</u>	\$ 5

13. DISPOSITION OF ASSETS

In March 2011, Rumford Paper Company ("Rumford"), a wholly-owned subsidiary of NewPage Corporation, and Rumford Cogeneration Company Limited Partnership, a subsidiary of Rumford, both indirect subsidiaries of NewPage Holding Corporation, entered into an Asset Sale Agreement for the sale of Rumford's cogeneration energy assets to ReEnergy Rumford LLC ("ReEnergy") for a sale price of \$61, payable one-half in cash at closing and one-half in the form of a 10-year note receivable. The transaction also contemplates an ongoing contractual relationship between the parties, pursuant to which Rumford will supply biomass fuel to ReEnergy and ReEnergy will supply electricity and thermal energy to Rumford. The transaction is not expected to have a material effect on the operating costs of the Rumford mill. We plan to invest the net cash proceeds in our business or use the proceeds to repay indebtedness, or both, in accordance with the terms of our debt instruments. The transaction, which is expected to close during the second or third quarter of 2011, is subject to customary conditions, including the receipt of regulatory approvals and the finalization of agreements outlined in term sheets.

In February 2010, Consolidated Water Power Company ("CWPCo"), an indirect wholly-owned subsidiary of NewPage, announced that it signed an agreement with Wisconsin Rapids Water Works Lighting Commission to sell the CWPCo utility transmission and distribution assets ("CWPCo Utility"), as amended. The purchase price will be equal to the net book value of the assets at the time of closing, subject to adjustment in the agreement. The transaction remains subject to regulatory approvals. In November 2010, CWPCo entered into an asset sale agreement to sell five hydroelectric projects to Great Lakes Utilities for a net cash sales price of approximately \$70. The closing of the transaction is subject to completion of the CWPCo Utility sale and regulatory approval. These transactions are expected to close in 2011. We continue to hold and operate the assets in the normal course of our operations.

14. COMMITMENTS AND CONTINGENCIES

In a notice dated January 25, 2011, staff at the Federal Energy Regulatory Commission, or FERC, reported that they had preliminarily determined that our subsidiary, Rumford Paper Company, or Rumford, violated FERC regulations in connection with Rumford's participation in ISO-New England's Day Ahead Load Response Program from July 2007 through February 2008. Under the program, participants were paid to reduce their electrical load on the New England power grid from their "baseline" load levels. FERC asserts that Rumford improperly established and maintained an artificially high baseline, allowing it to receive increased payments under the program. Rumford cooperated with FERC throughout its investigation and has denied any wrongdoing.

Claims have been made against us for the costs of environmental remediation measures taken or to be taken. We are also involved in various other litigation and administrative proceedings arising in the normal course of business. Reserves for these liabilities have been established and no insurance recoveries have been anticipated in the determination of the reserves. Although the ultimate outcome of

these matters cannot be predicted with certainty, we do not believe that the currently expected outcome of any matter, lawsuit or claim that is pending or threatened, or all of them combined, will have a material adverse effect on our financial condition, results of operations or liquidity.

15. SUPPLEMENTAL CONSOLIDATING INFORMATION

NewPage has issued \$1,770 face amount of First Lien Notes, \$1,031 face amount of Second Lien Notes and \$200 face amount of 12% senior subordinated notes. These notes are jointly and severally guaranteed on a full and unconditional basis by NewPage's 100%-owned subsidiaries, except CWPCo and Rumford Cogeneration Company Limited Partnership, our non-guarantor subsidiaries.

The following condensed consolidating financial statements have been prepared from financial information on the same basis of accounting as the consolidated financial statements. Investments in our subsidiaries are accounted for under the equity method.

NEWPAGE CORPORATION SUPPLEMENTAL CONDENSED CONSOLIDATING BALANCE SHEET MARCH 31, 2011

			Non-		
	NewPage	Guarantor	Guarantor	Consolidation/	
	Corporation	Subsidiaries	Subsidiaries	Eliminations	Consolidated
ASSETS					
Cash and cash equivalents	\$ 1	\$ -	\$ 8	\$ -	\$ 9
Accounts receivable	236	57	1	-	294
Inventories	229	322	_	_	551
Other current assets	12	6	<u> </u>		19
Total current assets	478	385	10	-	873
Intercompany receivables	1,272	552	179	(2,003)	_
Property, plant and equipment, net	25	2,437	47	_	2,509
Investment in subsidiaries	1,562	59	_	(1,621)	_
Port Hawkesbury biomass project	_	19	_	_	19
Other assets	94	6	1		101
TOTAL ASSETS	\$ 3,431	\$ 3,458	\$ 237	\$ (3,624)	\$ 3,502
LIABILITIES AND EQUITY (DEFICIT)					
Accounts payable	\$ 41	\$ 162	\$ 1	\$ -	\$ 204
Other current liabilities	164	96	4	_	264
Current maturities of long-term debt	1,783				1,783
Total current liabilities	1,988	258	5	-	2,251
Intercompany payables	544	1,296	163	(2,003)	_
Long-term debt	1,229	151	-	_	1,380
Proceeds from NSPI for Port Hawkesbury biomass					
project	_	89	_	_	89
Other long-term liabilities	416	102	10	_	528
Equity (deficit)	(746)	1,562	59	(1,621)	(746)
TOTAL LIABILITIES AND EQUITY (DEFICIT)	\$ 3,431	\$ 3,458	\$ 237	\$ (3,624)	\$ 3,502

NEWPAGE CORPORATION SUPPLEMENTAL CONDENSED CONSOLIDATING BALANCE SHEET DECEMBER 31, 2010

			Non-		
	NewPage	Guarantor	Guarantor	Consolidation/	
	Corporation	Subsidiaries	Subsidiaries	Eliminations	Consolidated
ASSETS					
Cash and cash equivalents	\$ 1	\$ -	\$ 7	\$ -	\$8
Accounts receivable	226	65	1	_	292
Inventories	218	305	_	_	523
Other current assets	12	7	1		20
Total current assets	457	377	9	-	843
Intercompany receivables	1,250	504	162	(1,916)	_
Property, plant and equipment, net	26	2,485	47	_	2,558
Port Hawkesbury biomass project	-	5	_	_	5
Investment in subsidiaries	1,554	57	_	(1,611)	_
Other assets	98	6	1		105
TOTAL ASSETS	\$ 3,385	\$ 3,434	\$ 219	\$ (3,527)	\$ 3,511
LIABILITIES AND EQUITY (DEFICIT)					
Accounts payable	\$ 50	\$ 144	\$ 1	\$ -	\$ 195
Other current liabilities		133	6		210
Total current liabilities	121	277	7	_	405
Intercompany payables	504	1,267	145	(1,916)	_
Long-term debt	3,005	152	_	_	3,157
Proceeds from NSPI for Port Hawkesbury biomass					
project	_	80	_	-	80
Other long-term liabilities	412	104	10	_	526
Equity (Deficit)	(657)	1,554	57	(1,611)	(657)
TOTAL LIABILITIES AND EQUITY (DEFICIT)	\$ 3,385	\$ 3,434	\$ 219	\$ (3,527	\$ 3,511

NEWPAGE CORPORATION SUPPLEMENTAL CONSOLIDATING STATEMENT OF OPERATIONS FIRST QUARTER ENDED MARCH 31, 2011

			Non-		
	NewPage	Guarantor	Guarantor	Consolidation/	
	Corporation	Subsidiaries	Subsidiaries	Eliminations	Consolidated
Net sales	\$ 798	\$ 814	\$ 19	\$ (727)	\$ 904
Cost of sales	765	802	18	(727)	858
Selling, general and administrative expenses	39	1	-	-	40
Equity in (earnings) loss of subsidiaries	(10)	(1)	_	11	_
Interest expense	92	3	-	-	95
Other (income) expense, net		(2)			(2)
Income (loss) before income taxes	(88)	11	1	(11)	(87)
Income tax (benefit)					1
Net income (loss)	\$ (88)	\$ 10	\$ 1	<u>\$ (11)</u>	\$ (88)

NEWPAGE CORPORATION SUPPLEMENTAL CONSOLIDATING STATEMENT OF OPERATIONS FIRST QUARTER ENDED MARCH 31, 2010

			Non-		
	NewPage	Guarantor	Guarantor	Consolidation/	
	Corporation	Subsidiaries	Subsidiary	Eliminations	Consolidated
Net sales	\$ 732	\$ 758	\$ 21	\$ (694)	\$ 817
Cost of sales	752	770	21	(694)	849
Selling, general and administrative expenses	49	_	-	-	49
Equity in (earnings) loss of subsidiaries	27	_	_	(27)	_
Interest expense	93	4	-	-	97
Other (income) expense, net	(14)	11			(3)
Income (loss) before income taxes	(175)	(27)	-	27	(175)
Income tax (benefit)					
Net income (loss)	\$ (175)	\$ (27)	\$ -	\$ 27	\$ (175)

NEWPAGE CORPORATION SUPPLEMENTAL CONDENSED CONSOLIDATING STATEMENT OF CASH FLOWS FIRST QUARTER ENDED MARCH 31, 2011

				Non-		
	NewPag	e	Guarantor	Guarantor	Consolidation/	
	Corporati	on	Subsidiaries	Subsidiaries	Eliminations	Consolidated
CASH FLOWS FROM OPERATING						
ACTIVITIES						
Net cash provided by (used for) operating						
activities	\$ -		\$ 19	\$ 1	\$ -	\$ 20
CASH FLOWS FROM INVESTING						
ACTIVITIES						
Capital expenditures	(1)	(13)	_	_	(14)
Proceeds from sales of assets	-		1	-	-	1
Expenditures for Port Hawkesbury biomass						
project	_		(10)			(10)
Net cash provided by (used for) investing						
activities	(1)	(22)	-	-	(23)
CASH FLOWS FROM FINANCING						
ACTIVITIES						
Payment of financing costs	(1)	_	-	-	(1)
Borrowings on revolving credit facility	244		_	_	_	244
Payments on revolving credit facility	(242)	_	-	-	(242)
Proceeds from NSPI for Port Hawkesbury						
biomass project	_		2			2
Net cash provided by (used for) financing						
activities	1		2	-	-	3
Effect of exchange rate changes on cash and cash						
equivalent	-		1	_		1
Net increase (decrease) in cash and cash						
equivalents	_		_	1	_	1
Cash and cash equivalents at beginning of period	1		_	7	_	8
Cash and cash equivalents at end of period	\$ 1		\$ -	\$ 8	<u></u>	\$ 9
1						

NEWPAGE CORPORATION SUPPLEMENTAL CONDENSED CONSOLIDATING STATEMENT OF CASH FLOWS FIRST QUARTER ENDED MARCH 31, 2010

					Non-				
	NewPage		Guaranto	r	Guarantor	Consolidation/			
	Corporation		Subsidiari	es	Subsidiary	Eliminations	Consolidat	Consolidated	
CASH FLOWS FROM OPERATING									
ACTIVITIES									
Net cash provided by (used for) operating									
activities	\$ 3		\$ (9)	\$ 2	\$ -	\$ (4)	
CASH FLOWS FROM INVESTING									
ACTIVITIES									
Capital expenditures	(2)	(9)	_	_	(11)	
Proceeds from sales of assets	_	_	11				11		
Net cash provided by (used for) investing									
activities	(2)	2		-	_	_		
CASH FLOWS FROM FINANCING									
ACTIVITIES									
Proceeds from issuance of long-term debt	67		_		_	_	67		
Payment of financing costs	(4)	_		-	-	(4)	
Borrowings on revolving credit facility	124		-		-	-	124		
Payments on revolving credit facility	(176)					(176)	
Net cash provided by (used for) financing									
activities	11		_		-	-	11		
Effect of exchange rate changes on cash and cash									
equivalent	_	_	1	_			1		
Net increase (decrease) in cash and cash									
equivalents	12		(6)	2	_	8		
Cash and cash equivalents at beginning of period	1	_	_	_	4		5		
Cash and cash equivalents at end of period	\$ 13	=	\$ (6	_)	\$ 6	<u>\$</u> –	\$ 13	_	

ITEM 2. MANAGEMENT' S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Overview

Company Background

We believe that we are the largest coated paper manufacturer in North America based on production capacity. Coated paper is used primarily in media and marketing applications, such as high-end advertising brochures, direct mail advertising, coated labels, magazines, magazine covers and inserts, catalogues and textbooks. We operate paper mills located in Kentucky, Maine, Maryland, Michigan, Minnesota, Wisconsin and Nova Scotia, Canada.

Trends in Our Business

North American coated paper demand is primarily driven by advertising and print media usage. In particular, the demand for certain grades of coated paper is affected by spending on catalogue and promotional materials by retailers and spending on magazine advertising, which affects the number of printed pages in magazines. During the first quarter of 2011, North American coated paper demand was flat compared to the first quarter of 2010. In February 2011, we realigned our capacity needs with the closure of our Whiting, Wisconsin mill and we did not take any market-related downtime in the first quarter of 2011. We took 39,000 tons of market-related downtime in the first quarter of 2010. We will consider the need for additional market-related downtime from time to time based on market conditions.

Coated paper production capacity in North America was lower during the first quarter of 2011 compared to the first quarter of 2010, primarily as a result of North American capacity closures, including the closure of our Whiting, Wisconsin mill in February 2011. We believe imports continue to affect the North American market, placing downward pressure on North American coated paper prices as a result of foreign overcapacity.

North American prices for coated paper products historically have been determined more by North American supply and demand than directly by raw material costs or other costs of sales. However, we believe that the announced capacity closures in coated papers and inflationary trends in producer input costs may put upward pressure on prices for our products. As a result, we expect average prices for coated paper in the last three quarters of 2011 to be higher than the last three quarters of 2010.

We experienced significant increases in our input costs during the first quarter of 2011 compared to the first quarter of 2010, due to higher prices for wood, chemicals and purchased pulp. In particular, costs of certain petroleum-based chemicals and transportation costs have increased during the first quarter of 2011 compared to first quarter of 2010, principally as a result of increases in the price of crude oil. Because of the lag time between the change in oil prices and the change in costs of certain petroleum-based chemicals, we anticipate this trend will continue throughout 2011. Overall, we expect crude oil and energy costs to remain volatile for the foreseeable future.

Port Hawkesbury Biomass Project

On November 3, 2010, NewPage Port Hawkesbury Corp. ("NPPH"), an indirect wholly-owned subsidiary of NewPage, completed the sale of certain assets, including a boiler and land at the Port Hawkesbury, Nova Scotia mill, to Nova Scotia Power Inc. ("NSPI") for a cash sales price of \$79 million. In addition, NSPI and NPPH have entered into an engineering, procurement and construction contract for NPPH to construct for NSPI a 60 MW biomass cogeneration utility plant for the generation of electricity by

December 31, 2012 for approximately C\$93 million. All costs of construction up to C\$93 million will be paid or reimbursed by NSPI. NSPI and NPPH also entered into a management, operations and maintenance agreement related to NPPH operating the utility assets for and under the control of NSPI for an initial period of 25 years plus the period prior to completion of the turbine, with three automatic five-year renewal terms, unless notice of termination is given. See Note 7 to the consolidated financial statements.

Whiting Mill Closure

During the first quarter of 2011, we permanently closed our Whiting, Wisconsin mill. Approximately 360 employees at the facility were affected. We made the decision to close the facility in order to better align capacity with market demand. As a result of this action, including the evaluation of capacity needs, during the fourth quarter of 2010, we recorded asset impairment charges for the Whiting mill and the previously-closed Kimberly mill of \$196 million in other (income) expense and classified these assets as held for sale. During the fourth quarter of 2010 and first quarter of 2011, we recognized \$2 million and \$5 million of employee-related costs recorded in cost of sales. In addition, during the first quarter of 2011 we recognized \$2 million in cost of sales for contract termination and other closure related costs and we expect to recognize additional charges in cost of sales of approximately \$15 million to \$20 million for contract termination and other closure related costs during 2011 and subsequent years as contract terms expire. The facility closure and related activities are expected to be substantially completed during the first half of 2011. The severance and other closure cost require the outlay of cash while the asset impairment charge represents a non-cash charge. We expect no material effect on future earnings or cash flows as a result of the Whiting mill closure.

Results of Operations

The following table sets forth our historical results of operations for the first quarter ended March 31, 2011 and 2010. The following discussion of our financial condition and results of operations should be read in conjunction with our financial statements and notes thereto included in this Quarterly Report. All assets, liabilities, income, expenses and cash flows presented for all periods represent those of NewPage Holding's wholly-owned subsidiary, NewPage, except for NewPage Holding's debt and equity activity and income tax effects. Unless otherwise noted, the information provided pertains to both NewPage Holding and NewPage.

First Quarter 2011 Compared to First Quarter 2010

	NewPage Holding First Quarter Ended Mar. 31,			
				31,
	20	2011 2010		10
(in millions)	\$	%	\$	%
Net sales	904	100.0	817	100.0
Cost of sales	858	94.9	849	103.9
Selling, general and administrative expenses	40	4.5	49	6.1
Interest expense	100	11.0	101	12.4
Other (income) expense, net	(2)	(0.3)	(3)	(0.5)
Income (loss) before income taxes	(92)	(10.1)	(179)	(21.9)
Income tax (benefit)	1	0.1	-	-
Net income (loss)	(93)	(10.2)	(179)	(21.9)
Supplemental Information				
Adjusted EBITDA	\$85		\$15	

Net sales for the first quarter of 2011 were \$904 million compared to \$817 million for the first quarter of 2010, an increase of \$87 million or 11%. The increase was primarily the result of higher sales volume of core paper (\$35 million) and higher average core paper prices (\$42 million) in the first quarter of 2011 compared to the first quarter of 2010. Core volume is principally coated freesheet, coated groundwood and supercalendered paper products sold in North America. Average core paper prices increased to \$912 per ton in the first quarter of 2011 compared to \$858 per ton in the first quarter of 2010. Core paper sales volume increased to 829,000 tons in the first quarter of 2011 compared to 792,000 tons in the first quarter of 2010. In February 2011, we realigned our capacity needs with the closure of our Whiting, Wisconsin mill and we did not take any market-related downtime in the first quarter of 2011. We took 39,000 tons of market-related downtime in the first quarter of 2010. We will consider the need for additional downtime from time to time based on market conditions.

Cost of sales for the first quarter of 2011 was \$858 million compared to \$849 million for the first quarter of 2010, an increase of \$9 million or 1%. The increase was primarily the result of higher core paper sales volume (\$27 million) partially offset by lower sales volumes of non-core paper. In addition, during the first quarter of 2011, in connection with the closure of the Whiting mill, we recognized charges of \$5 million for employee-related costs and \$2 million for contractual termination and other closure related costs. Gross margin (loss) for the first quarter of 2011 was 5.1% compared to (3.9)% for the first quarter of 2010 primarily as a result of higher average core paper sales prices (\$42 million) in the first quarter of 2011.

Maintenance expense at our mills totaled \$72 million and \$71 million in the first quarter of 2011 and 2010. We expect to incur significantly higher maintenance expense during the second and third quarters of 2011 compared to the same periods in 2010, as a result of required planned major maintenance.

Selling, general and administrative expenses decreased to \$40 million for the first quarter of 2011 from \$49 million for the first quarter of 2010, primarily as a result of \$8 million lower stock compensation expense. Included in the amount of equity award expense for the first quarter of 2010 is \$6 million to reflect the effects of a modification on the vested time-based options and the vesting of the prior-year unearned performance-based options. As a percentage of net sales, selling, general and administrative expenses decreased in the first quarter of 2011 to 4.5% from 6.1% in the first quarter of 2010.

Interest expense for the first quarter of 2011 was \$100 million compared to \$101 million for the first quarter of 2010. Interest expense for NewPage for the first quarter of 2011 was \$95 million compared to \$97 million for the first quarter of 2010.

Other (income) expense was \$(2) million for the first quarter of 2011 and \$(3) million for the first quarter of 2010. The amount recognized in the first quarter of 2010 includes \$11 million of net loss on disposal and impairment of assets, offset by \$(13) million of (income) recognized for alternative fuel mixture tax credits.

Income tax expense (benefit) for the first quarter of 2011 and 2010 was \$1 million and zero. For the first quarter of 2011 and 2010, we recorded a valuation allowance against our net deferred income tax benefit for federal income taxes and for certain states as it was more likely than not that we would not realize those benefits.

Net income (loss) was \$(93) million in the first quarter of 2011 compared to \$(179) million in the first quarter of 2010. Net income (loss) for NewPage was \$(88) million in the first quarter of 2011 compared to \$(175) million in the first quarter of 2010. The improvements were primarily the result of higher average paper prices and higher sales volume of core paper.

Adjusted EBITDA was \$85 million for the first quarter of 2011 compared to \$15 million for the first quarter of 2010. See "Reconciliation of Net Income (Loss) to EBITDA and Adjusted EBITDA" for further information on the use of EBITDA and Adjusted EBITDA as measurement tools.

Reconciliation of Net Income (Loss) to EBITDA and Adjusted EBITDA

EBITDA is defined as net income (loss) before interest expense, income taxes, depreciation and amortization. EBITDA and Adjusted EBITDA are not measures of our performance under U.S. GAAP, are not intended to represent net income (loss), and should not be used as alternatives to net income (loss) as indicators of performance. EBITDA and Adjusted EBITDA are shown because they are bases upon which our management assesses performance and are primary components of certain covenants under our revolving credit facility. In addition, our management believes EBITDA and Adjusted EBITDA are useful to investors because they and similar measures are frequently used by securities analysts, investors and other interested parties in the evaluation of companies with substantial financial leverage. The use of EBITDA and Adjusted EBITDA instead of net income (loss) has limitations as an analytical tool, and you should not consider them in isolation, or as a substitute for analysis of our results as reported under U.S. GAAP. Some of these limitations are:

EBITDA and Adjusted EBITDA do not reflect our current cash expenditure requirements, or future requirements, for capital expenditures or contractual commitments

EBITDA and Adjusted EBITDA do not reflect changes in, or cash requirements for, our working capital needs

EBITDA and Adjusted EBITDA do not reflect the significant interest expense, or the cash requirements necessary to service interest or principal payments, on our debt

although depreciation and amortization are non-cash charges, the assets being depreciated and amortized often will have to be replaced in the future, and EBITDA and Adjusted EBITDA do not reflect any cash requirements for such replacements

our measures of EBITDA and Adjusted EBITDA are not necessarily comparable to other similarly titled captions of other companies due to potential inconsistencies in the methods of calculation

Because of these limitations, EBITDA and Adjusted EBITDA should not be considered as discretionary cash available to us to reinvest in the growth of our business.

The following table presents a reconciliation of net income (loss) to EBITDA and Adjusted EBITDA:

	NewPage Holding First Quarter Ended Mar. 31,	
(in millions)	2011	2010
Net income (loss)	\$(93)	\$(179)
Interest expense	100	101
Income tax (benefit)	1	_
Depreciation and amortization	62	68
EBITDA	70	(10)
Equity awards	_	8
(Gain) loss on disposal and impairment of assets	1	6
Non-cash U.S. pension expense	6	9
Integration and related severance costs	8	2
Adjusted EBITDA	\$85	\$15

Liquidity and Capital Resources

Available Liquidity

As of March 31, 2011, our principal sources of liquidity include cash generated from operating activities and availability under our revolving credit facility. The amount of borrowings and letters of credit available to NewPage pursuant to the revolving credit facility is limited to the lesser of \$500 million or an amount determined by reference to a borrowing base (\$416 million as of March 31, 2011).

In January 2011, we amended our revolving credit facility to extend its termination date. For those lenders who accepted this amendment (aggregating to \$470 million), the revolving credit facility matures on December 21, 2012, unless we do not repay or refinance our floating rate and 10% second-lien senior secured notes ("Second-Lien Notes") by December 2, 2011, in which case the revolving credit facility matures on March 1, 2012. For the remaining \$30 million of commitment, the revolving credit facility matures on December 21, 2012 unless we do not repay or refinance our Second-Lien Notes by July 4, 2011, in which case the \$30 million portion of the revolving credit facility matures on October 3, 2011. As of March 31, 2011, we had \$161 million available for borrowing in excess of the \$50 million required minimum, after reduction for \$110 million in letters of credit and \$95 million in outstanding borrowings under the revolving credit facility. During the first quarter of 2011, our average daily balance outstanding under our revolving credit facility was \$81 million with a weighted-average interest rate of 4.6%.

The 11.375% senior secured first-lien notes (the "First-Lien Notes") mature on the earlier of (i) December 31, 2014 or (ii) the date that is 31 days prior to the maturity date of the Second-Lien Notes, the senior subordinated notes, the NewPage Holding PIK notes or any refinancing thereof. The Second-Lien Notes mature on May 1, 2012. However, if we do not repay or refinance our Second-Lien Notes by January 31,

2012, the First-Lien Notes would mature on March 31, 2012. The senior subordinated notes mature on May 1, 2013. The NewPage Holding PIK Notes mature on November 1, 2013. Aggregate indebtedness as of March 31, 2011 totaled \$3,475 million, which includes \$3,247 million at NewPage.

As of March 31, 2011, our current liabilities exceeded our current assets by \$1,378 million. This deficit is driven by the classification of our First-Lien Notes and our revolving credit facility as current liabilities. Absent a refinancing of our Second-Lien Notes, an additional \$1,030 million will become current during the second quarter of 2011. We are exploring various alternatives to address our ongoing capital needs. However, we cannot assure you that we will be able to refinance any of our indebtedness, or that we will be able to do so on commercially reasonable terms. Nor can we assure you that these measures will be sufficient to satisfy our scheduled debt service obligations, or that we will be able to fund our business and operations without additional capital. If we are unable to generate sufficient cash flow or refinance our debt, including our revolving credit facility which is a principal source of our liquidity, prior to the scheduled maturities, it would have a material adverse effect on our financial condition, the value of the outstanding debt and our ability to make required cash payments under our indebtedness.

Cash Flows

Cash provided by (used for) operating activities was \$20 million during the first quarter of 2011 compared to \$(4) million during the first quarter of 2010, primarily the result of improvements in operations, including higher sales volume of core paper and higher average paper prices. Investing activities in first quarter of 2011 include \$14 million for capital expenditures and \$10 million of expenditures for the Port Hawkesbury biomass project. Financing activities primarily consisted of borrowings and payments on the revolving credit facility.

Capital Expenditures

Capital expenditures were \$14 million and \$11 million for the first quarter ended March 31, 2011 and 2010.

Financial Discussion

During the fourth quarter of 2010, we announced the permanent closure of the Whiting, Wisconsin mill by the end of February 2011. As a result of this action, during the fourth quarter of 2010, we recorded asset impairment charges for the Whiting mill and the previously-closed Kimberly mill of \$196 million in other (income) expense and classified these assets as held for sale. During the fourth quarter of 2010 and first quarter of 2011, we recognized \$2 million and \$5 million of employee-related costs recorded in cost of sales. In addition, during the first quarter of 2011, we recognized \$2 million in cost of sales for contract termination and other closure related costs and we expect to recognize additional charges in cost of sales of approximately \$15 million to \$20 million for contract termination and other closure related costs during 2011 and subsequent years as contract terms expire.

In March 2011, Rumford Paper Company ("Rumford"), a wholly-owned subsidiary of NewPage Corporation, and Rumford Cogeneration Company Limited Partnership, a subsidiary of Rumford, both indirect subsidiaries of NewPage Holding Corporation, entered into an Asset Sale Agreement for the sale of Rumford's cogeneration energy assets to ReEnergy Rumford LLC ("ReEnergy") for a sale price of \$61 million, payable one-half in cash at closing and one-half in the form of a 10-year note receivable. The transaction also contemplates an ongoing contractual relationship between the parties, pursuant to which Rumford will supply biomass fuel to ReEnergy and ReEnergy will supply electricity and thermal energy to Rumford. The transaction is not expected to have a material effect on the operating costs of the Rumford mill. We plan to invest the net cash proceeds in our business or use the proceeds to repay

indebtedness, or both, in accordance with the terms of our debt instruments. The transaction, which is expected to close during the second or third quarter of 2011, is subject to customary conditions, including the receipt of regulatory approvals and the finalization of agreements outlined in term sheets.

On November 3, 2010, NPPH completed the sale of certain assets, including a boiler and land at the Port Hawkesbury, Nova Scotia mill, to NSPI for a cash sales price of \$79 million. In addition, NSPI and NPPH have entered into an engineering, procurement and construction contract for NPPH to construct for NSPI a 60 MW biomass cogeneration utility plant for the generation of electricity by December 31, 2012 for approximately C\$93 million. NSPI and NPPH also entered into a management, operations and maintenance agreement related to NPPH operating the utility assets for NSPI. NPPH was required to post \$10 million of letters of credit under these arrangements in 2010 and an additional \$5 million in 2011. See Note 7 to the consolidated financial statements.

In February 2010, CWPCo announced that it signed an agreement with Wisconsin Rapids Water Works Lighting Commission to sell the CWPCo Utility, as amended. The purchase price will be equal to the net book value of the assets at the time of closing, subject to adjustment in the agreement. The transaction remains subject to regulatory approvals. On November 1, 2010, CWPCo entered into an asset sale agreement to sell five hydroelectric projects to Great Lakes Utilities for a net cash sales price of approximately \$70 million. The closing of the transaction is subject to completion of the CWPCo Utility sale and regulatory approval. We continue to hold and operate the assets in the normal course of our operations.

On March 21, 2011, U.S. Environmental Protection Agency (USEPA) finalized its Boiler MACT rule, which provides for Industrial Boiler Maximum Achievable Control Technology (MACT) standards to regulate emissions of hazardous air pollutants for industrial, commercial and institutional boilers and process heaters. Compliance with the final Boiler MACT rule is currently required no later than March 21, 2014. Our preliminary evaluation of the final Boiler MACT rule indicates significant capital expenditures for additional emission control equipment may be needed at our mills with solid fuel boilers to comply with the rule. In addition, our mills will incur increased operating expenses associated with compliance and operation of the new control equipment. USEPA has also announced they will be reconsidering portions of the Boiler MACT rule because the final rule differs from what was proposed. At this time it is not known what changes to the rule might occur as a result of the reconsideration process. We continue to follow the reconsideration process and will adjust our Boiler MACT compliance evaluation as needed for any changes to the compliance requirements.

We have various investments held by our defined-benefit pension plan trusts. The returns on these assets have generally matched the broader market. We are monitoring the effects of our underfunded pension plans on our minimum pension funding requirements and pension expense for future periods. We currently do not anticipate material increases in our minimum funding requirements during 2011.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

Interest Rate Risk

As of March 31, 2011 and December 31, 2010, \$548 million and \$541 million of our debt consisted of borrowings with variable interest rates. If market interest rates increase, variable-rate debt will create higher debt service requirements, which could adversely affect our cash flow or compliance with our debt covenants. The potential annual increase in interest expense resulting from a 100 basis point increase in quoted interest rates on our debt balances outstanding at March 31, 2011 and December 31, 2010, would be \$5 million.

Foreign Currency Risk

Our Canadian subsidiary makes a portion of its purchases and sales in U.S. dollars. As a result, it is subject to transaction exposures that arise from foreign exchange movements between the date that the foreign currency transaction is recorded and the date it is consummated. Foreign currency exchange contracts may be used periodically to manage the variability in cash flows and revenues from the forecasted payment or receipt of currencies other than our functional currency, the U.S. dollar. As of March 31, 2011 and December 31, 2010, we had no foreign currency forward contracts outstanding.

ITEM 4. CONTROLS AND PROCEDURES

We maintain a system of internal accounting controls designed to provide reasonable assurance that transactions are properly recorded and summarized so that reliable financial records and reports can be prepared and assets safeguarded. In addition, a system of disclosure controls is maintained to ensure that information required to be disclosed is recorded, processed, summarized and reported in a timely manner to management responsible for the preparation and reporting of our financial information.

Management, with the participation of our chief executive officer and chief accounting officer, assessed the disclosure control systems as being effective as they encompass material matters for the quarter ended March 31, 2011. To the best of our knowledge, there were no changes in the internal controls over financial reporting that occurred during the quarter ended March 31, 2011 that have materially affected, or are reasonably likely to materially affect, our internal controls over financial reporting.

PART II OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

In a notice dated January 25, 2011, staff at the Federal Energy Regulatory Commission, or FERC, reported that they had preliminarily determined that our subsidiary, Rumford Paper Company, or Rumford, violated FERC regulations in connection with Rumford's participation in ISO-New England's Day Ahead Load Response Program from July 2007 through February 2008. Under the program, participants were paid to reduce their electrical load on the New England power grid from their "baseline" load levels. FERC asserts that Rumford improperly established and maintained an artificially high baseline, allowing it to receive increased payments under the program. Rumford cooperated with FERC throughout its investigation and has denied any wrongdoing.

In September 2009, NewPage, along with two other U.S. paper producers and the United Steelworkers Union, filed antidumping and countervailing duty petitions with the U.S. Department of Commerce and the U.S. International Trade Commission alleging that manufacturers of certain coated paper in China and Indonesia are dumping their products in the United States and that these manufacturers have been subsidized by their governments in violation of U.S. trade laws. Following affirmative final determinations by both the U.S. Department of Commerce and the U.S. International Trade Commission, the U.S. Department of Commerce imposed duties to offset the threat of dumping and government subsidies. Producers in China and Indonesia have appealed these determinations to the U.S. Court of International Trade. Pending a decision on the appeal, these duties will remain in effect for five years, subject to annual administrative reviews.

ITEM 5. OTHER INFORMATION

Resignation and Election of Director

On May 10, 2011, the boards of directors of NewPage Group Inc., NewPage Holding Corporation and NewPage Corporation (collectively, the "NewPage Companies") accepted the resignation of Robert L. Nardelli from the boards of directors of the NewPage Companies, effective as of May 10, 2011. There were no disagreements between Mr. Nardelli and the NewPage Companies that resulted in Mr. Nardelli's resignation.

On May 10, 2011, the boards of directors of the NewPage Companies elected Thomas L. Zambelli to serve as a director of the NewPage Companies, effective May 10, 2011. Since August 2007, Mr. Zambelli, age 60, has been a vice president of Cerberus Operating and Advisory Company LLC, an affiliate of the indirect controlling stockholder of the NewPage Companies. From March 2009 to May 2010, he was the chief investment officer of Chrysler Financial Services Americas LLC. From April 2000 to August 2007 he was a senior director at Zolfo Cooper LLC, a financial advisory firm.

ITEM 6. EXHIBITS

Exhibit	
Number	Description
10.1	Asset Sale Agreement dated March 16, 2011, among Rumford Paper Company, Rumford Cogeneration Company Limited Partnership and ReEnergy Rumford LLC
31.1	Certification of Chief Executive Officer pursuant to Rules 13a-14 and 15d-14, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002
31.2	Certification of Chief Accounting Officer pursuant to Rules 13a-14 and 15d-14, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002
32.1	Certification of Chief Executive Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002
32.2	Certification of Chief Accounting Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, NewPage Holding Corporation and NewPage Corporation have duly caused this Report to be signed on their behalf by the undersigned, thereunto duly authorized.

NEWPAGE HOLDING CORPORATION

By: /s/ George F. Martin

George F. Martin

President and Chief Executive Officer

(Principal Executive Officer)

Date: May 12, 2011

By: /s/ Ronald J. Arling

Ronald J. Arling

Controller and Chief Accounting Officer

(Principal Accounting Officer)

Date: May 12, 2011

NEWPAGE CORPORATION

By: /s/ George F. Martin

George F. Martin

President and Chief Executive Officer

(Principal Executive Officer)

Date: May 12, 2011

By: /s/ Ronald J. Arling

Ronald J. Arling

Controller and Chief Accounting Officer

(Principal Accounting Officer)

Date: May 12, 2011

ASSET SALE AGREEMENT

Between

RUMFORD COGENERATION COMPANY LIMITED PARTNERSHIP

And

RUMFORD PAPER COMPANY,

As Selling Parties,

And

REENERGY RUMFORD LLC,

As Buyer

Dated as of March 16, 2011

TABLE OF CONTENTS

ARTICLE 1	DEFINITIONS	
Section 1.1	Definitions	1
Section 1.2	Interpretation	10
ARTICLE 2	PURCHASE AND SALE OF ASSETS	
Section 2.1	Purchase and Sale of Assets	11
Section 2.2	Excluded Assets	12
Section 2.3	Assumed Liabilities	13
Section 2.4	Excluded Liabilities	13
Section 2.5	Conditional Assignment	13
Section 2.6	Purchase Price and Payment	14
Section 2.7	Prorations and Expenses; Certain Tax Matters	14
Section 2.8	Purchase Price Allocation	15
ARTICLE 3	REPRESENTATIONS AND WARRANTIES OF THE SELLING PARTIES	
Section 3.1	Scope	16
Section 3.2	Organization and Good Standing	16
Section 3.3	Enforceability, Authority and No Conflict	16
Section 3.4	Third Party Consents	17
Section 3.5	Financial Records	17
Section 3.6	Sufficiency of Assets	17
Section 3.7	Tangible Personal Property	17
Section 3.8	Real Property	17
Section 3.9	Material Contracts	19
Section 3.10	Absence of Certain Changes and Events	19
Section 3.11	Environmental	20
Section 3.12	Compliance with Law and Permits	21
Section 3.13	Litigation and Orders	22
Section 3.14	Taxes	22
Section 3.15	Intellectual Property	22
Section 3.16	Insurance	23
Section 3.17	Labor Matters	23
Section 3.18	No Commissions	23
Section 3.19	Disclaimer	23
ARTICLE 4	REPRESENTATIONS AND WARRANTIES OF BUYER	
Section 4.1	Scope	23
Section 4.2	Organization and Good Standing	23
Section 4.3	Enforceability, Authority and No Conflict	24
Section 4.4	Third Party Consents	24
Section 4.5	Litigation and Orders	24
Section 4.6	Due Diligence	24
Section 4.7	Availability of Funds	25
Section 4.8	No Commissions	25
Section 4.9	No Buyer Material Adverse Effect	25
Section 4.10	Disclaimer	25

ARTICLE 5	COVENANTS	
Section 5.1	Conduct of Business	25
Section 5.2	Consents, Approvals and Contacts	26
Section 5.3	Access and Confidential Information	28
Section 5.4	Expenses	29
Section 5.5	Efforts to Complete the Transactions	29
Section 5.6	Publicity	30
Section 5.7	Tax Matters	30
Section 5.8	Notices of Changes	31
Section 5.9	Updates to Schedules and Exhibits	31
Section 5.10	Interconnection Agreement	31
Section 5.11	Further Agreements	32
Section 5.12	Real Estate	32
Section 5.13	Casualty	33
Section 5.14	Environmental Matters	34
Section 5.15	Separation	35
Section 5.16	Conditions to Closing	35
Section 5.17	Governing Documents	35
Section 5.18	Other Transactions	35
Section 5.19	Project Employees	36
Section 5.20	Inventory Credit	36
ARTICLE 6	CONDITIONS TO THE CLOSING	
Section 6.1	Conditions to the Obligations of Buyer	36
Section 6.2	Conditions to the Obligations of the Selling Parties	38
ARTICLE 7	CLOSING	
Section 7.1	Closing	39
Section 7.2	Timing and Effectiveness of Actions	39
Section 7.3	Selling Parties' Deliveries	40
Section 7.4	Buyer's Deliveries	41
ARTICLE 8	TERMINATION	
Section 8.1	Termination	42
Section 8.2	Effect of Termination	43
ARTICLE 9	INDEMNIFICATION	
Section 9.1	Survival	44
Section 9.2	Indemnification By the Selling Parties	44
Section 9.3	Indemnification By Buyer	45
Section 9.4	Indemnification Procedures	46
Section 9.5	Settlement and Defense of Third Party Claims	47
Section 9.6	Adjustment to Purchase Price	47
Section 9.7	General Limitations	47
Section 9.8	Exclusive Remedies	49
ARTICLE 10	MISCELLANEOUS	
Section 10.1	Entire Agreement; Amendment	50
Section 10.2	Notices	50

Section 10.3	Consents and Approvals	50
Section 10.4	Counterparts	51
Section 10.5	Parties in Interest; Assignment	51
Section 10.6	No Waiver	51
Section 10.7	No Third Party Beneficiaries	51
Section 10.8	Good Faith	51
Section 10.9	Governing Law	51
Section 10.10	Consent to Jurisdiction and Venue	51
Section 10.11	Disclosures and Announcements	51
Section 10.12	Non-Recourse	51
Section 10 13	Exhibits	52

ASSET SALE AGREEMENT

THIS ASSET SALE AGREEMENT ("Agreement") is made and entered into as of the 16th day of March, 2011, by and among RUMFORD COGENERATION COMPANY LIMITED PARTNERSHIP, a Maine limited partnership ("Seller"), RUMFORD PAPER COMPANY, a Delaware corporation ("RPC" and together with Seller, the "Selling Parties"), and REENERGY RUMFORD LLC, a Delaware limited liability company ("Buyer"). In consideration of the representations, warranties and covenants contained in this Agreement and for other good and valuable consideration, the receipt and sufficiency of which are acknowledged, and intending to be legally bound, the Parties agree as follows:

ARTICLE 1 DEFINITIONS

- 1.1 <u>Definitions</u>. For purposes of this Agreement, the terms set forth below are defined as follows:
- "Additional Survey Work" is defined in Section 5.12.1.
- "Affiliate" means, (1) with respect to the Selling Parties, NewPage Corporation, a Delaware corporation, and any entity in which NewPage Corporation owns, directly or indirectly, a majority of the capital stock or other equity interests, and (2) with respect to Buyer, ReEnergy Holdings LLC, a Delaware limited liability company, and any entity in which ReEnergy Holdings LLC owns, directly or indirectly, a majority of the capital stock or other equity interests.
 - "Agreement" is defined in the preamble to this Agreement.
- "Ancillary Agreements" means each of (1) the Assignment and Assumption Agreement, (2) the Bill of Sale, (3) the Fuel Supply Agreement, (4) the Interconnection Assignment Agreement, (5) the Joint Use Agreement, (6) the Mortgage, (7) the Quitclaim Deed, (8) the Energy Supply Agreement, (9) the Reciprocal Easement Agreement, (10) the Recovery Boiler Agreement, (11) the Secured Note, (12) the Security Agreement, (13) the Shared Services Agreement, (14) the Undivided Ownership Agreement, (15) the NewPage Guaranty; (16) the ReEnergy Guaranty; (17) the Joint Operating Committee Agreement; (18) the Non-Solicitation Agreement, and (19) any other written contract, agreement or commitment entered into by and between the Parties simultaneous with the Closing and relating to this Agreement or the transactions contemplated by this Agreement.
- "Assignment and Assumption Agreement" means the Assignment and Assumption Agreement to be executed and delivered by the Selling Parties and Buyer at Closing, in the form of Exhibit 1.1(A).
 - "Assumed Contract" is defined in Section 2.1(4).
 - "Assumed Liabilities" is defined in Section 2.3.
 - "Bill of Sale" means the bill of sale to be executed and delivered by the Selling Parties at the Closing, in the form of Exhibit 1.1(B).
- "Business Day" means a day other than a Saturday, Sunday or other day on which commercial banks in Portland, Maine are authorized or required by Law to be closed.

- "Buyer" is defined in the preamble to this Agreement.
- "Buyer Confidential Information" is defined in Section 5.3.2.
- "Buyer Group" is defined in Section 9.2.
- "Buyer's Intended Use" means Buyer's intended use of the Project for generation of electrical energy and/or steam as configured as of the Closing Date as modified by the modifications and improvements to the Project contemplated under Schedule 5.14.2.2.
- "Buyer's Knowledge" means, with respect to a particular fact, circumstance or condition, the actual knowledge by an individual listed below of (1) that particular fact, circumstance or condition, or (2) underlying facts, circumstances or conditions that would reasonably be expected to put a person having comparable responsibilities to that particular individual on notice of that particular fact, circumstance or condition or that would reasonably be expected to cause a person having comparable responsibilities to that particular individual to make due inquiry or investigation by contacting the appropriate party: Larry Richardson, Chief Executive Officer, Thomas Beck, Chief Commercial Officer, William Ralston, Chief Risk Officer, and Stephen Hall, Regional Manager-Energy Operations.
- "Buyer Material Adverse Effect" means a violation, inaccuracy, breach, default, failure to comply, loss, effect, fact, agreement, arrangement, commitment, understanding or obligation, which, individually or collectively taken as a whole, has had or could reasonably be expected to have a material adverse effect or material adverse impact on (1) the business, financial conditions, assets, results of operations or prospects of Buyer or ReEnergy Holdings LLC; or (2) the ability of Buyer or ReEnergy Holdings LLC to perform their respective obligations under this Agreement or any Ancillary Agreements or to consummate the transactions contemplated in this Agreement or any Ancillary Agreement, and will exclude any effect resulting or arising from any (a) change in Law, (b) change in interest rates or general economic conditions in the United States, or (c) national or international political or social conditions, including the engagement by the United States in hostilities, whether or not pursuant to the declaration of a national emergency or war, or the occurrence of any military or terrorist attack upon the United States or any of its territories, possessions, or diplomatic or consular offices or upon any military installation, equipment or personnel of the United States.
 - "Buyer Required Regulatory Approvals" means the approvals set forth on Exhibit 1.1(C).
 - "Cash Purchase Price" is defined in Section 2.6.1(b).
- "Claim" means any action, arbitration, hearing, litigation, petition, claim, demand, suit or notice of noncompliance or violation (whether civil, criminal, administrative, regulatory, judicial or investigative, whether public or private) by or before any Governmental Authority or other Person.
 - "Closing" is defined in Section 7.1.
 - "Closing Date" is defined in Section 7.1.
 - "CMP" means Central Maine Power Company, a Maine public utility.
 - "Code" means the Internal Revenue Code of 1986, as amended, and the regulations adopted thereunder.
 - "Collective Bargaining Agreements" is defined in Section 3.17.

- "Confidential Information" is defined in the Confidentiality Agreement.
- "Confidentiality Agreement" means the Confidentiality Agreement between NewPage Corporation and Buyer dated June 9, 2010.
- "Contract" means any agreement, contract, contractual license, lease, commitment, consensual obligation, promise, arrangement or undertaking (whether written or oral and whether express or implied).
- "CPA Firm" means a nationally recognized firm of independent certified public accountants that has not provided services to Seller, Buyer or any of their respective Affiliates during the two years prior to the Execution Date, and that is mutually appointed by Seller and Buyer.
- "Documents" means all existing and available files, documents, instruments, papers, books, reports, tapes, microfilms, photographs, maps, drawings, blue prints, designs and as-built plans, letters, ledgers, journals, maintenance records, soil and other Environmental Condition reports, engineering and building plans, regulatory filings, operating data and plans, technical and user documentation, and other similar materials primarily related to the Project, the Purchased Assets, or the Assumed Liabilities, in each case whether or not in electronic form. Documents do not include: (1) information which, if provided to Buyer, would violate any Law or Order or the Governing Documents of the Selling Parties or their Affiliates, (2) bids, letters of intent, expressions of interest, or other proposals received by the Selling Parties or their Affiliates from any Person in connection with the transactions contemplated by this Agreement or otherwise, and information and analyses relating to such communications and transactions, (3) any information, the disclosure of which would jeopardize any legal privilege available to the Selling Parties or their Affiliates relating to such information or would cause the Selling Parties or their Affiliates to breach a confidentiality obligation by which it is bound, except for such information that is material to Buyer's Intended Use and the disclosure of which can be arranged subject to a mutually-agreeable written confidentially arrangement that is in effect at the time of disclosure, (4) any valuations of or related to the Project, the Purchased Assets or the Assumed Liabilities, or (5) any information management systems of Seller or any of its Affiliates that are not used exclusively in connection with the Purchased Assets, exclusive of any data contained therein primarily related to the Project, the Purchased Assets or the Assumed Liabilities, which shall be deemed included within the definition of Documents.

"Defect Notice" is defined in Section 5.12.2.

"Easement Rights" means the easements to be granted by RPC to Buyer in the Quitclaim Deed and the Reciprocal Easement Agreement, as applicable.

"Effective Time" means 12:01 a.m. on the Closing Date.

"Energy Supply Agreement" means the energy supply agreement to be executed and delivered by RPC and Buyer at the Closing, the summary terms of which are set forth on Exhibit 1.1(D).

"Environment" means all or any of the following: real property, soil, land surface and subsurface strata, water vapor, surface waters (including navigable waters, streams, ponds, drainage basins, and wetlands), groundwater, drinking water supply, sediments, ambient air (including the air within buildings and the air within other natural or man-made structures above or below ground), fish, plant and animal life, and any other natural resource.

"*Environmental Claim*" means any past, present or future Claim relating to an Environmental Condition, including a related Claim by any Governmental Authority or other Person seeking Remedial Action or other relief pursuant to any Environmental Law.

"Environmental Condition" means any condition, known or unknown, foreseen or unforeseen, relating to or arising out of: (1) a Release, (2) violation of any Environmental Law or Environmental Permit, (3) the presence of any Hazardous Material, or (4) the handling of Hazardous Materials.

"Environmental Laws" means any Law or Order, in each case as amended and in effect as of the Closing Date, regulating or otherwise relating to (1) the protection of human health or the protection, preservation or restoration of the Environment; (2) the exposure to, or the management, manufacture, possession, presence, use, storage, recycling, abatement, removal, remediation, treatment, generation, transportation, processing, handling, labeling, production, or disposal of any Hazardous Material, or (3) the Release or threatened Release of Hazardous Material into the Environment. Environmental Laws include the Comprehensive Environmental Response, Compensation, and Liability Act, 42 U.S.C. § 9601, et seq.; the Federal Insecticide, Fungicide and Rodenticide Act, 7 U.S.C. § 136, et seq.; the Resource Conservation and Recovery Act, 42 U.S.C. § 6901, et seq.; the Toxic Substances Control Act, 15 U.S.C. § 2601, et seq.; the Clean Air Act, 42 U.S.C. § 7401, et seq.; the Federal Water Pollution Control Act, 33 U.S.C. § 1251, et seq.; the Oil Pollution Act, 33 U.S.C. § 2701, et seq.; the Endangered Species Act, 16 U.S.C. § 1531, et seq.; the National Environmental Policy Act, 42 U.S.C. § 4321, et seq.; the Occupational Safety and Health Act, 29 U.S.C. § 651, et seq.; the Safe Drinking Water Act, 42 U.S.C. § 300f, et seq.; Emergency Planning and Community Right-to-Know Act, 42 U.S.C. § 11001, et seq.; Atomic Energy Act, 42 U.S.C. § 2014, et seq.; Nuclear Waste Policy Act, 42 U.S.C. § 10101, et seq.; and their state and local counterparts or equivalents, all as amended from time to time, and regulations issued pursuant to any of those statutes.

"Environmental Permits" means any consent, license, registration, permit or other authorization issued, granted, given or otherwise made available by or under the authority of any Governmental Authority pursuant to applicable Environmental Laws that are used or held by either Selling Party for the operation of the Purchased Assets.

"Estimated Restoration Costs" is defined in Section 5.13.1.

"Excluded Assets" is defined in Section 2.2.

"Excluded Liabilities" is defined in Section 2.4.

"Excluded Real Property" is defined in Section 5.12.5.

"Execution Date" means the day and year set forth in the preamble to this Agreement.

"Expiration Date" is defined in Section 8.1(2).

"FERC" means the Federal Energy Regulatory Commission or its successor.

"Final Order" means, with respect to any Required Regulatory Approval, an Order granting the Required Regulatory Approval that has not been revised, stayed, enjoined, set aside, annulled, or suspended, and with respect to which all conditions to effectiveness prescribed in the Required Regulatory Approval or the Order or otherwise by Law or Order have been satisfied.

"Financial Records" is defined in Section 3.5.

- "Fuel Supply Agreement" means the fuel supply agreement to be executed and delivered by RPC and Buyer at the Closing, the summary terms of which are set forth on Exhibit 1.1(E).
- "Governing Documents" of a Person means, as applicable, its articles or certificate of incorporation or association, bylaws, operating agreement, or comparable governing documents.
- "Governmental Authority" means any nation, state, city, locality, municipality, or other political subdivision and any body or authority exercising judicial, legislative, regulatory or administrative functions for any of the foregoing (including any agency, department, board or commission), or any court or arbitrator.
- "Hazardous Material" means any chemical, material, substance, or waste which is now or subsequently defined, listed, designated or classified as, or included in the definition of, a hazardous, toxic, extremely hazardous, radioactive, a "pollutant," or "contaminant," or otherwise regulated under any applicable Environmental Laws, including: any petroleum, petroleum hydrocarbons, petroleum products (including crude oil or any fraction thereof), natural gas, natural gas liquids, liquefied natural gas or synthetic gas useable for fuel (or mixtures of natural gas and such synthetic gas), or oil and gas exploration or production waste, polychlorinated biphenyls (PCBs), asbestos-containing materials, mercury, urea-formaldehyde foam insulation (UFFI), mold, mold spores and mycotoxins, lead-based paints, and any other chemical, material, substances, waste, or mixture thereof which is prohibited, limited, or regulated by Environmental Laws.
- "*Historical Environmental Liabilities*" means those Liabilities arising out of an Environmental Condition at, on or under the Purchased Assets that existed prior to May 2, 2005, excluding those Environmental Conditions set forth on Exhibit 1.1(F).
- "HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder.
 - "Indemnified Party" is defined in Section 9.4.1.
 - "Indemnifying Party" is defined in Section 9.4.1.
 - "Indemnity Payment" is defined in Section 9.4.3.
- "Indemnifiable Environmental Liabilities" mean (1) those Environmental Conditions described on Exhibit 1.1(F), (2) those Liabilities arising out of an Environmental Condition at, on or under the Purchased Assets, that first arose after May 2, 2005 and prior to the Effective Time and, (3) any Liabilities resulting from exacerbation between May 2, 2005 and the Effective Time of an Environmental Condition that existed prior to May 2, 2005.
 - "Interconnection Agreement" is defined in Section 5.9.
 - "Interconnection Assignment Agreement" is defined in Section 5.10.
 - "Intellectual Property" is defined in Section 3.15.1.
- "*Inventory*" means the Selling Parties' inventories of fuel, spare parts and all other materials, tools, parts and supplies to be used or consumed by Selling Parties either (1) solely in connection with the operation of the Project and listed on Exhibit 2.1(3)(A), or (2) in connection with the operation of the Project and also in connection with its other operations and listed on Exhibit 2.1(3)(B).

- "Joint Operating Committee Agreement" means the Joint Operating Committee Agreement to be executed and delivered by RPC and Buyer at the Closing, the summary terms of which are set forth on Exhibit 1.1(G).
- "Joint Use Agreement" means the joint use agreement to be executed and delivered by RPC and Buyer at the Closing, the summary terms of which are set forth on Exhibit 1.1(H).
- "Laws" means applicable laws, statutes, charters, ordinances, decisional law, common law, rules and regulations, and any binding administrative interpretations thereof.
 - "Licensed Intellectual Property" is defined in Section 3.15.3.
- "Liabilities" means liabilities, obligations, losses, damages, payments, costs, and expenses, including the costs and expenses of any and all related actions, suits, proceedings, assessments, judgments, settlements, and compromises (including all costs of investigation and defense and reasonable attorneys' and consultants' fees and expenses), whether or not involving a Third Party Claim.
- "*Lien*" means a lien, mortgage, pledge, security interest, real property license, easement, quasi-easement, covenant, condition, declaration, imperfection of title or other encumbrance or restriction.
 - "Material Contracts" is defined in Section 3.9.
 - "Minimum Survey Requirements" is defined in Section 5.12.1.
- "Mortgage" means the mortgage, security agreement and financing statement in which Buyer grants to RPC a mortgage in the Real Property and fixtures included in the Purchased Assets as security for payment of the Secured Note, to be executed and delivered by Buyer at the Closing, the summary terms of which are set forth on Exhibit 1.1(I).
- "NewPage Guaranty" means the guaranty from NewPage Corporation, a form of which is attached as Exhibit 1.1(J) and the scope of which is set forth in Exhibit 1.1(J).
- "Non-Solicitation Agreement" means that non-solicitation agreement to be executed and delivered by RPC and Buyer at the Closing, the form and substance of which are set forth on Exhibit 1.1(K).
- "Order" means an order, injunction, judgment, decree, ruling, writ, assessment or arbitration award of a Governmental Authority or arbitrator, but excluding all Permits and Environmental Permits.
- "Ordinary Course of Business" an action taken by a Person that (1) is consistent in nature, scope and magnitude with the past practices of that Person and is taken in the ordinary course of the normal, day-to-day operations of that Person, (2) does not require authorization by the board of directors, partners, or members of that Person (or by any Person or group of Persons exercising similar authority) and does not require any other separate or special authorization of any nature, and (3) does not constitute a breach of any contract or agreement or a tort or infringement or violation of Law by that Person.
 - "Owned Intellectual Property" is defined in Section 3.15.2.
 - "Owned Real Property" is defined in Section 3.8.1.
 - "Party" means any of Buyer, Seller or RPC, as indicated by the context, and "Parties" means Buyer and the Selling Parties.

"Permit" means any consent, license, registration, permit or other authorization issued, granted, given or otherwise made available by or under the authority of any Governmental Authority or pursuant to any Law (excluding any Environmental Law) that relates to, or is necessary for, the ownership or the operation of the Purchased Assets, but excluding Environmental Permits.

"Permitted Liens" means (1) those Liens described in Exhibit 1.1(L); (2) Liens securing or created by or in respect of any of the Assumed Liabilities; (3) statutory Liens for current Taxes or assessments not yet due and payable; (4) mechanics', carriers', workers', repairers', landlords', and other similar Liens arising or incurred and which will be paid, to the extent payable prior to the Effective Time or allocable to the period prior to the Effective Time, in the Ordinary Course of Business relating to obligations as to which there is no default on the part of a Selling Party or the validity or amount of which is being contested in good faith by appropriate proceedings, provided any such Liens affecting Real Property are insured over by the Title Company at the Closing; (5) zoning, entitlement, restriction, and other land use and environmental regulations by any Governmental Authorities; (6) any Liens contemplated or created by this Agreement or the Ancillary Agreements; and (7) any covenants, conditions, restrictions, or other Liens, or survey matters set forth on the Title Commitment or Survey and accepted or deemed accepted by Buyer pursuant to Section 5.12.2 or 5.12.3.

"Permitted Activity" is defined in Section 5.14.2.2.

"Person" means an individual, partnership, corporation, limited liability company, association, municipality, municipal service corporation, joint stock company, joint venture, trust, estate, unincorporated organization, labor union, Governmental Authority or any other entity.

"Phase 1 Level Environmental Inspection" means an investigation conducted in conformance with and not exceeding the scope of the American Society for Testing and Materials (ASTM) Standard Practice E1527-05 (Phase I Environmental Site Assessment Process).

"*Project*" means the cogeneration facility with a nameplate rating of 100 megawatts located in Rumford, Maine, including boilers No. 6 and No. 7, turbine generator No. 4, backup boilers No. 3 and No. 5, and turbine generator No. 3.

"Project Employee" is defined in Section 5.19.

"Prudent Industry Practice" means any of the practices, methods, and activities generally accepted in the integrated manufacturing cogeneration industry in the United States as good practices applicable to cogeneration generating facilities of similar proximity to an industrial host, design, size and capacity as the Project, including any of the practices, methods or activities which, in the exercise of reasonable judgment by a prudent cogeneration generating facility operator in light of the facts known at the time the decision was made, would have been expected to accomplish the desired result at a reasonable cost consistent with good business practices, reliability, safety, expedition and Laws, including Environmental Laws and Laws relating to the protection of public health and safety. "Prudent Industry Practice" is not intended to be limited to the optimal practices, methods or acts to the exclusion of all others, but rather to the practices, methods and acts generally accepted in the cogeneration utility industry in the United States.

"Purchase Price" is defined in Section 2.6.1.

"Purchased Assets" is defined in Section 2.1.

"Quitclaim Deed" means the quitclaim deed to be delivered by RPC conveying to Buyer title to the Owned Real Property and, if applicable, Easement Rights, a form of which is attached as Exhibit 1.1(M).

"Real Property" is defined in Section 2.1(1).

"Real Property Lease" is defined in Section 3.8.2.

"*Reciprocal Easement Agreement*" means the reciprocal easement agreement to be executed and delivered by RPC and Buyer at the Closing, the summary terms of which are set forth on Exhibit 1.1(N).

"*Recovery Boiler Agreement*" means the Recovery Boiler Agreement to be executed and delivered by RPC and Buyer at the Closing, the summary terms of which are set forth on Exhibit 1.1(O).

"*ReEnergy Guaranty*" means the guaranty from ReEnergy Holdings LLC, a form of which is attached as <u>Exhibit 1.1(P)</u> and the scope of which is set forth in <u>Exhibit 1.1(P)</u>.

"*Release*" means any release, spill, emission, leaking, emptying, pumping, pouring, injection, deposit, disposal, discharge, dispersal, dumping, escaping, leaching, or migration of any Hazardous Material at, into or onto the Environment, including movement or migration through or in the air, soil, surface water or groundwater, whether sudden or non-sudden and whether accidental or non-accidental, or any release, emission or discharge as those terms are defined in any applicable Environmental Law.

"Remedial Action" means any response action, removal action, remedial action, corrective action, monitoring program, sampling program, investigation or other actions taken to (a) clean up, remove, remediate, treat, monitor, assess or evaluate Hazardous Materials in the Environment; (b) prevent or minimize a Release or threatened Release of Hazardous Materials so that they do not migrate or endanger or threaten to endanger, public health or welfare or the Environment; or (c) perform pre-remedial studies and investigations and post-remedial operation and maintenance activities.

"*Representatives*" means, with respect to any Person, that Person's accountants, employees, counsel, environmental consultants, contractors, financial advisors, and other agents or other representatives.

"Required Regulatory Approval" means a Buyer Required Regulatory Approval or a Seller Required Regulatory Approval.

"Restricted Activity" is defined in Section 5.14.2.1.

"RPC" is defined in the preamble to this Agreement.

"Secured Note" is defined in Section 2.6.1(c).

"Security Agreement" means the security agreement in which Buyer grants to RPC a security interest in the personal property included in the Purchased Assets as security for payment of the Secured Note and the performance of Buyer's obligations pursuant to the Energy Supply Agreement, to be executed and delivered by RPC and Buyer at the Closing.

"Seller" is defined in the preamble to this Agreement.

"Seller Group" is defined in Section 9.3.

"Seller's Current Use" means use of the Project for generation of electrical energy and/or steam in a manner consistent with the Selling Parties' operation of the Project.

"Selling Parties" is defined in the preamble to this Agreement.

"Selling Parties' Knowledge" means, with respect to a particular fact, circumstance or condition, the actual knowledge by an individual listed below of (1) that particular fact, circumstance or condition, or (2) underlying facts, circumstances or conditions that would reasonably be expected to put a person having comparable responsibilities to that particular individual on notice of that particular fact, circumstance or condition or that would reasonably be expected to cause a person having comparable responsibilities to that particular individual to make due inquiry or investigation by contacting the appropriate party: Gerald LeClaire, Mill Manager, Rumford Operations, John Fuller, General Manager, Production, Scott Alley, General Superintendent, Utilities, Tony Lyons, Director, Fiber Supply and Governmental Affairs, Michael Drapeau, Mill Controller, and Scott Reed, Environmental Manager.

"Seller Material Adverse Effect" means a violation, inaccuracy, breach, default, failure to comply, loss, effect, fact, agreement, arrangement, commitment, understanding or obligation, which, individually or collectively taken as a whole, has had or could reasonably be expected to have a material adverse effect or material adverse impact on (1) the business, financial conditions, assets (including the Purchased Assets (taken as a whole)), results of operations or prospects of the Selling Parties or NewPage Corporation; or (2) the ability of the Selling Parties or NewPage Corporation to perform their respective obligations under this Agreement or any Ancillary Agreements or to consummate the transactions contemplated in this Agreement or any Ancillary Agreement, and will exclude any effect resulting or arising from any (a) change in Law, (b) change in interest rates or general economic conditions in the United States, or (c) national or international political or social conditions, including the engagement by the United States in hostilities, whether or not pursuant to the declaration of a national emergency or war, or the occurrence of any military or terrorist attack upon the United States (other than attacks that directly impact the physical condition of the Purchased Assets), or any of its territories, possessions, or diplomatic or consular offices or upon any military installation, equipment or personnel of the United States.

"Seller Required Regulatory Approvals" means the approvals set forth on Exhibit 1.1(R).

"Shared Services Agreement" means the shared services agreement to be executed and delivered by RPC and Buyer at the Closing, the summary terms of which are set forth on Exhibit 1.1(S).

"Subdivision" is defined in Section 5.12.5.

"Survey" is defined in Section 5.12.1.

"Tax" means any foreign, federal, state or local income, gross receipts, occupation, environmental (including taxes under Section 59A of the Code), customs, duties, registration, alternative or add-on minimum, estimated, withholding, payroll, employment, unemployment insurance, social security (or similar), excise, sales, use, value-added, franchise, real property, personal property, business and occupation, capital stock, stamp or documentary, transfer, workers' compensation or other tax, governmental fee or imposition of any kind whatsoever, including any interest, penalties or additions.

"*Tax Return*" means a return, report, declaration, claim for refund or reimbursement, estimate, election, or information statement or bill relating to a Tax, including any related schedule or attachment and any related amendment or restatement.

"Third Party Claim" is defined in Section 9.4.2.

- "Title Commitment" is defined in Section 5.12.1.
- "Title Company" means Monument Title Company.
- "Title Defect" means a Lien arising from an imperfection in title to the Real Property or from real estate surveying matters encumbering the Real Property which is not a Permitted Lien, excluding title imperfections or surveying matters that do not (i) materially detract from or diminish the value of the Real Property, or (ii) materially interfere with Buyer's Intended Use.
 - "Title Policy" is defined in Section 5.12.4.
 - "Transferable Environmental Permits" is defined in Section 2.1(6).
 - "Transferable Permits" is defined in Section 2.1(5).
 - "Transferable Warranties" is defined in Section 2.1(2).
- "*Undivided Ownership Agreement*" means the undivided ownership agreement to be executed and delivered by RPC and Buyer at the Closing, the summary terms of which are set forth on Exhibit 1.1(T).
 - "Undivided Ownership Assets" is defined in Section 2.1(8).
 - 1.2 Interpretation. Unless otherwise expressly provided, for purposes of this Agreement, the following rules of interpretation apply:
- **1.2.1** <u>Calculation of Time Period</u>. When calculating any period of time pursuant to this Agreement, the reference date used in calculating that period will be excluded. If the last day of any time period is a non-Business Day, the time period will be extended to the next Business Day.
 - 1.2.2 Dollars. Any reference in this Agreement to "dollars" or "\$" means U.S. dollars.
- 1.2.3 <u>Exhibits and Schedules</u>. Unless otherwise expressly indicated, any reference in this Agreement to an "**Exhibit**" or a "**Schedule**" refers to an Exhibit or Schedule to this Agreement. The Exhibits and Schedules to this Agreement are incorporated and made a part of this Agreement as if set forth in full in this Agreement and are an integral part of this Agreement. Capitalized terms used but not otherwise defined in a Schedule or Exhibit are as defined in this Agreement.
- **1.2.4** <u>Gender and Number</u>. Any reference in this Agreement to gender includes all genders, and the meaning of defined terms applies to both the singular and the plural of those terms.
- **1.2.5** <u>Headings</u>. The division of this Agreement into Articles, Sections, and other subdivisions and the insertion of headings are for convenience of reference only and do not affect, and will not be utilized in construing or interpreting, this Agreement. All references in this Agreement to a "Section" are to the corresponding Section of this Agreement unless otherwise specified.
- **1.2.6** <u>Including</u>. The word "**including**" or any variation of that word means "including without limitation" and does not limit any general statement that it follows to the specific or similar items or matters immediately following it.
- **1.2.7** <u>Interpretation</u>. If any term or condition of this Agreement is found to be ambiguous, the ambiguity will not be construed against any one particular Party or in favor of any one particular Party, and such ambiguous language will be in all cases construed as a whole according to its fair and reasonable meaning.

1.2.8 <u>Delivery</u>. Other than Schedules, Exhibits and Ancillary Agreements, any item or document to be delivered to a Party pursuant to this Agreement shall be deemed to have been delivered whether provided in hard copy or electronic form, including those items made available to Buyer in an electronic data room.

ARTICLE 2 PURCHASE AND SALE OF ASSETS

- 2.1 <u>Purchase and Sale of Assets.</u> Subject to the terms and conditions of this Agreement, at the Closing, each Selling Party will sell, assign, convey, transfer and deliver to Buyer, or cause to be sold, assigned, conveyed, transferred and delivered to Buyer, and Buyer will purchase and acquire from each Selling Party, free and clear of all Liens except Permitted Liens, all of each Selling Party's right, title and interest in, to and under the real and personal property, tangible or intangible, described below, wherever located, as they exist at the Effective Time, except to the extent that they are Excluded Assets (collectively, "*Purchased Assets*"):
 - (1) The Owned Real Property, the Easement Rights and the other real property and real property interests set forth on <u>Exhibit</u> 2.1(1) (collectively, "*Real Property*");
 - (2) The Project and all machinery, equipment, tools, furniture, furnishings and other fixed assets listed on <u>Exhibit 2.1(2)</u> and all related express or implied warranties by manufacturers or vendors to the extent that those warranties are transferable without the consent of any Person ("*Transferable Warranties*");
 - (3) The Inventory identified on Exhibit 2.1(3)(A), together with all rights of the Selling Parties against suppliers of the Inventory;
 - (4) The Contracts set forth on Exhibit 2.1(4) (each an "Assumed Contract");
 - (5) Subject to <u>Section 2.5</u>, the Permits listed on <u>Exhibit 2.1(5)</u>, in each case to the extent they are assignable ("*Transferable Permits*");
 - (6) Subject to Section 2.5, the Environmental Permits listed on Exhibit 2.1(6), in each case to the extent assignable ("Transferable Environmental Permits");
 - (7) The Documents;
 - (8) Undivided ownership interests in the facilities and equipment required to interconnect the Project to the CMP distribution grid as described on Exhibit 2.1(8) ("Undivided Ownership Assets");
 - (9) Claims of the Selling Parties (whether known or unknown, matured or unmatured, accrued or contingent), including rights to returned or repossessed goods, in each case related solely to the Purchased Assets;
 - (10) Rights of recovery, rights of warranty and indemnity, in each case related solely to the Purchased Assets;

- (11) Prepaid expenses, security deposits, utility deposits and other deposits, in each case related solely to the Purchased Assets and only to the extent paid to the Selling Parties by Buyer pursuant to Section 2.6.1(b)(ii);
- (12) Use of any telephone numbers used solely in connection with the Project, but only to the extent those telephone numbers are a direct line to the Project and are not routed through RPC's telephone system; and
- (13) The assets described on Exhibit 2.1(13).
- **2.2** Excluded Assets. The Selling Parties will not sell, Buyer will not purchase, and the Purchased Assets will not include any property or assets of either Selling Party not described in Section 2.1 and, notwithstanding any provision to the contrary contained in Section 2.1 or elsewhere in this Agreement, the Purchased Assets do not include any of the following property or assets of either Selling Party (collectively, "Excluded Assets"):
 - (1) Cash, cash equivalents and bank deposits;
 - (2) Real property or real property interests of either Selling Party not described on Exhibit 2.1(1);
 - (3) Machinery, equipment, tools, furniture, furnishings or other fixed assets of either Selling Party not identified on Exhibit 2.1(2);
 - (4) Inventory of either Selling Party not identified on Exhibit 2.1(3)(A), including the Inventory identified on Exhibit 2.1(3)(B);
 - (5) Receivables or other rights to receive payments to the extent attributable to any period prior to the Effective Time, except to the extent that the Selling Parties have been paid by Buyer for those receivables pursuant to Section 2.6.1(b)(ii);
 - (6) Books, records, or the like other than the Documents;
 - (7) Assets that have been disposed of prior to the Effective Time without violation of this Agreement;
 - (8) Rights of either Selling Party arising under or in connection with this Agreement, any Ancillary Agreement, or any certificate or other document delivered in connection with this Agreement or any Ancillary Agreement, or any of the transactions contemplated by this Agreement or by any Ancillary Agreement;
 - (9) Certificates of deposit, shares of stock, securities, bonds, debentures, evidences of indebtedness, and any other debt or equity interest in any Person;
 - (10) The names "NewPage," "Rumford Cogeneration Company Limited Partnership," "Rumford Paper Company" and any derivative of any of the foregoing, and any related, similar, and other trade names (except for ReEnergy Rumford LLC), trademarks, service marks, and logos of either Selling Party and their Affiliates;
 - (11) Subject to the prorations in Section 2.7, any Tax refund or credit (a) related to Taxes paid or required to be paid by or on behalf of either Selling Party, whether received as a payment or as a credit against future Taxes payable, or (b) relating to a period before the Effective Time;

- (12) Except as expressly provided in <u>Section 2.1(9)</u>, Claims of either Selling Party against any Person;
- (13) Insurance policies and rights under insurance policies, including any insurance policies and rights in respect of the Purchased Assets or the Project (including any right to receive proceeds for Liabilities associated with the Purchased Assets, or otherwise); and
- (14) The other assets and rights described on Exhibit 2.2.
- 2.3 Assumed Liabilities. Buyer will assume and agree to perform and discharge all of the: (i) executory obligations and Liabilities of the Selling Parties arising from and after the Effective Time pursuant to the Assumed Contracts, Transferable Permits and Transferable Environmental Permits and validly assigned to Buyer pursuant to this Agreement, in each case in accordance with their respective terms and subject to their respective conditions, and in each case excluding any Liabilities arising from or relating to any breach or violation prior to the Effective Time of Assumed Contracts, Transferable Permits and Transferable Environmental Permits by the Selling Parties or default prior to the Effective Time under Assumed Contracts, Transferable Permits and Transferable Environmental Permits by the Selling Parties, and (ii) Liabilities for personal injury and property damage first occurring from and after the Effective Time in connection with the Buyer's ownership, possession, use or operation of the Purchased Assets (collectively, the "Assumed Liabilities"). Nothing contained in this Agreement shall require Buyer to pay, perform or satisfy any of the Assumed Liabilities so long as Buyer shall in good faith contest or cause to be contested the amount or validity thereof or shall in good faith assert any defense or offset thereto and the failure to pay, perform or satisfy any of the Assumed Liabilities does not adversely affect either Selling Party or result in the imposition of a Lien on their assets. Each Selling Party shall provide reasonable assistance to Buyer in so contesting or defending such claims.
- **2.4** Excluded Liabilities. Buyer shall not assume or become liable for any Liabilities of Seller that are not specifically an Assumed Liability, whether or not related to the Purchased Assets (collectively, "Excluded Liabilities").
- 2.5 Conditional Assignment. The Parties will use commercially reasonable efforts (without obligation by any Party to pay any money, assume or incur any liability or agree to any change in the terms of any Assumed Contract, Permit, Environmental Permit or Transferable Warranty related solely to a Purchased Asset) to secure any required consent or approval. If a required consent or approval is not obtained prior to Closing, each Selling Party, as applicable, will cooperate with Buyer following the Closing in any commercially reasonable arrangement (without obligation by the Selling Parties to pay any money, assume or incur any liability or agree to any change in the terms of any Assumed Contract, Permit, Environmental Permit or Transferable Warranty related solely to a Purchased Asset) designed to provide Buyer with the benefits of the Assumed Contract, Permit, Environmental Permit or manufacturer's warranty to the extent not assigned. Subject to the Selling Parties' obligations under Section 5.2.2 and this Section 2.5, if the Selling Parties are unable to secure a required consent or approval, nothing in this Agreement shall be deemed to obligate the Selling Parties to transfer any such Assumed Contract, Permit, Environmental Permit or Transferable Warranty.

2.6 Purchase Price and Payment.

2.6.1 <u>Purchase Price</u>. The purchase price ("**Purchase Price**") for the purchase of the Purchased Assets will be (a) Buyer's assumption of the Assumed Liabilities, (b) the sum of (i) \$30,500,000, plus (ii) any cash deposits with utility providers or other security deposits that are being transferred by either Selling Party to Buyer at the Closing, plus or minus (iii) an adjustment for the value of the Inventory on hand at the time of the Closing as determined in accordance with <u>Schedule 2.6.1</u> (collectively, the "**Cash Purchase Price**"), and (c) Buyer's delivery of a negotiable promissory note payable to the Selling Parties in the principal amount of \$30,500,000 together with simple interest at the rate of 8% per annum, payable in 10 equal annual principal installments together with accrued and unpaid interest as of the payment date, beginning on the first anniversary of the Closing Date, secured by the Purchased Assets, in the form of <u>Exhibit 2.6.1</u> (the "**Secured Note**").

2.6.2 <u>Payment.</u> Buyer will pay the Purchase Price at the Closing by (a) executing and delivering to the Selling Parties the Assignment and Assumption Agreement, (b) delivering to the Selling Parties the Cash Purchase Price by wire transfer of immediately available funds to an account designated by the Selling Parties at least two Business Days prior to the Closing Date, and (c) executing delivering to the Selling Parties the Secured Note.

2.7 Prorations and Expenses; Certain Tax Matters.

2.7.1 <u>Generally</u>. All Taxes, utility charges or assessments, and similar items customarily prorated, including those listed below, to the extent relating to the Purchased Assets, will be prorated as of the Effective Time, with Seller or RPC, as applicable, liable to the extent those items relate to any period prior to the Effective Time, and Buyer liable to the extent those items relate to any period from and after the Effective Time. To the extent that the Selling Parties determine in good faith that amounts to be prorated under this <u>Section 2.7</u> can be reasonably estimated at Closing, the Selling Parties will provide the estimate to Buyer at least 10 Business Days prior to Closing and the Parties will adjust the amounts paid at Closing to reflect the prorations. The items to be prorated will include (a) personal property and real property Taxes, assessments and other similar charges, including charges for water, telephone, electricity, and other utilities, (b) any permit, license, registration, compliance assurance fees or other fees with respect to any Transferable Permits and Transferable Environmental Permits, and (c) rents under any leases of personal property identified on <u>Exhibit 2.1(4)</u>.

2.7.2 Real Property Taxes.

2.7.2.1 <u>Generally</u>. Any Taxes on Real Property will be deemed to be paid on a calendar year basis in which the Taxes are first billed. Taxes on Real Property which are first billed in years prior to the year of Closing will be paid by the applicable Selling Party in full prior to Closing, without any proration. Taxes on Real Property which are first billed in the year of Closing will be prorated so that RPC will be charged with those Taxes from the first day of that year up to the Effective Time, and Buyer will be charged with those Taxes for the balance of the year. Installments of special assessments due and payable in years prior to Closing will be paid by RPC; installments due and payable for the year of Closing will be prorated between Buyer and RPC up to the Effective Time; and installments due and payable after the Effective Time will be Buyer's responsibility. All prorations will be based upon the most recent available Tax rates, assessments, valuations and/or other available and applicable information. Any prorations will be made so as to avoid duplication of any items.

2.7.2.2 <u>Split Tax Bills</u>. If real property Taxes for any Owned Real Property are assessed on a larger parcel of which the Owned Real Property is a part, then real property Taxes allocated to the Owned Real Property for purposes of making the foregoing prorations and any post-Closing

payments (until the taxing authority issues a separate Tax bill for the Owned Real Property) will be determined by separately calculating a prorated Tax amount for the land and for any improvements in the Owned Real Property and combining those two amounts as follows:

- (1) The prorated Tax amount for land in the Owned Real Property is the real property Taxes for the larger tax parcel multiplied by (i) the fraction obtained by dividing the number of square feet of land in the Owned Real Property by the number of square feet of land in the larger tax parcel, and further multiplied by (ii) the fraction obtained by dividing the assessed value of the land for the larger tax parcel by the total assessed value (land and improvements) for the larger tax parcel, and
- (2) The prorated Tax amount for improvements in the Owned Real Property is the real property Taxes for the larger tax parcel multiplied by (i) the fraction obtained by dividing the number of square feet in the buildings located on the Owned Real Property by the number of square feet in the buildings located on the larger tax parcel, and further multiplied by (ii) the fraction obtained by dividing the assessed value of the improvements for the larger tax parcel by the total assessed value (land and improvements) for the larger tax parcel.

Until the taxing authority divides the bill for real property Taxes between the Owned Real Property and the remainder of the larger tax parcel, RPC will be responsible for and pay directly to the taxing authority, prior to delinquency, the entire tax bill for the amount of unsplit tax parcels and promptly send evidence of such payment to the Buyer, along with RPC's calculation of Buyer's share of those tax bill(s) calculated pursuant to this Section 2.7.2.2. Buyer will then remit to RPC, within 30 days of Buyer's receipt of RPC's reimbursement calculation, Buyer's required contribution applicable to those tax bills.

- **2.7.3** <u>Post-Closing Adjustments</u>. The proration of all items under this <u>Section 2.7</u> will be recalculated by Buyer within 30 days following the date upon which the actual amounts become available to Buyer. Buyer will promptly notify RPC of the recalculated amounts, and will provide RPC with all documentation relating to the recalculations, including tax statements and other notices from third parties. The Parties will make such payments to each other as are necessary to reconcile any estimated amounts prorated as of the Effective Time with the final amounts to be prorated. RPC and Buyer agree to furnish each other with such documents and other records as may be reasonably requested in order to confirm all proration calculations made pursuant to this <u>Section 2.7</u>.
- **2.7.4** <u>Disagreements</u>. If RPC and Buyer disagree about any calculation pursuant to this <u>Section 2.7</u>, the Parties will first attempt to resolve the disagreement amicably. If RPC and Buyer cannot agree on a resolution within 30 days after the disagreement arises, either Party may submit the remaining differences to the CPA Firm, acting as an expert and not as an arbitrator, for resolution. The decision of the CPA Firm will be final and binding and its expenses will be shared equally by RPC and Buyer.
- **2.8 Purchase Price Allocation.** The allocation of the Purchase Price, liabilities assumed and other relevant items for purposes of Section 1060 of the Code will be determined jointly by the Parties in accordance with Exhibit 2.8. Buyer and the Selling Parties will (a) be bound by the allocation for purposes of determining any federal, state or local income Taxes, (b) prepare and file their Tax Returns, including IRS Form 8594, on a basis consistent with the allocation, and (c) take no position inconsistent with the allocation on any Tax Return or in any proceeding before any taxing authority. The obligations of Buyer and the Selling Parties under this Section 2.8 will survive the Closing.

ARTICLE 3 REPRESENTATIONS AND WARRANTIES OF THE SELLING PARTIES

3.1 Scope. As an inducement to Buyer to enter into this Agreement and to consummate the transactions contemplated by this Agreement, the Selling Parties, jointly and severally, hereby make the representations and warranties to Buyer set forth in this Article 3. Any information set forth in any Schedule will be deemed to modify or qualify only the corresponding section or paragraph in this Agreement.

3.2 Organization and Good Standing.

- **3.2.1** <u>Seller</u>. Seller is a limited partnership duly formed, validly existing and in good standing under the Laws of the State of Maine and has full power, right and authority to conduct its business as it is now being conducted, to own or use the properties and assets that it purports to own or use, and to perform all its obligations under the Assumed Contracts. Seller is duly qualified to do business and is in good standing as a foreign limited partnership in each state or other jurisdiction in which the character of the Purchased Assets, or the nature of the activities conducted by Seller, requires such qualification.
- **3.2.2** <u>RPC</u>. RPC is a corporation duly incorporated, validly existing and in good standing under the Laws of the State of Delaware and has full power, right and authority to conduct its business as it is now being conducted, to own or use the properties and assets that it purports to own or use, and to perform all its obligations under the Assumed Contracts. RPC is duly qualified to do business and is in good standing as a foreign corporation in each state or other jurisdiction in which the character of the Purchased Assets, or the nature of the activities conducted by RPC, requires such qualification.

3.3 Enforceability, Authority and No Conflict.

- 3.3.1 Enforceability and Authority. This Agreement and each Ancillary Agreement to which Seller or RPC is a party constitutes the legal, valid and binding obligation of Seller and RPC, as applicable, enforceable against Seller and RPC, as applicable, in accordance with its respective terms, except as the enforceability thereof may be limited by applicable bankruptcy, insolvency or other similar laws affecting or relating to enforcement of creditors' rights generally or general principles of equity. Each of Seller and RPC has the absolute and unrestricted right, power and authority to execute and deliver this Agreement and each Ancillary Agreement to which Seller or RPC is a party and to perform its respective obligations under this Agreement and each Ancillary Agreement to which Seller or RPC is a party. The execution, delivery and full performance by Seller and RPC of this Agreement and each Ancillary Agreement to which Seller or RPC is a party, and the consummation of the transactions contemplated by this Agreement and such Ancillary Agreements, have been duly and validly authorized and approved by all necessary action on the part of Seller and RPC.
- **3.3.2** <u>No Conflict</u>. Except as set forth in <u>Schedule 3.3.2</u>, neither the execution and delivery by Seller or RPC of this Agreement or any Ancillary Agreement to which the Seller or RPC is a party nor the consummation or performance of any of the transactions contemplated hereby or thereby will, directly or indirectly (with or without notice or lapse of time):
 - (1) Conflict with, breach or violate the Governing Documents of Seller or RPC or any resolution adopted by the partners, directors or shareholders of Seller or RPC, as applicable;

- (2) Conflict with, breach, violate or give any Governmental Authority or other Person the right to challenge any of the transactions contemplated by this Agreement or the Ancillary Agreements under any Law or any Order to which Seller, RPC, or the Purchased Assets may be subject;
- (3) Conflict with, breach or violate any of the terms or requirements of, or give any Governmental Authority the right to revoke, withdraw, suspend, cancel, terminate or modify, any Permit held by Seller or RPC that relates to the Purchased Assets:
- (4) Conflict with, breach any provision of, or give any Person the right to declare a default or exercise any remedy under, or to accelerate the maturity or performance of, or payment under, or to cancel, terminate or modify, any Assumed Contract;
- (5) Result in the imposition or creation of any Lien upon or with respect to the Purchased Assets;
- (6) Result in the acceleration of the maturity of any payment date of any of the Assumed Liabilities, or increase or adversely affect the liability or obligations of Buyer under any of the Assumed Liabilities; or
- (7) Result in any shareholder of RPC having the right to exercise dissenters' appraisal rights.
- 3.4 <u>Third Party Consents.</u> Schedule 3.4 describes all material authorizations, consents, waivers and approvals of any Person or Governmental Authority necessary in connection with the execution and delivery of this Agreement or the Ancillary Agreements by the Selling Parties and the Selling Parties' consummation and performance of the transactions contemplated by this Agreement or the Ancillary Agreements.
- 3.5 <u>Financial Records</u>. The books of account and other financial records of Seller and RPC relating to the Purchased Assets and described in <u>Schedule 3.5</u> ("*Financial Records*"), copies of which have been made available to Buyer, are complete and correct in all material respects.
- 3.6 <u>Sufficiency of Assets</u>. Taken together, the Purchased Assets and the Required Regulatory Approvals, the rights granted to Buyer under the Ancillary Agreements, the shared Inventory listed on <u>Exhibit 2.1(3)(B)</u> and the Excluded Assets constitute all of the material tangible and intangible assets required to operate the Purchased Assets substantially in accordance with Seller's Current Use. To the Selling Parties' Knowledge, the Purchased Assets necessary for Seller's Current Use have been maintained and operated in accordance with Prudent Industry Practice since May 2, 2005.
- **3.7** <u>Tangible Personal Property</u>. Except as set forth on <u>Schedule 3.7(A)</u>, Seller or RPC, as applicable, has good, valid and transferable title to all of the tangible personal property owned by Seller or RPC that forms a part of the Purchased Assets, free and clear of all Liens other than Permitted Liens. <u>Schedule 3.7(B)</u> sets forth a list of all leases for the tangible personal property leased by Seller or RPC that forms a part of the Purchased Assets.

3.8 Real Property.

3.8.1 <u>Owned Real Property</u>. <u>Schedule 3.8.1</u> sets forth a true, correct and complete legal description, street address and tax parcel identification number of all tracts, parcels and subdivided lots of the Real Property that Seller or RPC owns in fee simple (the "**Owned Real Property**"). Except as

set forth on Schedule 3.8.1, Seller or RPC, as applicable, owns good, valid and insurable fee simple title to all of the Owned Real Property, free and clear of all Liens except for Permitted Liens. True and complete copies of all deeds and an existing title insurance policy pertaining to or otherwise including the Owned Real Property have been delivered to Buyer.

3.8.2 <u>Leased Real Property.</u> The Lease and Access, Piping and Construction Easement by and between Seller and RPC dated October 21, 1987, as amended, is the only leasehold interest with respect to the Real Property (the "*Real Property Lease*").

3.8.3 *Generally*. Except as set forth on Schedule 3.8.3:

- (1) Neither Seller nor RPC has received written notice of violation of any building code or zoning or other local ordinance regarding the improvements on the Owned Real Property that remains uncured and to the Selling Parties' Knowledge there are no such material violations;
- (2) Neither Seller nor RPC has received written notice of any pending or threatened condemnation, action in eminent domain, planned public improvement, annexation, general or special assessment, zoning or subdivision change or other adverse claim affecting any portion of the Owned Real Property;
- (3) To the Selling Parties' Knowledge, there is no private restrictive covenant or governmental use restriction (including zoning) affecting any portion of the Owned Real Property that prohibits Seller's Current Use;
- (4) All utilities necessary for the use and operation of all Owned Real Property in accordance with Seller's Current Use are available to the Project;
- (5) To the actual knowledge of Gerald LeClaire, John Fuller, Tony Lyons or Scott Reed, neither Seller or RPC has in its possession any studies or reports that specifically detail any defects in the design or construction of any of the improvements on the Owned Real Property that could reasonably be expected to have a Seller Material Adverse Effect with respect to the Purchased Assets;
- (6) No Person, other than Buyer, has any right, option, right of first refusal or any other contract, whether oral or written, with respect to the purchase, lease, sublease or assignment, use or occupancy of the Owned Real Property. There are no leases or other agreements relating to occupancy of the Real Property, except for the Real Property Lease. No parties other than the Selling Parties are actually occupying any portion of the Real Property;
- (7) The Real Property is not separately assessed for real property tax assessment purposes. Neither of the Selling Parties has received written notice of any contemplated or actual re-assessments of any part of the Real Property for general real estate tax purposes. To the Selling Parties' Knowledge, no assessments for public improvements, impact fees or similar exactions have been made against the Real Property which are currently due and owing and remain unpaid;
- (8) Each Selling Party, as applicable, is, and at all times since January 1, 2009, has been, in compliance with each Law that is or was applicable to the Real Property where the failure to comply would result in a Seller Material Adverse Effect;

- (9) To the Selling Parties' Knowledge, (i) no event has occurred or circumstance exists that (with or without notice or lapse of time) may constitute or result in a violation by either Selling Party of, or (ii) a failure on the part of either Selling Party to comply with, any Law that is or was applicable to the Real Property, in each case, which alone or in the aggregate could reasonably be expected to have a Seller Material Adverse Effect; and
- (10) Neither Selling Party has received, at any time since January 1, 2009, written notice from any Governmental Authority or any other Person regarding any actual, alleged, possible or potential violation of, or failure to comply with, any Law that is or was applicable to the Real Property and that if left uncured would likely create a Seller Material Adverse Effect.
- 3.9 <u>Material Contracts</u>. <u>Schedule 3.9(A)</u> contains an accurate and complete list, and, except as set forth on <u>Schedule 3.9(A)</u>, the Selling Parties have delivered to Buyer accurate and complete copies, of each Contract by which the Purchased Assets are bound and that is material to the ownership and operation of the Purchased Assets in accordance with Seller's Current Use (a "*Material Contract*"). Except as set on Schedule 3.9(B):
 - Each Assumed Contract is in full force and effect and is valid and enforceable in accordance with its terms and will
 continue to be so on identical terms following the consummation of the transactions contemplated by this Agreement and
 the Ancillary Agreements;
 - (2) Each Assumed Contract may be assigned to Buyer without the consent of any other Person;
 - (3) Neither Seller nor RPC nor, to the Selling Parties' Knowledge, any other party thereto, is in breach of or default under any Assumed Contract;
 - (4) Neither Seller nor RPC has given to or received from any other Person, any written notice or other written communication regarding any actual, alleged or potential material violation or breach of, or material default under, a Assumed Contract; and
 - (5) To Selling Parties' Knowledge, no event has occurred or circumstance exists that (with or without notice or lapse of time) may contravene, conflict with, or result in a violation or breach of, or give either Seller or RPC or any other Person the right to declare a default or exercise any remedy under, or to accelerate the maturity or performance of, or to cancel, terminate, or modify, any Assumed Contract.
- **3.10** <u>Absence of Certain Changes and Events</u>. Except as set forth in <u>Schedule 3.10</u> and except as may otherwise be consented to by Buyer between the Execution Date and the Effective Time, since June 9, 2010, the Selling Parties have conducted their business with respect to the Purchased Assets only in the Ordinary Course of Business, and neither Selling Party has:
 - (1) Suffered or incurred any damage to or destruction or loss of any Purchased Asset in excess of \$50,000 in the aggregate, whether or not covered by insurance;
 - (2) Entered into any Material Contract that will remain in effect after the Effective Time or terminated or received notice of termination of any Assumed Contract;

- (3) Cancelled or waived any claims or rights with a value in excess of \$25,000 with respect to the Purchased Assets;
- (4) Suffered any Seller Material Adverse Effect; or
- (5) Taken any other action with respect to the Purchased Assets which is not either in the Ordinary Course of Business or provided for in this Agreement.

3.11 Environmental.

- **3.11.1** <u>Scope</u>. The only representations and warranties given with respect to Environmental Laws, Environmental Permits, Environmental Claims, Environmental Conditions, or other environmental matters are those contained in this <u>Section 3.11</u>, and no other representations or warranties contained in this Agreement will be deemed to constitute, directly or indirectly, a representation and warranty with respect to Environmental Laws, Environmental Permits, Environmental Claims, Environmental Conditions or other environmental matters.
- 3.11.2 <u>Violations of Environmental Law</u>. Except as set forth on <u>Schedule 3.11.2</u>, (a) to the Selling Parties' Knowledge, there is no uncured violation of any Environmental Law with respect to any of the Selling Parties' operations of the Purchased Assets that could reasonably be expected to have a Seller Material Adverse Effect, and (b) neither Selling Party has received any written notice from any Governmental Authority of violation or Liability under Environmental Laws, including any notice of any investigatory, remedial or corrective obligation, related to the Purchased Assets which remains uncured or outstanding as of the Execution Date. <u>Schedule 3.11.2</u> lists all written notices from any Governmental Authority of violation or Liability under Environmental Law with respect to the Selling Parties' operation of the Purchased Assets, whether or not cured or outstanding, received by the Selling Parties after January 1, 2009.
- 3.11.3 Environmental Permits. Except as set forth on Schedule 3.11.3, (a) the Selling Parties, collectively, possess all Environmental Permits necessary to operate the Project as it is currently being operated, (b) to the Selling Parties' Knowledge, the Selling Parties are in compliance in all material respects with the requirements of all Environmental Permits and Environmental Law, and (c) to the Selling Parties' Knowledge, the Selling Parties have not taken any action, or failed to take any action, that would likely to result in any Environmental Permit being modified, revoked, suspended, amended, not renewed, or not issued (as applicable).
- **3.11.4** Environmental Claims. Except as set forth on Schedule 3.11.4, neither Selling Party has received written notice from any Governmental Authority, prior owner of the Real Property, or current or prior owner of real property adjacent to the Real Property regarding any pending or threatened Environmental Claim relating to the Seller's or RPC's, as applicable, ownership and operation of the Purchased Assets, and to the Selling Parties' Knowledge, no such notification is pending or threatened.
- **3.11.5** <u>Releases</u>. Except as set forth on <u>Schedule 3.11.5</u>, neither Selling Party has received written notice from any Governmental Authority, prior owner of the Real Property, or current or prior owner of real property adjacent to the Real Property that it is a potentially responsible party under any Environmental Law, and remains a potentially responsible party as of the Execution Date, with respect to Releases of Hazardous Materials at the Real Property, and to the Selling Parties' Knowledge, no such notification is pending or threatened. Except as set forth on <u>Schedule 3.11.5</u>, to the Selling Parties' Knowledge, no Release has occurred at the Owned Real Property which is reasonably likely to give rise to an Environmental Claim against Buyer.

- 3.11.6 <u>Hazardous Materials</u>. Except as set forth on <u>Schedule 3.11.6(A)</u>, to the Selling Parties' Knowledge, there is no pollution or contamination of the Environment, or presence of Hazardous Materials on, in, at, under or migrating to or from any Real Property that will give rise to any Environmental Claim or Liability under any Environmental Laws. Except as set forth on <u>Schedule 3.11.6(B)</u>, to the Selling Parties' Knowledge, neither Selling Party has released, transported, arranged for the transportation of, or disposed of any Hazardous Materials in a manner that would reasonably be expected to result in an Environmental Claim under any Environmental Law.
- **3.11.7** <u>Documents</u>. To the Selling Parties' Knowledge, each Selling Party has delivered to Buyer true and complete copies of all material Documents that are in the Selling Parties' custody or control related to Environmental Conditions affecting the Purchased Assets.

3.12 Compliance with Law and Permits.

- **3.12.1** <u>Laws</u>. Except as set forth on <u>Schedule 3.12</u>, and excluding any representation or warranty regarding Environmental Laws or Environmental Permits, each of which is governed exclusively by <u>Section 3.11</u>, and further excluding any representation or warranty regarding Real Property, which is governed exclusively by <u>Section 3.8</u>:
 - (1) Each Selling Party, as applicable, is, and at all times since January 1, 2009, has been, in compliance with each Law that is or was applicable to the Purchased Assets where the failure to comply would result in a Seller Material Adverse Effect;
 - (2) To the Selling Parties' Knowledge, (i) no event has occurred or circumstance exists that (with or without notice or lapse of time) may constitute or result in a violation by either Selling Party of, or (ii) a failure on the part of either Selling Party to comply with, any Law that is or was applicable to the Purchased Assets, in each case, which alone or in the aggregate could reasonably be expected to have a Seller Material Adverse Effect; and
 - (3) Neither Selling Party has received, at any time since January 1, 2009, written notice from any Governmental Authority or any other Person regarding any actual, alleged, possible or potential violation of, or failure to comply with, any Law that is or was applicable to the Purchased Assets and that if left uncured would likely create a Seller Material Adverse Effect.
- **3.12.2** <u>Permits.</u> <u>Schedule 3.12.2</u> contains a complete and accurate list of each material Permit held by Seller or RPC that relates to the Purchased Assets. Each Permit listed in <u>Schedule 3.12.2</u> is valid and in full force and effect and the Permits listed in <u>Schedule 3.12.2</u> collectively constitute all of the Permits necessary to permit the Selling Parties to lawfully operate the Purchased Assets in accordance with Seller's Current Use. Except as set forth in <u>Schedule 3.12.2</u>:
 - (1) Each Selling Party is, and at all times since May 2, 2005 has been, in compliance with the terms and requirements of each Permit identified or required to be identified in <u>Schedule 3.12.2</u>, except for any defaults or noncompliance which have not had, or would not be reasonably expected to have had, a Seller Material Adverse Effect;
 - (2) To the Selling Parties' Knowledge, no event has occurred and continues or circumstance exists that may (with or without notice or lapse of time) result directly or indirectly in the revocation, withdrawal, suspension, cancellation or termination of, or any modification to, any Permit listed in Schedule 3.12.2; and

- (3) Neither Seller nor RPC has received, at any time since January 1, 2009, any written notice from any Governmental Authority or any other Person regarding (i) any actual, alleged, possible or potential violation of or failure to comply with any term or requirement of any Permit identified in Schedule 3.12.2 or (ii) any actual, proposed, possible or potential revocation, withdrawal, suspension, cancellation, termination of or modification to any Permit identified or required to be identified in Schedule 3.12.2.
- **3.13** <u>Litigation and Orders</u>. Except as described on <u>Schedule 3.13</u>, there are no Claims pending or, to the Selling Parties' Knowledge, threatened against or otherwise related to Seller, RPC, the Purchased Assets that could reasonably be expected to have, individually or in the aggregate, a Seller Material Adverse Effect. Except as described on <u>Schedule 3.13</u>, there is no Order by which either Selling Party is bound relating to the Purchased Assets or to which the Purchased Assets are subject.
- 3.14 <u>Taxes</u>. All Tax Returns relating to the Purchased Assets required to have been filed by either Selling Party, as applicable, have been timely filed and properly prepared, and all Tax Returns filed are complete and accurate. All Taxes with respect to the Purchased Assets (whether or not shown on filed Tax Returns) have been timely paid in full. Other than Permitted Liens, there are no Liens for Taxes upon any of the Purchased Assets. There are no audits or other administrative or judicial proceedings pending, or to the Selling Parties' Knowledge, threatened, by any state, local, or other jurisdiction alleging that either Selling Party, with respect to the Purchased Assets, has a duty to file Tax Returns and pay Taxes or is otherwise subject to the taxing authority of any jurisdiction where the Selling Party does not currently file a Tax Return, nor has either Selling Party received any notice or questionnaire from any such jurisdiction which suggests or asserts that either Selling Party, with respect to the Purchased Assets, may have a duty to file such Tax Returns and pay such Taxes.

3.15 Intellectual Property.

- **3.15.1** <u>Intellectual Property</u>. Except as set forth on <u>Schedule 3.15.1</u>, Seller or RPC owns or has a license to use the computer software and registered intellectual property rights used solely in connection with the Purchased Assets ("*Intellectual Property*"), including sufficient rights in each copy possessed by Seller or RPC, free and clear of all Liens other than Permitted Liens.
- 3.15.2 <u>Owned Intellectual Property</u>. <u>Schedule 3.15.2</u> lists and describes each item of Intellectual Property that is owned by the Selling Parties ("**Owned Intellectual Property**"), including as applicable the jurisdiction of grant, registration or pendency and the nature of the applicable Selling Party's interest. All Owned Intellectual Property is valid and enforceable and has been obtained and maintained in material compliance with Laws. Neither Seller nor RPC has distributed or disclosed any Owned Intellectual Property in a manner that diminishes the value or the Selling Parties' ownership rights in and to such Owned Intellectual Property.
- **3.15.3** <u>Licensed Intellectual Property</u>. <u>Schedule 3.15.3</u> lists and describes all Intellectual Property that is owned by a third party and licensed to Seller or RPC ("*Licensed Intellectual Property*"). Seller has not sold, distributed, granted access to or sublicenses to use any Licensed Intellectual Property without any required consent of the applicable third party owner. Except as set forth on <u>Schedule 3.15.3</u>, no royalties or fees are payable by either Selling Party to anyone for use of the Licensed Intellectual Property.

- **3.16** <u>Insurance</u>. The Selling Parties or their Affiliates maintain the insurance policies listed at <u>Schedule 3.16</u>, all of which are in full force and effect.
- 3.17 <u>Labor Matters</u>. Except as set forth in <u>Schedule 3.17</u>, (i) there is no labor strike, slowdown or stoppage occurring or, to Selling Parties' Knowledge, threatened against the Selling Parties involving any of the Project Employees, and (ii) to the Selling Parties' Knowledge, and except for the parties to the existing collective bargaining agreements, the Selling Parties have not been approached by any individual or organization claiming or seeking to represent any of the Project Employees, and no individual or organization is seeking or claiming to represent the Project Employees. True and correct copies of the collective bargaining agreements that are currently in effect with respect to any of the Project Employees have been provided to Buyer (the "*Collective Bargaining Agreements*"). Except as set forth in <u>Schedule 3.17</u>, no other collective bargaining agreement is currently in effect or being negotiated with respect to any of the Project Employees. There has been no strike, slowdown, work stoppage with respect to the Project Employees since January 1, 2009.
- **3.18** <u>No Commissions</u>. No commissions or brokers' or finders' fees are payable by, through or on account of any acts of either Selling Party in connection with this Agreement or the transactions contemplated by this Agreement.
- 3.19 <u>Disclaimer</u>. EXCEPT AS EXPRESSLY SET FORTH IN THIS <u>ARTICLE 3</u> OR IN ANY ANCILLARY AGREEMENT, BUYER IS ACQUIRING THE PURCHASED ASSETS "AS IS, WHERE IS" AND "WITH ALL FAULTS" AND NEITHER SELLER NOR RPC HAS MADE AND DOES NOT MAKE ANY REPRESENTATION OR WARRANTY OF ANY NATURE, INCLUDING REPRESENTATIONS OR WARRANTIES AS TO THE PURCHASED ASSETS (INCLUDING ANY WARRANTY OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE), OR ANY REPRESENTATION OR WARRANTY AS TO THE VALUE, CONDITION, NATURE, CAPABILITY, INCLUDING ELECTRICAL OUTPUT OR STEAM OUTPUT CAPABILITY, OF ANY PURCHASED ASSET, AND BUYER BY THIS AGREEMENT EXPRESSLY ACKNOWLEDGES THAT NO SUCH OTHER REPRESENTATIONS OR WARRANTIES HAVE BEEN MADE BY SELLER OR RPC OR RELIED UPON BY BUYER. NEITHER SELLING PARTY IS LIABLE OR BOUND IN ANY MANNER BY ANY EXPRESSED OR IMPLIED WARRANTIES, GUARANTIES, PROMISES, STATEMENTS, REPRESENTATIONS OR INFORMATION PERTAINING TO THE PURCHASED ASSETS OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT OR IN ANY ANCILLARY AGREEMENT MADE OR FURNISHED BY ANY AGENT, EMPLOYEE, SERVANT OR OTHER PERSON REPRESENTING OR PURPORTING TO REPRESENT EITHER SELLING PARTY, UNLESS THE WARRANTIES, GUARANTIES, PROMISES, STATEMENTS, REPRESENTATIONS OR INFORMATION ARE EXPRESSLY AND SPECIFICALLY SET FORTH IN THIS <u>ARTICLE 3</u> OR IN ANY ANCILLARY AGREEMENT.

ARTICLE 4 REPRESENTATIONS AND WARRANTIES OF BUYER

- **4.1** <u>Scope.</u> As an inducement to the Selling Parties to enter into this Agreement and to consummate the transactions contemplated by this Agreement, Buyer makes the representations and warranties to the Selling Parties set forth in this <u>Article 4</u>.
- **4.2** Organization and Good Standing. Buyer is a limited liability company duly formed, validly existing and in good standing under the Laws of the State of Delaware and has full power, right and authority to conduct its business as it is now being conducted, to own or use the properties and assets that it purports to own or use. Buyer is duly qualified to do business and is in good standing as a foreign limited liability company in each state or other jurisdiction in which the nature and activities conducted by Buyer requires such qualification.

4.3 Enforceability, Authority and No Conflict.

- **4.3.1** Enforceability and Authority. This Agreement and each Ancillary Agreement to which Buyer is a party constitutes the legal, valid and binding obligation of Buyer, enforceable against Buyer in accordance with its respective terms, except as the enforceability thereof may be limited by applicable bankruptcy, insolvency or other similar laws affecting or relating to enforcement of creditors' rights generally or general principles of equity. Buyer has the absolute and unrestricted right, power and authority to execute and deliver this Agreement and each Ancillary Agreement to which Buyer is a party and to perform its obligations under this Agreement and each Ancillary Agreement to which Buyer is a party. The execution, delivery and full performance by Buyer of this Agreement and each Ancillary Agreement to which Buyer is a party, and the consummation of the transactions contemplated by this Agreement and such Ancillary Agreements, have been duly and validly authorized and approved by all necessary action on the part of Buyer.
- **4.3.2** <u>No Conflict</u>. Except as set forth on <u>Schedule 4.3.2</u>, neither the execution and delivery by Buyer of this Agreement or any Ancillary Agreement to which Buyer is a party nor the consummation or performance of any of the transactions contemplated hereby or thereby will, directly or indirectly (with or without notice or lapse of time):
 - (1) Conflict with, breach or violate the Governing Documents of Buyer or any resolution adopted by the members or managers of Buyer;
 - (2) Conflict with, breach, violate or give any Governmental Authority or other Person the right to challenge any of the transactions contemplated by this Agreement or the Ancillary Agreements under any Law or any Order to which Buyer or any of its assets may be subject; or
 - (3) Prevent the Closing or Buyer's performance of this Agreement or any other Ancillary Agreement to which Buyer is a party.
- **4.4** Third Party Consents. Schedule 4.4 describes all authorizations, consents, waivers and approvals of any Person or Governmental Authority necessary in connection with the execution and delivery of this Agreement or the Ancillary Agreements by Buyer and the Buyer's consummation and performance of the transactions contemplated by this Agreement or the Ancillary Agreements.
- **4.5** <u>Litigation and Orders</u>. There are no Claims or Orders pending or, to Buyer's Knowledge, threatened against or otherwise related to Buyer, which may impair the ability of Buyer to perform this Agreement, the Ancillary Agreements or any other agreement to be entered into at the Closing or which questions the validity or propriety of this Agreement, the Ancillary Agreements, or any such other agreement or of any action taken under this Agreement, the Ancillary Agreements, or any such other agreement in connection with this Agreement.
- **4.6 <u>Due Diligence</u>.** Buyer believes in good faith that it is capable of satisfying its obligations under this Agreement and the other agreements, instruments and documents contemplated by this Agreement or the Ancillary Agreements, and is capable of consummating these transactions. The Selling Parties have provided Buyer with such access to the personnel, properties, premises and, to the records of each Selling Party. In negotiating and entering into this Agreement and agreeing to consummate the transactions contemplated by this Agreement, Buyer has relied solely upon its own investigation and the

express representations and warranties of the Selling Parties set forth in Article 3 or in any Ancillary Agreement, and neither Selling Party nor any of their respective officers, directors, shareholders, employees, Affiliates, agents or other Representatives has made any representation or warranty as to the Selling Parties, the Purchased Assets, this Agreement, the Ancillary Agreements or the Project, except as expressly set forth in this Agreement or in any Ancillary Agreement. To the fullest extent permitted by Law, none of Selling Party's officers, directors, shareholders, employees, Affiliates, agents or other Representatives will have any liability to Buyer for any information made available to, or statements made to, Buyer (or any of Buyer's agents, officers, directors, employees, Affiliates or other Representatives), except as otherwise expressly set forth in this Agreement or in any Ancillary Agreement.

- **4.7** <u>Availability of Funds</u>. Buyer has sufficient funds available to pay the Cash Purchase Price in immediately available funds at the Closing to consummate the transactions contemplated by this Agreement.
- **4.8** <u>No Commissions</u>. No commissions or brokers' or finders' fees are payable by, through or on account of any acts of Buyer or its Representatives in connection with this Agreement or the transactions contemplated by this Agreement.
- **4.9** No Buyer Material Adverse Effect. Except as set forth in Schedule 4.9 and except as may otherwise be consented to by RPC between the Execution Date and the Effective Time, since June 9, 2010, Buyer has not suffered a Buyer Material Adverse Effect.
- 4.10 <u>Disclaimer</u>. EXCEPT AS EXPRESSLY SET FORTH IN THIS <u>ARTICLE 4</u> OR IN ANY ANCILLARY AGREEMENT, BUYER HAS NOT MADE AND DOES NOT MAKE ANY REPRESENTATION OR WARRANTY OF ANY NATURE, AND SELLER AND RPC BY THIS AGREEMENT EXPRESSLY ACKNOWLEDGE THAT NO SUCH OTHER REPRESENTATIONS OR WARRANTIES HAVE BEEN MADE BY BUYER OR RELIED UPON BY SELLER OR RPC. BUYER IS NOT LIABLE OR BOUND IN ANY MANNER BY ANY EXPRESSED OR IMPLIED WARRANTIES, GUARANTIES, PROMISES, STATEMENTS, REPRESENTATIONS OR INFORMATION PERTAINING TO THE TRANSACTIONS CONTEMPLATED IN THIS AGREEMENT OR IN THE ANCILLARY AGREEMENTS MADE OR FURNISHED BY ANY AGENT, EMPLOYEE, SERVANT OR OTHER PERSON REPRESENTING OR PURPORTING TO REPRESENT BUYER, UNLESS THE WARRANTIES, GUARANTIES, PROMISES, STATEMENTS, REPRESENTATIONS OR INFORMATION ARE EXPRESSLY AND SPECIFICALLY SET FORTH IN THIS <u>ARTICLE 4</u> OR IN ANY ANCILLARY AGREEMENT.

ARTICLE 5 COVENANTS

- **5.1** Conduct of Business. Except as otherwise contemplated by this Agreement, as required by Law or as may be approved by Buyer, during the period from the Execution Date to the earlier of the Effective Time or the termination of this Agreement in accordance with Article 8, Seller and RPC, each on its own behalf, will operate the Project and maintain the Purchased Assets it owns, in each case in the Ordinary Course of Business, consistent with Prudent Industry Practice, in accordance with Laws and Environmental Permits, and will not, without Buyer's consent:
 - (1) Sell or transfer any of the Purchased Assets other than Inventory comprising a part of the Purchased Assets and then only with respect to such Inventory if such sale or transfer is in the Ordinary Course of Business, provided, however, the

- Selling Parties may, without Buyer's consent but with prior notice to Buyer, convey or otherwise cause the Purchased Assets owned by Seller to be transferred, by written instruments of transfer, to RPC and upon such transfer, RPC will assume and perform Seller's obligations under this Agreement with respect to such Purchased Assets;
- (2) Amend, modify, cancel, or waive any rights under or grant any consent relating to any Assumed Contracts, Transferable Permits or Transferable Environmental Permits, except in each case as specifically provided in this Agreement, or enter into any material contract that extends beyond the Effective Time relating to the Purchased Assets or the Project;
- (3) Incur any material obligations or Liabilities not in existence on the Execution Date that will become an Assumed Liability or otherwise become the obligation of Buyer in connection with the acquisition of the Purchased Assets;
- (4) Permit, allow, or suffer any Purchased Asset to be subjected to any Liens other than Permitted Liens or Liens that will otherwise be discharged prior to or at the Closing;
- (5) Grant, issue or give any Person any rights or options with respect to the Purchased Assets;
- (6) Settle any Tax claims, make or change any Tax elections or file any amended Tax Returns if such action would increase the Tax liability of the Buyer;
- (7) Take any affirmative action, or fail to take any reasonable action, as a result of which any of the events or changes listed in Section 3.10 is likely to occur; or
- (8) Enter into any written or oral agreement or obligation to do any of the foregoing.

5.2 Consents, Approvals and Contacts.

5.2.1 HSR Act. The Parties will each file or cause to be filed with the Federal Trade Commission and the United States Department of Justice, Antitrust Division any notifications required to be filed under the HSR Act with respect to the transactions contemplated by this Agreement. The Parties will consult and cooperate with each other as to the appropriate time of filing such notifications and will (a) make any such required filings at the agreed upon time (but in no event later than 45 days after the Execution Date), (b) respond promptly to any requests for additional information made by either of such agencies, and (c) use their commercially reasonable efforts to cause the waiting periods under the HSR Act to terminate or expire at the earliest possible date after the date of such filings. As used in the Section 5.2.1, "commercially reasonable efforts" shall not include the willingness of the Selling Parties or the Buyer to accept an Order (1) agreeing to the divestiture, or the holding separate, of any assets of Selling Parties or Buyer or (2) that restricts, limits or impairs in any material way the business of Buyer or either Selling Party or any of their Affiliates. The Buyer and the Selling Parties shall work together in good faith to direct and shall afford each other a reasonable opportunity to participate in any proceedings or negotiations with any Governmental Authority relating to any of the foregoing, provided that the Buyer and the Selling Parties shall not knowingly take any action or fail to take any action that, individually or in the aggregate, adversely affects such proceedings or negotiations.

5.2.2 Other Required Regulatory Approvals. Without limiting Section 5.2.1, each Selling Party and Buyer will cooperate with each other and use commercially reasonable efforts to (a) promptly prepare and file all necessary applications, notices, petitions, and filings, and execute all agreements and documents to the extent required by Law or Order for consummation of the transactions contemplated by this Agreement (including all Required Regulatory Approvals, except for the matters covered by Section 5.2.1, which will be governed by that Section), (b) obtain the transfer to Buyer of all Transferable Permits and Transferable Environmental Permits, and the reissuance to Buyer of all Permits that are not Transferable Permits and all Environmental Permits that are not Transferable Environmental Permits, (c) obtain the consents, approvals, and authorizations of all Governmental Authorities to the extent required by Law or Order for consummation of the transactions contemplated by this Agreement (including all Required Regulatory Approvals, except for the matters covered by Section 5.2.1, which will be governed by that Section), and (d) obtain all consents, approvals, and authorizations of all other Persons to the extent necessary to consummate the transactions contemplated by this Agreement as required by the terms of any note, bond, mortgage, indenture, deed of trust, license, franchise, permit, concession, Contract, lease, or other instrument to which a Selling Party or Buyer is a party or by which either of them is bound. Each Selling Party and Buyer each will have the right to review in advance all characterizations of the information relating to it or the transactions contemplated by this Agreement which appear in any filing made by the other in connection with the transactions contemplated by this Agreement.

5.2.3 Governmental Authority Contacts. Except (i) as expressly provided in Section 5.2.1, (ii) as permitted in this Section 5.2.3, or (iii) with respect to any communications by Buyer pertaining to a Buyer Regulatory Approval (subject to the restrictions contained in Section 5.14.1) or by either Selling Party pertaining to a Seller Required Regulatory Approval, neither Party will on an ex parte basis initiate, directly or indirectly, any communications, meetings, or other contacts with any Governmental Authority in connection with the transactions contemplated by this Agreement without in each case the prior consent of the other Party. Except as expressly provided in Section 5.2.1, or in connection with any communications, meetings, or other contacts, formal or informal, oral or written, with any Governmental Authority in connection with the transactions contemplated by this Agreement or any Required Regulatory Approval or any matters otherwise covered by Section 5.2.2, each Party will exercise commercially reasonable efforts to (a) inform the other in advance of any communication, meeting, or other contact which that Party proposes or intends to make, including the subject matter, contents, intended agenda, and other aspects of transactions contemplated by this Agreement, (b) consult and cooperate with the other Party and take into account the comments of such other Party, (c) arrange for Representatives of the other Party to participate to the extent reasonably possible, and (d) notify the other Party of any substantive oral or written communications with any Governmental Authority relating to any of the foregoing. Each Party further agrees to provide copies of all written communications with any Governmental Authority in connection with any Required Regulatory Approval. Nothing in this Section 5.2.3 is intended to or will restrict any communication or other action by the Selling Parties in connection with the Purchased Assets in the Ordinary Course of Business, or by Buyer in the Ordinary Course of Business, so long as those communications and actions do not relate to this Agreement, the Ancillary Agreements, the transactions contemplated in this Agreement, or (in Buyer's case) the Purchased Assets, For the avoidance of doubt, neither Party, nor their respective affiliates nor any other Person at the request or on behalf of a Party shall, without the prior consent of the other Party, initiate, directly or indirectly, any communications, meetings or other contacts, formal or informal, with any Governmental Authority regarding the applicability of sales tax in the State of Maine to the transactions contemplated by this Agreement or the Ancillary Agreements.

5.2.4 <u>Regulatory Matters</u>. Buyer and the Selling Parties will take all commercially reasonably necessary steps to avoid or eliminate every impediment under any applicable Law that is asserted by any Governmental Authority with respect to the transactions contemplated by this Agreement so as to enable the consummation of those transactions to occur as expeditiously as possible, but neither Buyer nor any Selling Party nor any of their respective Affiliates will be under any obligation to enter into negotiations, provide information, make proposals, enter into or perform agreements or submit to Orders, or, pursuant to any such agreement or Order or otherwise, sell or otherwise dispose of, or hold separate (through the establishment of a trust or otherwise), particular assets or categories of assets of Buyer, either Selling Party or any of their respective Affiliates, including any of the Purchased Assets.

5.3 Access and Confidential Information.

5.3.1 Pre-Closing Access. Subject to Section 5.14.1, between the Execution Date and the Effective Time, each Selling Party will, during ordinary business hours and upon reasonable notice, (a) give Buyer and Buyer's Representatives reasonable access to the Purchased Assets, including on-site technical and engineering data, to which Buyer is not denied access by Law and to which the Selling Party has the right to grant access without the consent of any other Person (and where consent of another Person is required, subject to any terms and conditions imposed by that other Person), (b) permit Buyer to make such inspections of the Purchased Assets as Buyer may reasonably request on reasonable advance notice to each Selling Party, so long as Buyer's access does not unreasonably interfere with any of Seller's or RPC's operations, (c) furnish Buyer with such financial and operating data and other information with respect to the Purchased Assets as Buyer may from time to time reasonably request, and (d) furnish Buyer with a copy of each material report, schedule, or other document principally relating to the Purchased Assets filed by Seller or RPC with, or received by Seller or RPC from, any Governmental Authority, Any such investigation will be conducted in such a manner as not to interfere unreasonably with the operation of the Project or any other Person, and Buyer will indemnify and hold harmless the Seller Group (separate and apart from Buyer's obligations in Article 9 and without regard to any basket, cap or other conditions or limitations contained in Article 9) from and against any Liabilities caused by any action of Buyer or Buyer's Representatives or Affiliates while present on any of the Purchased Assets or other premises to which Buyer is granted access under this Agreement (including restoring the premises to substantially the same condition as prior to the investigation). Neither Selling Party will be required to take any action which would constitute a waiver of the attorney-client privilege, and neither Selling Party need supply Buyer with any information which such Selling Party is under a contractual or other legal obligation not to supply, but a Selling Party will provide Buyer with a description of the information withheld and the basis for withholding that information to the fullest extent possible without causing a waiver of the attorney-client privilege or a violation of a contractual or legal obligation. Notwithstanding anything in this Section 5.3 to the contrary, Buyer will not have access to personnel and medical records that could, in either Selling Party's good faith judgment, subject either Selling Party or its Affiliates to risk of liability or otherwise violate the Health Insurance Portability and Accountability Act of 1996, as amended.

5.3.2 <u>Confidential Information</u>. Unless and until the transactions contemplated by this Agreement have been consummated, Buyer will, and will cause its Affiliates and Buyer's Representatives to, hold in strict confidence and not use or disclose to any other Person any Confidential Information, except as and to the extent specifically authorized in the Confidentiality Agreement. If the transactions contemplated by this Agreement are consummated, (a) the Selling Parties will, and will cause their respective employees, officers, directors and Affiliates to, keep confidential all information (whether written or otherwise) concerning the Purchased Assets and Buyer's business ("**Buyer Confidential Information**") and will refrain from using any of the Buyer Confidential Information and shall not disclose Buyer Confidential Information to any third party except on behalf of the Buyer and its Affiliates and as necessary to prepare and file Tax Returns, and (b) the Confidentiality Agreement will no longer

apply to Buyer Confidential Information but will continue in full force and effect with respect to all other Confidential Information of either Selling Party or its Affiliates. If the Selling Parties are required to disclose Buyer Confidential Information to comply with any applicable Law (other than as necessary to prepare and file Tax Returns), to the extent allowed by applicable Law, prior to that disclosure, the Selling Parties must give the Buyer prompt notice of the requirement or request for disclosure and, to the extent the Selling Parties' or their Affiliates' legal rights are not prejudiced thereby or the Selling Parties' or their Affiliates' legal obligations are not impeded thereby, use reasonable efforts to secure or cooperate with the Buyer, in each case at Buyer's request and expense, in securing a protective Order or other arrangement to limit disclosure of Buyer Confidential Information only to parties agreeing to be bound by the terms of the protective Order or other arrangement. The Selling Parties' obligations of confidentiality and restriction on use in this Section 5.3.2 do not apply to any Buyer Confidential Information that (1) was in the public domain prior to the Execution Date or subsequently came into the public domain through no fault of the Selling Parties, or (2) was lawfully received by either Selling Party from a third party for which either Selling Party had no reason to know was restricted from disclosing the Buyer Confidential Information, or (3) is independently developed by either Selling Party without reference to the Buyer Confidential Information.

5.3.3 <u>Post-Closing Access</u>. For seven years after the Effective Time, each Party and its Representatives will have reasonable access to all of the books and records relating to the Purchased Assets in the possession of the other Party to the extent that access may reasonably be required in connection with the Assumed Liabilities or the Excluded Liabilities or other matters relating to or affected by the operation of the Purchased Assets. Such access will be afforded during normal business hours upon reasonable advance written notice to the Party in possession of the books and records in question, but (a) any review of books and records will be conducted in such a manner as not to interfere unreasonably with the operation of the business of any Party or its Affiliates, (b) no Party will be required to take any action which would constitute a waiver of the attorney-client privilege or a violation of applicable Law, (c) no Party need supply any information which that Party is under a contractual or other legal obligation to keep confidential or not to supply or disclose, and (d) if appropriate or desirable, access under this Section 5.3.3 may be augmented subject to a mutually-agreeable written confidentially arrangement that becomes effective post-Closing. A Party validly exercising its rights of access under this Section 5.3.3 will do so solely at its own cost and, in addition, will reimburse the Party against whom the right is validly exercised for reasonable out-of-pocket costs incurred in satisfying requests for information, including costs for copying, non-incidental labor and travel. If the Party in possession of books and records desires to dispose of any of those books and records prior to the expiration of the seven-year period following the Effective Time, that Party will, prior to disposition, give the other Party a reasonable opportunity at the other Party's expense to segregate and take possession of those books and records.

5.4 Expenses. Except to the extent specifically provided in this Agreement, and irrespective of whether the transactions contemplated by this Agreement are consummated, all costs and expenses incurred in connection with this Agreement and the transactions contemplated by this Agreement will be borne by the Party incurring those costs and expenses, including specifically: (a) each Party will pay its own filing fee associated with its own HSR approval, if any, and (b) all filing, recording, transfer or other fees or charges of any nature payable pursuant to any Law or any Order in connection with the sale, transfer, and assignment of the Purchased Assets and the Assumed Liabilities, if any, will, except to the extent provided in Section 5.7, be borne by Buyer.

5.5 Efforts to Complete the Transactions. Subject to the terms and conditions of this Agreement, each Party will use commercially reasonable efforts to take, or cause to be taken, all action, and to do, or cause to be done, all things necessary, proper, or advisable to consummate and make effective the transactions contemplated by this Agreement, including negotiating in good faith the form

and substance of each Ancillary Agreement and using commercially reasonable efforts to obtain satisfaction of the conditions precedent to each Party's obligations under this Agreement within its reasonable control. Neither Party will, without the prior consent of the other Party, take any action which would reasonably be expected to prevent or materially impede, interfere with or delay the transactions contemplated by this Agreement.

5.6 <u>Publicity.</u> Neither Buyer nor the Selling Parties will, without the prior consent of the other Party, make any public disclosure (by means of public announcement, press release or otherwise) concerning the subject matter of this Agreement or any of the transactions contemplated by this Agreement, unless otherwise required by Law. Any public disclosure is subject to <u>Section 5.3.2</u>.

5.7 Tax Matters.

5.7.1 <u>Taxes</u>. All state and local sales and use Taxes incurred in connection with this Agreement and the transactions contemplated by this Agreement, if any, will be paid one half by the Selling Parties and one half by Buyer. All other Real Property transfer, state and local documentary, stamp and registration Taxes incurred in connection with this Agreement and the transactions contemplated by this Agreement, if any, will be paid one-half by Buyer and one-half by the Selling Parties, including transfer taxes related to the conveyance of the Real Property. Buyer and each Selling Party will prepare at their own expense and file, to the extent required by Law, all necessary Tax Returns and other documentation with respect to all such transfer or sales and use Taxes. Neither Buyer nor any Selling Party will take any position on any Tax Return, including any amendment or refund claim, or before any Tax authority that is inconsistent with the application of the occasional sale exemption provided to the transactions contemplated by this Agreement, without the express written consent of RPC. Any refund of sales and use Tax described in this <u>Section 5.7.1</u> will be shared one half by the Selling Parties and one half by Buyer. Buyer will promptly pay to that Selling Party one half of the amount of any such refund (including any credit which reduces the amount of Taxes Buyer would otherwise owe) promptly upon receipt or when such credit is issued.

5.7.2 Tax Returns. Each Selling Party will be responsible for the preparation and timely filing of any Tax Returns reflecting any Taxes payable by such Selling Party (including any federal, state or local income and franchise Taxes imposed on the Selling Parties or their Affiliates in connection with this Agreement and the transactions contemplated by this Agreement) and the timely payment of any Taxes shown to be due on such returns. Buyer will be responsible for the preparation and timely filing of any Tax Returns reflecting any Taxes payable by Buyer and the timely payment of any Taxes shown to be due on such returns. Any Tax Return that reflects Taxes to be prorated in accordance with Section 2.7 will be subject to the approval of the Party not preparing the Tax Return. Each Party will make any such Tax Return prepared by it available for the other Party's review and approval no later than twenty (20) Business Days prior to the due date for filing such Tax Return. Within fifteen (15) Business Days after receipt of such Tax Return, the approving Party will pay to the Party preparing the Tax Return the amount of such prorated Taxes shown as due on such approved Tax Return for which the approving Party is responsible under Section 2.7.

5.7.3 <u>Cooperation and Assistance</u>. Buyer and the Selling Parties will provide each other with such assistance as may reasonably be requested by the other Party in connection with the preparation of any Tax Return, any audit or other examination by any taxing authority, or any judicial or administrative proceedings relating to liability for Taxes, and each Party will retain and provide the other with any records or information which may be relevant to such return, audit or examination or proceedings. Any information obtained pursuant to this <u>Section 5.7.3</u> or pursuant to any other Section of this Agreement providing for the sharing of information in connection with the preparation of, or the review of, any Tax Return or other schedule relating to Taxes will be kept confidential in accordance with <u>Section 5.3.2</u>. The terms and conditions of this <u>Section 5.7</u> will survive the Closing.

- **5.7.4** <u>Claims and Disputes</u>. Each Party will timely notify the other Party of any Tax examination, refund claim, dispute, or proceeding involving Taxes subject to <u>Section 2.7</u> or <u>5.7.1</u> and allow the other Party to participate in the resolution of that action (at each Party's own expense) and will not settle such action without the other Party's consent. Each Party will timely notify the other Party of any Tax examination, refund, claim, dispute, or proceeding involving Taxes relating to the allocation of Purchase Price provided for in Section 2.8, but in no event will Buyer have access to any federal, state or local income Tax Returns of the Selling Parties or their Affiliates.
- **5.7.5** <u>Tax Clearance Certificates and Cooperation</u>. Buyer and each Selling Party on behalf of itself, will upon request use commercially reasonable efforts to obtain any certificate or other document from any Governmental Authority as may be necessary to mitigate, reduce, or eliminate any Tax that could be imposed on the Purchased Assets (including Taxes imposed on the transactions contemplated by this Agreement or by way of successor liability).

5.8 Notice of Changes.

- **5.8.1** By the Selling Parties. From the Execution Date to the Closing Date, the Selling Parties will promptly notify Buyer of (i) any Seller Material Adverse Effect, (ii) the institution of or, to the Selling Parties' Knowledge, threat of institution of, any Claim against either Selling Party or any of the Purchased Assets that could reasonably be expected to have a Seller Material Adverse Effect, or (iii) any breach of this Agreement by a Selling Party.
- **5.8.2** <u>By Buyer</u>. From the Execution Date to the Closing Date, Buyer will promptly notify the Selling Parties of (i) any Buyer Material Adverse Effect, (ii) the institution of or, to the Buyer's Knowledge, threat of institution of, any Claim against the Buyer that could reasonably be expected to have a Buyer Material Adverse Effect, or (iii) any breach of this Agreement by Buyer.
- **5.9** <u>Updates to Schedules and Exhibits</u>. From the Execution Date to the Closing Date, if the Party responsible for a Schedule or Exhibit becomes aware of any fact or condition that requires a change to that Schedule or Exhibit or any information that should have been, but was not, included in that Schedule or Exhibit, the responsible Party will promptly so notify the other Party and provide an updated Schedule or Exhibit reflecting the correct information. The receiving Party may, as its sole remedy, terminate this Agreement by notice to the responsible Party within 15 Business Days after receiving the updated Schedule or Exhibit if any of the Closing conditions of the updating Party under Article 6 could not be satisfied absent such update. If the Party receiving an updated Schedule or Exhibit does not timely terminate this Agreement as provided in this <u>Section 5.9</u>, the updated Schedule or Exhibit will replace the original Schedule or Exhibit and become part of this Agreement.
- **5.10** Interconnection Agreement. The Project is interconnected to CMP's transmission system pursuant to an interconnection agreement, dated January 1, 2006, a copy of which has been provided to Buyer ("Interconnection Agreement"). Buyer and RPC will work cooperatively to seek the consents required for the assignment of the Interconnection Agreement from NewPage Corporation to Buyer, effective as of the Effective Time, pursuant to an instrument in form and substance reasonably satisfactory to RPC, Buyer and the other parties whose consent is required, and otherwise approved by the Independent System Operator-New England ("Interconnection Agreement Assignment"). The terms of the Interconnection Agreement Assignment are set forth on Exhibit 5.10.

5.11 Further Agreements. Each Selling Party will promptly forward to Buyer (properly endorsed for deposit by Buyer) any checks or other funds such Selling Party receives on account of or in respect of the Purchased Assets on or after the Effective Time, other than checks or funds that are Excluded Assets. Buyer will promptly forward to the appropriate Selling Party (properly endorsed for deposit by such Selling Party) any checks or other funds Buyer receives on account of or in respect of the Excluded Assets on or after the Effective Time. Each Party will undertake commercially reasonable efforts to ensure that third parties direct mail and other communications to the proper Party after the Effective Time. Following the Closing, each Party will execute and deliver such documents and take such actions as the other Party may reasonably request from time to time in order to give effect to the transactions contemplated by this Agreement. Without limiting the generality of the foregoing, the Selling Parties hereby agree to take all reasonable actions as may be requested by Buyer from time to time (without obligation by the Selling Parties to pay any money or assume or incur any liability) to cure all Title Defects.

5.12 Real Estate.

5.12.1 <u>Title Commitment and Survey.</u> Prior to or as soon as practicable after the Execution Date, (a) Buyer will obtain a current title insurance commitment for the Real Property ("**Title Commitment**") issued by the Title Company, together with copies of documents referred to in the Title Commitment, and (b) RPC will deliver to Buyer a current survey (the "**Survey**") of the Real Property (as separate from the Excluded Real Property) that (i) (A) meets the 2011 Minimum Detail Requirements For ALTA/ACSM Land Title Surveys of the Real Property to the extent reasonably required by the Title Company to remove from, or reasonably limit in, Buyer's Title Policy the so-called survey exception to those items shown on an accurate survey, (B) depicts the plottable items affecting the Real Property as set forth on the Title Commitment, (C) excludes the depiction of any wires, pipes or other similar lines that are above ground level, (D) excludes the Additional Survey Work, and (E) is otherwise reasonably satisfactory to Buyer ("**Minimum Survey Requirements**"); and (ii) includes any other additional items reasonably requested by Buyer, including the depiction of any wires, pipes or other similar lines that are above ground level (collectively, "**Additional Survey Work**").

5.12.2 <u>Title Defects</u>. Within 15 Business Days after Buyer receives both of the Title Commitment and the Survey (or any update to the Title Commitment or Survey), Buyer will notify RPC of any matter disclosed in or on that document (and in the case of an update, not disclosed in or on the prior version of that document) that Buyer reasonably considers to be a Title Defect, describing the nature of the Title Defect in reasonable detail (a "**Defect Notice**"). Any matter disclosed in the Title Commitment or on a Survey or in any update to either document as to which Buyer does not provide a timely Defect Notice will be deemed a Permitted Lien. Any monetary Lien arising by, through or under third parties (whether senior or junior) encumbering easements located outside the Real Property that benefit the Purchased Assets will be deemed a Permitted Lien and will not be deemed to be a Title Defect. Within 10 Business Days after receipt of a Defect Notice, RPC will notify Buyer as to whether RPC undertakes to cure the Title Defects described in the Defect Notice, in which case RPC will have until the Closing Date to cure those Title Defects.

5.12.3 <u>Remedies</u>. If RPC fails to timely notify Buyer of its intent to cure a Title Defect described in a timely Defect Notice, or if having so notified Buyer RPC fails to cure the Title Defect prior to the Closing Date, then Buyer may, as Buyer's sole remedy, terminate this Agreement by notice to the Selling Parties within 10 Business Days after the date RPC's cure notice was due (if no cure notice was timely given) or on the Closing Date (if the cure is not completed by the Closing Date). Otherwise, Buyer will be deemed to have accepted title to the Real Property subject to the uncured Title Defect and the uncured Title Defect will be deemed a Permitted Lien.

- **5.12.4** <u>Title Policy</u>. Buyer will obtain from the Title Company an ALTA 2006 Owner's Title Insurance Policy based on the Title Commitment, insuring Buyer's title to the Real Property as of the Effective Time in an amount not less than the amount of the Purchase Price allocated to the Real Property and subject only to the Permitted Liens ("*Title Policy*").
- **5.12.5** <u>Subdivision</u>. RPC will use commercially reasonable efforts to obtain, and Buyer will cooperate with RPC in obtaining, from the appropriate Governmental Authorities any necessary approval for the subdivision of the Real Property from any other lands owned by RPC and not included in the Purchased Assets ("*Excluded Real Property*") that may be necessary to convey the Real Property to Buyer ("*Subdivision*").
- **5.12.6** <u>Costs</u>. RPC will pay for the cost of the Minimum Survey Requirements and costs associated with the Subdivision pursuant to <u>Section 5.12.5</u>. Buyer will pay for the cost of the Title Commitment, the premium for Title Policy and all endorsements to the Title Policy, and any Additional Survey Work requested by Buyer.
- **5.12.7** <u>Reciprocal Easement Agreement</u>. RPC and Buyer will at the Closing enter into a Reciprocal Easement Agreement granting and reserving easements that will benefit RPC for the operation of its manufacturing facilities and Buyer for the operation of the Purchased Assets. The Reciprocal Easement Agreement will be a Permitted Lien and will be included in the Title Policy as an insured appurtenance to the Owned Real Property.

5.13 Casualty.

- **5.13.1** Estimated Restoration Cost. If any Purchased Assets are damaged or destroyed after the Execution Date but prior to the Effective Time through no fault of Buyer or its Representatives, the Selling Parties will promptly so notify Buyer and, as soon as practicable, provide from a nationally recognized engineering firm with experience in biomass cogeneration projects that is reasonably acceptable to the Selling Parties and Buyer (a) an estimated cost of, and reasonable construction schedule for, restoring the damaged or destroyed assets to a condition reasonably comparable to their condition immediately prior to the loss, and (b) the anticipated impact of the loss, before and after repair, on Buyer's Intended Use.
- **5.13.2** Effect of Casualty Loss. The Selling Parties will bear the risk of loss or damage to the Purchased Assets resulting from fire, theft or other casualty (except reasonable wear and tear or any casualty caused by Buyer or any of its Representatives) that occurs after Execution Date and prior to the Effective Time. If any such loss or damage (i) is so substantial as to prevent normal operation of any portion of the Project or would otherwise materially impact the Buyer's Intended Use, or (ii) if the estimated restoration costs exceed \$500,000, the Buyer and the Selling Parties shall negotiate in good faith to resolve the actions required to be taken, if any, in connection with any such loss or damage. If a resolution acceptable to each of the Buyer and the Selling Parties cannot be reached within thirty (30) days after receipt of notice from the Selling Parties as required by Section 5.13.1, each of the Buyer and the Selling Parties shall have the right, in their sole discretion, to terminate this Agreement. If a negotiation notice is not timely given or if a notice of termination is not timely given, this Agreement will continue in effect and the Selling Parties will complete the restoration in accordance with the estimate. Unless the Parties otherwise agree, Buyer will have no right or interest in any insurance proceeds that either Selling Party may receive or be entitled to receive as a result of any casualty loss or damage under insurance policies held by or on behalf of either Selling Party.

5.14 Environmental Matters.

5.14.1 <u>Pre-Closing</u>. Except as otherwise contemplated by this Agreement or as may be approved by the Selling Parties in their sole discretion, from the Execution Date to the earlier of the Effective Time or the termination of this Agreement in accordance with <u>Article 8</u>, neither Buyer nor any of its Affiliates or Representatives may undertake or cause to be undertaken any Restricted Activity. Any investigation of Historical Environmental Liabilities, Indemnifiable Environmental Liabilities or Environmental Conditions by or on behalf of Buyer will not be considered a Restricted Activity if it is limited to visual inspections and site visits commonly included in the scope of a Phase 1 Level Environmental Inspection. If a Governmental Authority initiates any communication (written, electronic or verbal) with Buyer (or any of its Affiliates or Representatives, all of whom will be instructed as follows) prior to the Effective Time, Buyer will refrain from responding and immediately refer the communication to RPC for response.

5.14.2 Post-Closing.

- 5.14.2.1 <u>Restricted Activity</u>. Except as specifically authorized in this <u>Section 5.14.2</u>, neither Buyer nor any of its Affiliates or Representatives nor any of Buyer's or its Affiliates' successors or assigns, may conduct or cause or permit any other Person to conduct any of the following activities (each, a "**Restricted Activity**"):
- (i) From and after the Effective Time and on or before May 2, 2013, and except to the extent such activities are conducted in connection with a Permitted Activity (1) any activity that would materially disturb the subsurface of any portion of the Real Property, (2) any surface sampling or analysis or any subsurface sampling, analysis or investigation on or about or with respect to the Real Property, or (3) any communication with a Governmental Authority by or on behalf of Buyer or any of its Affiliates after the Closing Date relating to Historical Environmental Liabilities or Indemnifiable Environmental Liabilities; and
- (ii) From and after the Effective Time and solely with respect to that portion of the Real Property shown on Exhibit 5.14.2(A) and except to the extent such activities are conducted in connection with a Permitted Activity (1) any activity that would materially disturb the subsurface of that portion of the Real Property, (2) any surface sampling or analysis or any subsurface sampling, analysis or investigation on or about or with respect to that portion of the Real Property, or (3) any communication with a Governmental Authority by or on behalf of Buyer or any of its Affiliates after the Closing Date relating to the Environmental Condition described in Exhibit 1.1(F); and
- (iii) From and after the Effective Time and solely with respect to that portion of the Real Property shown on <u>Exhibit 5.14.2(B)</u> and except to the extent such activities are conducted in connection with a Permitted Activity (1) any groundwater sampling or testing on that portion of the Real Property, or (2) any communication with a Governmental Authority by or on behalf of Buyer or any of its Affiliates after the Closing Date relating to the Environmental Condition described in <u>Exhibit 1.1(F)</u>.
- 5.14.2.2 <u>Permitted Activity</u>. A "**Permitted Activity**" means any activity that (i) occurs in the Ordinary Course of Business to the extent the proposed activity does not occur as a result of an expansion of the Project or a change in the use of the Purchased Assets from a cogeneration facility to an alternate use or due diligence conducted by a future purchaser or financing source; (ii) is consistent with the activities conducted under a Phase 1 Level Environmental Inspection; (iii) is required by an Environmental Law or in response to an unsolicited request of a Governmental Authority; (iv) is conducted by Buyer in connection with repairs, modifications, maintenance or construction activities

related to the Purchased Assets that are conducted consistent with Prudent Industry Practices and provided such construction activities do not result in an expansion of the Project or a change in the use of the Purchased Assets from a cogeneration facility to an alternate use, or (v) is in connection with the proposed improvements to the Purchased Assets as outlined on Schedule 5.14.2.2. After the Effective Time, if the Buyer or any of its Affiliates or Representatives desires to conduct a Restricted Activity or cause or permit a Restricted Activity to be conducted, Buyer shall in each instance provide RPC prior written notice specifying the nature and reason for the proposed Restricted Activity. If RPC reasonably determines that the proposed Restricted Activity would be likely to materially adversely impact RPC's business or manufacturing facilities, RPC may so notify Buyer within 10 Business Days after RPC's receipt of Buyer's notice, and the Parties will work together in good faith to identify and implement a commercially reasonable alternative. If despite their best good faith efforts the Parties are unable to identify a commercially reasonable alternative within 30 Business days after RPC's notice, the proposed activity will remain a Restricted Activity. If RPC fails to timely object to a proposed activity that otherwise complies with the foregoing requirements, RPC will be deemed to have consented to the activity as proposed and it will be deemed a Permitted Activity. Buyer will conduct Permitted Activities in a commercially reasonable manner so as to minimize any adverse impact on RPC's business and manufacturing facilities. Any communications regarding the matters in this Section 5.14.2 will be made without prejudice and will be treated as confidential settlement discussions that may not be used as evidence in any litigation, administrative proceeding, arbitration or other dispute (other than a dispute involving compliance with this

- **5.15 Separation.** RPC will work in good faith to complete electrical and physical separation of the Project and its associated facilities and equipment from RPC's manufacturing facilities and equipment and utility assets, in the manner more fully described in Exhibit 5.15, as soon as practicable after the Closing. Buyer will cooperate with and assist RPC in good faith in the design and implementation of the separation. The separation will consist of tasks to be completed by the Selling Parties and by Buyer, as described in Exhibit 5.15, which will include the estimated total cost to complete separation at the Project and the Party responsible for paying those costs. RPC and Buyer will update Exhibit 5.15 at Closing to reflect any separation tasks that have been completed as of the Closing and any resulting adjustment to the total cost to complete separation of the Purchased Assets. Each Party will pay its own costs for all separation work completed by such Party, which costs will include such Party's reasonable attorneys' fees, consultant fees, staff salaries and other indirect costs of the separation.
- **5.16** Conditions to Closing. Prior to the Closing, (a) each Selling Party will promptly notify Buyer of any matter of which such Selling Party becomes aware that would likely cause a failure of any of the conditions to Closing referenced in Section 6.1, and (b) Buyer will promptly notify the Selling Parties of any matter of which Buyer becomes aware that would likely cause a failure of any of the conditions to Closing referenced in Section 6.2.
- **5.17** Governing Documents. Between the Execution Date and the Effective Time or the termination of this Agreement, no Party may modify or amend their Governing Documents in a way that could reasonably be expected to have a Buyer Material Adverse Effect or a Seller Material Adverse Effect.
- **5.18 Other Transactions.** Between the Execution Date and the Effective Time or the earlier termination of this Agreement, Buyer and the Selling Parties will deal exclusively and in good faith with each other regarding the transactions contemplated by this Agreement. Between the Execution Date and the Effective Time or the earlier termination of this Agreement, the Selling Parties may not directly or indirectly solicit proposals or offers from any Person, participate in discussions with any Person, provide non-public information to any Person, or enter into a Contract with any Person in furtherance of an acquisition or potential acquisition of the Purchased Assets, whether by asset sale, merger or otherwise, by any Person other than Buyer.

5.19 Project Employees. To the extent permitted by Law, RPC will provide to Buyer information about permanent non-union and union personnel employed by RPC who currently provide all or substantially all of their services in support of the Project (a "Project Employee") as Buyer may reasonably request, including the names, job functions, current compensation and a compensation history from January 1, 2009. To the extent permitted by Law, Buyer will condition any offer of employment to a non-union Project Employee on a waiver by that Project Employee of severance and other employment separation payments due from the Selling Parties to that Project Employee and will give such Project Employee credit for their accrued and unused vacation time with RPC.

5.20 <u>Inventory Credit</u>. Buyer will receive from RPC a one-time credit under the Shared Services Agreement in an amount to be determined by the Parties prior to the Closing Date, to be applied toward the purchase of shared Inventory described in <u>Exhibit 2.1(3)(B)</u>.

ARTICLE 6 CONDITIONS TO THE CLOSING

- **6.1** Conditions to the Obligations of Buyer. The obligations of Buyer to complete the transactions contemplated by this Agreement at the Closing will be subject to the fulfillment (or waiver by Buyer) on or prior to the Closing Date of the following conditions:
 - (1) All agreements and covenants required by this Agreement to be complied with or performed by the Selling Parties at or prior to the Closing (i) that are qualified as to materiality or a Seller Material Adverse Effect have been complied with or performed in all respects; and (ii) that are not so qualified have been complied with or performed in all material respects.
 - (2) All representations and warranties of the Selling Parties in this Agreement (i) that are qualified as to materiality or a Seller Material Adverse Effect will be true and correct as of the Execution Date and as of the Effective Time (or if a different time is expressly stated, as of that time) and (ii) that are not so qualified will be true and correct in all material respects as of the Execution Date and the Effective Time (or if a different time is expressly stated, as of that time).
 - The Required Regulatory Approvals with respect to Buyer's Intended Use of the Purchased Assets have been obtained and become Final Orders. No terms or conditions in addition to the Required Regulatory Approvals have been imposed in connection with the Final Orders by any Governmental Authority that, individually or in the aggregate, (a) would likely create a Buyer Material Adverse Effect or a Seller Material Adverse Effect with respect to the Purchased Assets, or (b) would subject Buyer or its Affiliates to any regulation or oversight by any Governmental Authority post-Closing that is materially greater than that experienced by the Selling Parties as of the Effective Time, but expressly excluding any regulation or oversight by any Governmental Authority that the Selling Parties are not subject to as of the Closing Date. Without limiting the generality of the foregoing, the Subdivision has been completed and approved by the applicable Governmental Authorities and the Real Property (as excluded from the Excluded Real Property) shall comply with all applicable Laws with respect to Buyer's Intended Use, including shoreland zone requirements and applicable zoning requirements.

- (4) All third party consents and approvals required for completion of the transactions contemplated by this Agreement have been obtained (including the consents and approvals set forth in <u>Schedule 3.4</u> and <u>Schedule 4.4</u>), other than any consents or approvals that, if not obtained, would not, in the aggregate, likely create a Buyer Material Adverse Effect or a Seller Material Adverse Effect with respect to the Purchased Assets.
- (5) No Order prevents the consummation of any material aspect of the transactions contemplated by this Agreement or any Ancillary Agreement (each Party will use its commercially reasonable efforts to have any such Order lifted) and no Law prohibits the consummation of the transactions contemplated by this Agreement or the Ancillary Agreements.
- (6) Buyer has received those documents to be delivered to Buyer by the Selling Parties in accordance with Section 7.3.
- (7) If the form of any Ancillary Agreement (including any schedules and exhibits to that Ancillary Agreement) is not attached in final form as an Exhibit on the Execution Date and is not subsequently agreed to in writing by the Parties prior to the Closing Date as evidenced by delivery of a final signed written agreement, the form of that Ancillary Agreement as delivered on the Closing Date must be satisfactory in all respects to Buyer. The determination of whether this condition has been satisfied will be made by Buyer in its sole discretion based on its evaluation of all of the transactions contemplated by this Agreement and the Ancillary Agreements, taken as a whole, regardless of whether the form of Ancillary Agreement delivered at the Closing is consistent or inconsistent with any summary or term sheet attached as an Exhibit on the Execution Date.
- (8) The Selling Parties shall have cured, corrected, bonded over and/or removed of record any Title Defects that the Selling Parties have undertaken to cure pursuant to Section 5.12.2 in a manner reasonably satisfactory to Buyer.
- (9) Buyer has received UCC searches of security interest filings in form and scope and substance reasonably satisfactory to Buyer, confirming the release or termination of all Liens other than Permitted Liens.
- (10) Buyer and the Selling Parties shall have mutually agreed upon the purchase price allocation pursuant to Section 2.8.
- (11) The Purchased Assets are, as of the Closing Date, in substantially the same condition as of the close of business as of the Execution Date, ordinary wear and tear excepted or, in the event of a loss or damage described under <u>Section 5.13</u>, are in the condition mutually agreed to pursuant to and in accordance with <u>Section 5.13.2</u> with respect to the Purchased Assets which are the subject of the loss or damage.
- (12) There has been no Seller Material Adverse Effect.

- (13) There is no actual, or to the Selling Parties' Knowledge, threatened, strike, slowdown or work stoppage with respect to the Project Employees.
- (14) Buyer has received from each Selling Party as applicable (i) an affidavit substantially in the form described in Treasury Regulation Section 1.1445-2(b)(2)(iv), and stating under penalty of perjury each such Selling Party's United States taxpayer identification number and that each such Selling Party is not a foreign person, and (ii) a certificate, form, affidavit, or other similar documentation stating that each Selling Party is not subject to state tax withholding.
- (15) The Real Property Lease has been terminated and the Selling Parties have delivered to Buyer written evidence of such termination, the form and substance of which is reasonably satisfactory to Buyer.
- (16) The Parties and the local and national labor unions representing any of the Project Employees have come to terms reasonably satisfactory to the Parties pertaining to those Project Employees' discontinuance of employment by the Seller and employment by the Buyer, including acknowledgement by such unions of satisfaction with compliance by the Parties of any successor requirements which may be contained within the Collective Bargaining Agreements.
- **6.2** Conditions to the Obligations of the Selling Parties. The obligation of the Selling Parties to effect the transactions contemplated by this Agreement will be subject to the fulfillment (or waiver by each Selling Party) on or prior to the Closing Date of the following conditions:
 - (1) All agreements and covenants required by this Agreement to be complied with or performed by Buyer at or prior to the Closing (i) that are qualified as to materiality or a Buyer Material Adverse Effect have been complied with or performed in all respects; and (ii) that are not so qualified have been complied with and performed in all material respects.
 - (2) All representations and warranties of Buyer in this Agreement (i) that are qualified as to materiality or a Buyer Material Adverse Effect will be true and correct as of the Execution Date and as of the Effective Time (or if a different time is expressly stated, as of that time) and (ii) that are not so qualified will be true and correct in all material respects as of the Execution Date and the Effective Time (or if a different time is expressly stated, as of that time).
 - (3) The Required Regulatory Approvals have been obtained and become Final Orders. No terms or conditions in addition to the Required Regulatory Approvals have been imposed in connection with the Final Orders by any Governmental Authority that, individually or in the aggregate, (a) would likely create a Seller Material Adverse Effect, (b) would decrease the net proceeds the Selling Parties anticipate realizing from the transactions contemplated by this Agreement, or (c) would subject either Selling Party or its Affiliates to new or additional regulation or oversight by any Governmental Authority post-Closing.
 - (4) All third party consents and approvals required for the consummation of the transactions contemplated by this Agreement have been obtained (including the

- consents and approvals set forth in <u>Schedule 3.4</u> and <u>Schedule 4.4</u>), other than any consents or approvals that, if not obtained, would not, in the aggregate, likely create a Seller Material Adverse Effect.
- (5) No Order prevents the consummation of any material aspect of the transactions contemplated by this Agreement or any Ancillary Agreement (each Party will use its commercially reasonable efforts to have any such Order lifted) and no Law prohibits the consummation of the transactions contemplated by this Agreement or the Ancillary Agreements.
- (6) There has been no Buyer Material Adverse Effect.
- (7) The Buyer and the Selling Parties shall have mutually agreed upon the purchase price allocation pursuant to <u>Section 2.8</u>.
- (8) The Selling Parties have received those documents to be delivered to the Selling Parties in accordance with Section 7.4.
- (9) If the form of any Ancillary Agreement (including any schedules and exhibits to that Ancillary Agreement) is not attached in final form as an Exhibit on the Execution Date and is not subsequently agreed to in writing by the Parties prior to the Closing Date as evidenced by delivery of a final signed written agreement, the form of that Ancillary Agreement as delivered on the Closing Date must be satisfactory in all respects to the Selling Parties. The determination of whether this condition has been satisfied will be made by the Selling Parties in their sole discretion based on their evaluation of all of the transactions contemplated by this Agreement and the Ancillary Agreements, taken as a whole, regardless of whether the form of Ancillary Agreement delivered at the Closing is consistent or inconsistent with any summary or term sheet attached as an Exhibit on the Execution Date.
- (10) The Parties and the local and national labor unions representing any of the Project Employees have come to terms reasonably satisfactory to the Parties pertaining to those Project Employees' discontinuance of employment by the Seller and employment by the Buyer, including acknowledgement by such unions of satisfaction with compliance by the Parties of any successor requirements which may be contained within the Collective Bargaining Agreements.

ARTICLE 7 CLOSING

- **7.1** Closing. The closing of the transactions contemplated by this Agreement (the "Closing") will take place at the offices of RPC's counsel in Portland, Maine, commencing at 10:00 a.m. on the third Business Day after all conditions precedent in Article 6 have been satisfied in full or waived, or at such other time and place as the Parties may agree (the "Closing Date"). Risk of loss for the Purchased Assets will transfer from the Selling Parties to Buyer as of the Effective Time.
- 7.2 <u>Timing and Effectiveness of Actions</u>. At the Closing, the Parties will take the actions described in this <u>Article 7</u>. All actions will be deemed to have occurred simultaneously and as of the Effective Time and, unless Buyer and the Selling Parties otherwise agree, the effectiveness of any action taken at the Closing will be conditioned upon the taking of all other actions required by this Article 7.

7.3 Selling Parties' Deliveries. At the Closing, RPC or Seller, as applicable, will deliver the following:

- (1) The Assignment and Assumption Agreement, executed by the Selling Parties;
- (2) The Bill of Sale executed by the Selling Parties;
- (3) The Fuel Supply Agreement executed by RPC;
- (4) The Interconnection Assignment Agreement executed by RPC and consented to by all required parties;
- (5) The Joint Use Agreement executed by RPC;
- (6) The Joint Operating Committee Agreement by RPC;
- (7) The Energy Supply Agreement executed by RPC;
- (8) The Quitclaim Deed executed and acknowledged by RPC;
- (9) The Reciprocal Easement Agreement executed and acknowledged by RPC;
- (10) The Recovery Boiler Agreement executed by RPC;
- (11) The Shared Services Agreement executed by RPC;
- (12) The Undivided Ownership Agreement executed by all required parties other than Buyer;
- (13) The NewPage Guaranty executed by NewPage Corporation;
- (14) The Non-Solicitation Agreement executed by RPC;
- (15) A certificate signed by the President or a Vice President of RPC and the general partner of Seller, each solely on behalf of itself, and dated the Closing Date as to the matters set forth in Sections 6.1(1), 6.1(2) and 6.1(12);
- (16) Consents, waivers or approvals obtained by the Selling Parties from third parties in connection with this Agreement;
- (17) Terminations or releases of Liens on the Purchased Assets that are not Permitted Liens and a subordination of the Selling Parties' Affiliates' lender's liens on the Easement Rights;
- (18) A good standing certificate with respect to each Selling Party (dated as of a recent date prior to the Closing Date), issued by the applicable Secretary of State;
- (19) A copy, certified by an authorized officer of each of RPC and the general partner of Seller, of the resolutions authorizing the execution and delivery of this

Agreement and the Ancillary Agreements and instruments attached as Exhibits to this Agreement and to the Ancillary Agreements, and the consummation of the transactions contemplated by this Agreement and the Ancillary Agreements, together with a certificate by the Secretary of each of RPC and the general partner of Seller as to the incumbency of those officers authorized to execute and deliver this Agreement and the Ancillary Agreements on behalf of the Selling Parties;

- (20) The Documents, either physically at the Closing or through electronic access;
- (21) A letter from AECOM permitting Buyer to rely on the AECOM "Phase I Environmental Assessment, Rumford Mill, Rumford, Maine" (AECOM Project No. 60188893, January 2011);
- (22) Such other agreements, documents, instruments and writings as are required to be delivered by the Selling Parties at or prior to the Closing Date pursuant to this Agreement or any Ancillary Agreement; and
- (23) All instruments of transfer Buyer reasonably determines are appropriate to consummate the transactions contemplated by this Agreement in form and substance reasonably satisfactory to Buyer to vest in Buyer all of the Selling Parties' rights, title and interests in the Purchased Assets to be transferred to Buyer pursuant to, and in accordance with, the terms and conditions of this Agreement.

7.4 **Buyer's Deliveries.** At the Closing, Buyer will deliver the following:

- (1) The Cash Purchase Price, paid in the manner provided in <u>Section 2.6</u>;
- (2) The Secured Note executed by Buyer;
- (3) The Mortgage executed and acknowledged by Buyer;
- (4) The Security Agreement executed by Buyer;
- (5) The Assignment and Assumption Agreement executed by Buyer;
- (6) The Bill of Sale executed by Buyer;
- (7) The Fuel Supply Agreement executed by Buyer;
- (8) The Interconnection Assignment Agreement, as applicable, executed by Buyer and consented to by all required parties;
- (9) The Joint Use Agreement executed by Buyer;
- (10) The Joint Operating Committee Agreement by Buyer;
- (11) The Energy Supply Agreement executed by Buyer;
- (12) The Recovery Boiler Agreement executed by Buyer;

- (13) The Reciprocal Easement Agreement executed and acknowledged by Buyer;
- (14) The Shared Services Agreement executed by Buyer;
- (15) The Undivided Ownership Agreement executed by all required parties other than RPC;
- (16) The Non-Solicitation Agreement executed by Buyer and ReEnergy Holdings LLC;
- (17) All other instruments of assignment or conveyance as are reasonably required by either Selling Party in connection with the transfer of the Purchased Assets to Buyer in accordance with this Agreement;
- (18) A certificate signed by the President or a Vice President of Buyer and dated the Closing Date as to the matters set forth in Sections 6.2(1), 6.2(2) and 6.2(6);
- (19) A good standing certificate with respect to Buyer and ReEnergy Holdings LLC (dated as of a recent date prior to the Closing Date), issued by the Secretary of State for the State of Delaware;
- (20) A copy, certified by an authorized officer of Buyer, of the resolutions authorizing the execution and delivery of this Agreement and the Ancillary Agreements and instruments attached as Exhibits to this Agreement and to the Ancillary Agreements, and the consummation of the transactions contemplated by this Agreement and the Ancillary Agreements, together with a certificate by the managing member of Buyer as to the incumbency of those officers authorized to execute and deliver this Agreement and the Ancillary Agreements;
- (21) The ReEnergy Guaranty executed by ReEnergy Holdings LLC;
- (22) Consents, waivers or approvals obtained by Buyer from third parties in connection with this Agreement; and
- (23) Such other agreements, documents, instruments and writings as are required to be delivered by Buyer at or prior to the Closing Date pursuant to this Agreement or any Ancillary Agreement.

ARTICLE 8 TERMINATION

- **8.1** <u>Termination.</u> Notwithstanding anything in this Agreement to the contrary, this Agreement may be terminated at any time prior to the Closing, effective at the time specified by the terminating Party in its notice of termination to the other Party:
 - (1) By mutual consent of Buyer and the Selling Parties.
 - (2) By either Buyer or either Selling Party if the transactions contemplated by this Agreement have not been consummated on or before June 30, 2011 ("*Expiration Date*"), except that a Party seeking to terminate may not do so if it has failed to

perform any material obligation under this Agreement (including Section 5.5) or to fulfill any material condition under this Agreement within the control of that Party and that failure has been the proximate cause of, or has resulted in, the failure to complete the transactions contemplated by this Agreement on or before the Expiration Date. Upon notice to the other Party, either Party may extend the Expiration Date for a period of up to sixty (60) days if the Closing shall not have occurred by June 30, 2011 solely due to Governmental Authority's failure to issue a Final Order, provided such failure is not attributable to the acts or omissions of the Party seeking to extend the Expiration Date.

- (3) By Buyer:
 - (a) If one or more of the conditions in <u>Section 6.1</u> cannot be satisfied and have not been waived by Buyer; or
 - (b) If either Selling Party has breached any material covenant, representation, warranty or other provision in this Agreement that can be cured and is not cured as soon as reasonably practicable and in any event within 45 days after notice of the breach is given by Buyer to the Selling Parties; or
 - (c) Pursuant to Section 5.9; or
 - (d) Pursuant to Section 5.12.3; or
 - (e) Pursuant to Section 5.13.2.
- (4) By either Selling Party:
 - (a) If one or more of the conditions in <u>Section 6.2</u> cannot be satisfied and have not been waived by the applicable Selling Party; or
 - (b) If Buyer has breached any material covenant, representation, warranty or other provision in this Agreement that can be cured and is not cured as soon as reasonably practicable and in any event within 45 days after notice of the breach is given by either Selling Party to Buyer; or
 - (c) Pursuant to Section 5.9; or
 - (d) Pursuant to Section 5.13.2.

8.2 Effect of Termination. If this Agreement is terminated pursuant to Section 8.1, the provisions of this Agreement will immediately become void and of no further force and effect from and after the date of termination, except that the provisions of the Confidentiality Agreement and the rights of Buyer and the Selling Parties set forth in Section 10.11 and the last sentence of this Section 8.2 will survive termination of this Agreement. The right of either Party to terminate this Agreement will be in addition to and not in lieu of any other remedies of Buyer or either Selling Party available at law or in equity in respect of any breach of any covenant or agreement in this Agreement occurring prior to the date of termination of this Agreement.

ARTICLE 9 INDEMNIFICATION

9.1 Survival.

- 9.1.1 Representations and Warranties. The representations and warranties of the Parties in this Agreement will terminate and expire 18 months after the Closing Date, except that (i) the Selling Parties' and Buyer's respective representations and warranties in Sections 3.2, 3.3, 3.7, 3.8.1, 4.2, 4.3 and 4.6 will terminate and expire 36 months after the Closing Date, (iii) the Selling Parties' representations and warranties in 3.11 will terminate and expire 60 months following the Closing Date (subject to Section 9.1.3), (iv) the Selling Parties' representations and warranties in Section 3.14 will terminate and expire ninety (90) days following the expiration of the applicable statue of limitations for the assertion by any Person (including Governmental Authorities) of claims that would give rise to indemnification under this Article 9, and (v) the Selling Parties' and Buyer's representations and warranties in Sections 3.18 and 4.8 will survive the Closing indefinitely.
- **9.1.2** *Covenants*. Either party's right to seek indemnification for a breach of a covenant under this Agreement will survive the Closing indefinitely. The Parties obligation to perform and/or abide by the covenants set forth in Sections 5.2, 5.3.1, 5.5, 5.6, 5.8, 5.9, 5.12, 5.13, 5.14.1, 5.16, 5.17, 5.18, and 5.19 will terminate and expire at the Effective Time, (ii) the Selling Parties' obligations to perform and/or abide by the covenants under Section 5.1 will terminate and expire 30 days after the Effective Time, (iii) the Selling Parties' obligation to perform and/or abide by the covenants under Section 5.3.2 will terminate and expire five years after the Effective Time, (iv) the Parties' obligation to perform and/or abide by the covenants under Section 5.3.3 will terminate and expire seven years after the Effective Time; and (v) Buyer's obligations to perform and/or abide by the covenants set forth under Section 5.14.2.1(i) shall expire on May 2, 2013. All other covenants of the Parties will survive indefinitely unless a shorter survival period is provided for in this Agreement.
- **9.1.3** <u>Historical Environmental Liabilities</u>. For purposes of absolute clarity, the Selling Parties' indemnification obligations for Historical Environmental Liabilities under this <u>Article 9</u> shall not extend beyond May 1, 2013, notwithstanding a breach of Section 3.11.
- **9.1.4** <u>Notice</u>. Notwithstanding anything else to the contrary in this <u>Article 9</u>, any claim for indemnification under this <u>Article 9</u> will be valid only if the Party seeking indemnification notifies the other Party of the basis for the claim within the applicable survival period provided in <u>Section 9.1.1</u> and <u>9.1.3</u>; otherwise the claim will be deemed to have been waived by the Party seeking indemnification. Notwithstanding the foregoing, with respect to any claim for indemnity delivered on or prior to the applicable survival period provided in <u>Section 9.1.1</u> and <u>9.1.3</u>, the matters that are the subject of such indemnification claim shall survive with respect to such claim until such claim is fully and finally resolved or settled by written agreement of the Parties.
- **9.2** <u>Indemnification By the Selling Parties</u>. From and after the Effective Time, subject to the provisions of this <u>Article 9</u>, the Selling Parties, jointly and severally, will indemnify and hold Buyer and each of its Affiliates, and each of their respective directors, partners, shareholders, officers and employees (collectively, the "*Buyer Group*"), harmless from and against any Liabilities imposed upon or suffered or incurred by any of them which are based upon or arise out of or in connection with:
 - (a) Any misrepresentation or inaccuracy of a representation or warranty made by a Selling Party in this Agreement, excluding Liabilities (other than Liabilities arising in connection with the Selling Parties' fraud or intentional misrepresentation)

arising under Section 3.8.1 that result from a Lien or other risk covered by a standard form ALTA 2006 Owner's Policy of Title Insurance with the endorsements listed on Exhibit 9.2, (i) regardless of whether a Title Policy is obtained by Buyer or whether the Title Policy obtained by Buyer includes those endorsements and regardless of the dollar limits of Title Policy coverage selected by Buyer and (ii) disregarding the application of the standard "Exclusions From Coverage" that relate to defects, liens, encumbrances, adverse claims, or other matters (1) created suffered, assumed or agreed to by Buyer or (2) not Known (as defined in the standard "Exclusions From Coverage") to the title company issuing the Title Policy but Known (as defined in the standard "Exclusions From Coverage") to Buyer and not disclosed in writing by Buyer to the title company prior to the date of the issuance of the Title Policy.

- (b) Any breach or non-fulfillment of any covenant or agreement on the part of either Selling Party in this Agreement,
- (c) Indemnifiable Environmental Liabilities, excluding (i) Indemnifiable Environmental Liabilities resulting from Buyer's violation of Environmental Law or willful or reckless disregard of Prudent Industry Practices with respect to an Environmental Condition existing prior to the Effective Time; or (ii) Indemnifiable Environmental Liabilities with respect to the Environmental Condition described in Exhibit 1.1(F) directly or indirectly discovered or revealed as a result of a breach of Section 5.14.2.1; or (iii) Indemnifiable Environmental Liabilities with respect to the Environmental Condition described in Exhibit 1.1(F), to the extent that those Indemnifiable Environmental Liabilities result from the failure of Buyer, its successors or assigns, to observe the requirements of the Environmental Protocol to be agreed to by the Parties prior to the Closing Date, or
- (d) Historical Environmental Liabilities, excluding (i) Historical Environmental Liabilities resulting from Buyer's violation of Environmental Law or willful disregard of Prudent Industry Practices with respect to an Environmental Condition existing prior to the Effective Time; or (ii) Historical Environmental Liabilities directly or indirectly discovered or revealed as a result of a breach of Section 5.14.2.1(i).
- **9.3** <u>Indemnification By Buyer</u>. From and after the Effective Time and subject to the provisions of this <u>Article 9</u>, Buyer will indemnify and hold the Selling Parties and their Affiliates, and each of their respective directors, partners, shareholders, officers and employees (collectively, the "*Seller Group*"), harmless from and against Liabilities imposed upon or suffered or incurred by any of them which are based upon or arise out of or in connection with:
 - (a) Any misrepresentation or inaccuracy of a representation or warranty made by Buyer in this Agreement,
 - (b) Any breach or non-fulfillment of any covenant or agreement on the part of Buyer in this Agreement (excluding Buyer's breach or non-fulfillment of Section 5.14), or
 - (c) Any Third Party Claim resulting from Buyer's failure to perform under any Assumed Contract, Permit, Environmental Permit or Transferable Warranty under which Buyer is receiving benefits and that has not yet been assigned to Buyer as described Section 2.5, except to the extent such Liabilities could have been avoided through the exercise of reasonable care by the Selling Parties.

9.4 Indemnification Procedures.

9.4.1 <u>Asserting a Claim</u>. Without limiting the provisions of <u>Section 9.1</u>, if a Party (an "Indemnified Party") wishes to assert a claim for indemnification against the other Party (the "Indemnifying Party"), the Indemnified Party will give notice to the Indemnifying Party, setting forth with particularity the basis for the claim, promptly after the Indemnified Party becomes aware of any fact, condition, or event which gives rise to a Claim for which indemnification may be sought under this <u>Article 9</u>. Failure to notify the Indemnifying Party will not relieve the Indemnifying Party of liability it may have to the Indemnified Party, except to the extent that the Indemnifying Party has been materially prejudiced by the failure and except that nothing in this Agreement will be deemed to extend the time limits in <u>Section 9.1</u>. Each Indemnified Party will use commercially reasonable efforts to mitigate Liabilities for which it seeks indemnification under this <u>Article 9</u>.

9.4.2 Third Party Claims. If any lawsuit, enforcement action, demand or claim is brought or made by a Person who is not a member of the Seller Group or Buyer Group (a "Third Party Claim") against an Indemnified Party which is the basis for an indemnification claim pursuant to Section 9.2 or 9.3, the Indemnifying Party may, if it so elects, take control of the defense and investigation of the Third Party Claim and to employ and engage attorneys of its own choice reasonably acceptable to the Indemnified Party to handle and defend the Third Party Claim, at the Indemnifying Party's cost, risk and expense. Any election by the Indemnifying Party to take control of the defense and investigation of a Third Party Claim will not be deemed a waiver of the Indemnifying Party's right to determine at a later date that the Third Party Claim is not entitled to indemnification under this Agreement, in which case Indemnifying Party may, in the exercise of its sole discretion, determine not to continue to defend that Third Party Claim and any action taken by the Indemnifying Party in connection with that determination will be undertaken in a manner so as not to materially prejudice the defense or the rights of the Indemnified Party. The Indemnified Party will cooperate with the Indemnifying Party so as to minimize the risk of any such prejudice. The Indemnified Party will cooperate in all reasonable respects with the Indemnifying Party and its attorneys in the investigation, trial and defense of any Third Party Claim and any resulting appeal, including (a) furnishing such records, information and testimony, and attending such conferences, discovery proceedings, hearings, trials and appeals, as reasonably may be requested in connection with the Third Party Claim, (b) affording access during normal business hours to the Indemnifying Party to, and reasonable retention by the Indemnified Party of, records and information which are reasonably relevant to the Third Party Claim, and (c) making its employees available on a mutually convenient basis to provide additional information and explanation of any material provided to the Indemnifying Party under this Agreement. The Indemnified Party nevertheless may, at its own cost, participate in the investigation, trial and defense of a Third Party Claim or any resulting appeal.

9.4.3 Indemnity Payments. Each indemnification payment required to be made by the Selling Parties to Buyer under this Article 9 ("Indemnity Payment") will be payable fifty percent (50%) in cash with the remaining fifty percent (50%) paid to a member of the Buyer Group by reducing Buyer's next payment obligation, if any, under the Secured Note by an amount equal to fifty percent (50%) of such Indemnity Payment. Notwithstanding the foregoing, if Buyer's next payment obligation under the Secured Note is less than fifty percent (50%) of an Indemnity Payment, the balance of that Indemnity Payment will be applied to subsequent payment obligations under the Secured Note in full in the order in which they become due. If the Indemnity Payment reduces amounts owed under the Secured Note to zero or if the Secured Note has been paid in full at the time the Indemnity Payment becomes due and owing, then the remaining balance of the Indemnity Payment shall be paid to the Buyer in cash. The Selling Parties' acknowledge and agree that no interest shall accrue on any amount due under the Secured Note that is to be reduced by an Indemnity Payment. The Selling Parties may also set off against any cash payment contemplated by this Section 9.4.3 any amounts then due but unpaid under the Secured Note.

9.5 Settlement and Defense of Third Party Claims. If the Indemnifying Party has assumed control of, and is diligently proceeding with, the defense of a Third Party Claim pursuant to Section 9.4.2, the Indemnifying Party may consent to a settlement or compromise of, or the entry of any monetary judgment arising from, the Third Party Claim without the prior consent of the Indemnified Party if, and only if, the proposed settlement, compromise or entry (a) does not contain an admission of guilt or wrongdoing on the part of the Indemnified Party. (b) does not provide for any remedy or sanction against the Indemnified Party other than the payment of money which the Indemnifying Party agrees to pay and does pay in full, and (c) the Indemnified Party and its Affiliates are released by all parties asserting the Third Party Claim from all Liabilities that are or could have been alleged, claimed or demanded in the Third Party Claim to the reasonable satisfaction of the Indemnified Party. If the Indemnifying Party does not assume the defense of the Third Party Claim in accordance with Section 9.4 within 10 Business Days after receipt of notice of the Third Party Claim, the Indemnified Party may at the expense of the Indemnifying Party defend the Third Party Claim. If the Indemnified Party controls the defense of the Third Party Claim pursuant to the preceding sentence, prior to the settlement of any Third Party Claim, the Indemnified Party shall provide the Indemnifying Party notice of such settlement and shall provide the Indemnifying Party reasonable information pertaining to the Indemnified Party's defense of the Third Party Claim. The Parties shall cooperate in good faith and take reasonable steps to preserve the attorney-client privilege, provided, however and notwithstanding the foregoing, the Indemnified Party shall be under no obligation to take any action that would constitute (in the Indemnified Party's reasonable opinion) a waiver of the attorney-client privilege. Upon receipt of such notice and information, the Indemnifying Party shall promptly (taking into account the circumstances surrounding the proposed settlement), but no longer than 10 Business Days, inform the Indemnified Party if the Indemnifying Party elects to assume control of the defense of the Third Party Claim. If the Indemnifying Party elects to assume control of the defense of the Third Party Claim, the Indemnified Party shall relinquish control of the defense of the Third Party Claim if the Indemnified Party's defense of the Third Party Claim would not be materially prejudiced thereby. The Indemnifying Party's assumption of the control of the defense of any Third Party Claim shall at all times remain subject to the terms of Section 9.4 and 9.5. If the Indemnifying Party elects not to assume control of the Third Party Claim or the Indemnified Party's defense of the Third Party Claim would be materially prejudiced by relinquishing the defense of the Third Party Claim to the Indemnifying Party, the Indemnified Party may settle any such Third Party Claim in its reasonable discretion.

9.6 <u>Adjustment to Purchase Price</u>. Amounts paid with respect to indemnification pursuant to <u>Section 9.2</u> or <u>9.3</u> will be treated as an adjustment to the Purchase Price, and the Parties will make appropriate adjustments to the allocation of the Purchase Price pursuant to Section 2.8.

9.7 General Limitations.

9.7.1 <u>Basket</u>. The Selling Parties will not be liable to any member of the Buyer Group for any Liabilities under <u>Section 9.2(a)</u>, and Buyer will not be liable to any member of the Selling Group for any Liabilities under <u>Section 9.3(a)</u>, unless and until the aggregate cumulative amount of those Liabilities exceeds 1% of the Cash Purchase Price and the initial principal amount of the Secured Note ("**Basket**"), and will then be liable only to the extent that the aggregate cumulative amount of those Liabilities suffered or incurred by the Indemnified Party exceeds the Basket. This <u>Section 9.7.1</u> does not apply to (i) indemnifiable Liabilities suffered by Buyer as a result of a Selling Party's breach of <u>Sections 3.2, 3.3, 3.14</u> or <u>3.18</u>; (ii) indemnifiable Liabilities suffered by either Selling Party as a result of Buyer's breach of <u>Sections 4.2, 4.3 or, 4.8</u>, (iii) indemnifiable Liabilities suffered by either Party as a result of the other Party's failure to pay or otherwise discharge any Tax for which it is liable pursuant to this Agreement; or (iv) fraud or intentional misrepresentation.

- **9.7.2** <u>Cap.</u> In no event will the aggregate amount of the Selling Parties' cumulative Liabilities under <u>Section 9.2(a)</u>, or the aggregate amount of Buyer's cumulative Liabilities under Section 9.3(a), exceed 20% of the Cash Purchase Price and the initial principal amount of the Secured Note ("**Cap**"). This <u>Section 9.7.2</u> does not apply to (i) indemnifiable Liabilities suffered by Buyer as a result of either Selling Party's breach of <u>Section 3.2, 3.3, 3.14</u> or <u>3.18;</u> (ii) indemnifiable Liabilities suffered by either Selling Party as a result of Buyer's breach of <u>Section 4.2, 4.3</u>, or <u>4.8</u>, (iii) indemnifiable Liabilities suffered by either Party as a result of the other Party's failure to pay or otherwise discharge any Tax for which it is liable pursuant to this Agreement; or (iv) fraud or intentional misrepresentation.
- 9.7.3 Further Clarification. For purposes of absolute clarity, (i) nothing in this Agreement shall be deemed to limit any Party's Liabilities under Section 9.2(b), 9.2(c), 9.2(d), 9.3(b) or 9.3(c), even if such Liabilities could also be characterized as Liabilities under Section 9.2(a) or 9.3(a), as applicable, and (ii) all Liabilities shall be aggregated among all members of the Seller Group or all members of the Buyer Group, as applicable, for purposes of determining whether the Basket or Cap has been reached. For the sole purpose of calculating the Basket, any qualification or limitation of a representation, warranty or covenant by reference to materiality of matters stated therein or as to matters having or not having a "Material Adverse Effect" or words of similar effect, will be disregarded.
- 9.7.4 Limitation. NOTWITHSTANDING ANYTHING TO THE CONTRARY ELSEWHERE IN THIS AGREEMENT OR PROVIDED FOR UNDER ANY LAW, NO PARTY WILL, IN ANY EVENT, BE LIABLE TO THE OTHER PARTY, EITHER IN CONTRACT OR IN TORT, FOR ANY CONSEQUENTIAL, INCIDENTAL, INDIRECT, SPECIAL, OR PUNITIVE DAMAGES OF THE OTHER PARTY, INCLUDING LOSS OF FUTURE REVENUE, INCOME, OR PROFITS, OR LOSS OF BUSINESS REPUTATION OR OPPORTUNITY, RELATING TO THE BREACH OR ALLEGED BREACH OF THIS AGREEMENT OR OTHERWISE, WHETHER OR NOT THE POSSIBILITY OF SUCH DAMAGES HAS BEEN DISCLOSED TO THE OTHER PARTY IN ADVANCE OR COULD HAVE BEEN REASONABLY FORESEEN BY SUCH OTHER PARTY. THE EXCLUSION OF CONSEQUENTIAL, INCIDENTAL, INDIRECT, SPECIAL, AND PUNITIVE DAMAGES AS SET FORTH IN THE PRECEDING SENTENCE DOES NOT APPLY TO ANY SUCH DAMAGES SOUGHT BY THIRD PARTIES AGAINST BUYER OR ANY OF THE SELLING PARTIES, AS THE CASE MAY BE, IN CONNECTION WITH ANY LIABILITY THAT MAY BE INDEMNIFIED PURSUANT TO THIS ARTICLE 9.
- 9.7.5 Collateral Sources. In calculating the amount of Liabilities for which an Indemnified Party is entitled to indemnification under Section 9.2 or 9.3, as applicable, the amount of indemnifiable Liabilities will be reduced by (a) any insurance proceeds net of expenses actually received by the Indemnified Party from an insurance carrier with respect to those Liabilities, provided that if the Indemnified Party fails to either diligently pursue any such insurance proceeds reasonably available to it or to assign a valid right to the Indemnifying Party to pursue such insurance proceeds, the Indemnifying Party's obligation will be reduced by the amount of insurance proceeds reasonably available to the Indemnified Party, and (b) any amounts actually received by the Indemnified Party from third parties with respect to Liabilities pursuant to indemnification, warranty or other similar rights, provided that if the Indemnified Party fails to diligently pursue such rights, the Indemnifying Party's obligation will be reduced by the amounts available to the Indemnified Party. If any Liabilities for which indemnification is actually provided by an Indemnifying Party to an Indemnified Party under Section 9.2 or 9.3, as applicable, are subsequently reduced by or the cause or source of any insurance payment or recovery from a third party, the Indemnified Party will promptly remit the amount of the payment or recovery to the Indemnifying Party. Any reduction of Liabilities due to proceeds received by an Indemnified Party from

a third party or from its own insurance carrier, and thus either not paid by or reimbursed to the Indemnifying Party, will not be counted or otherwise apply in determining the aggregate cumulative amount of Liabilities of the Indemnified Party that count toward the Basket or the Cap.

9.7.6 Additional Limitations.

9.7.6.1 <u>Disclaimer of Liability of the Selling Parties</u>. For the purposes of absolute clarity, except as otherwise specifically agreed elsewhere in this Agreement or in any Ancillary Agreement, except to the extent related to, caused by or resulting from the acts or omissions of the Selling Parties, its Affiliates and/or their Representatives and subject in all cases to the indemnification obligations of the Selling Parties pursuant to <u>Section 9.2</u> of this Agreement, neither Selling Party will be liable under this Agreement for any (a) Assumed Liabilities; (b) Environmental Conditions first existing or arising with respect to the Purchased Assets on or after the Effective Time and unrelated to Historical Environmental Liabilities or Indemnifiable Environmental Liabilities; (c) Liabilities resulting from Buyer's violation of Environmental Law; (d) Liabilities accruing or arising on or after the Effective Time (i) under any FERC Orders or Independent System Operator-New England Orders applicable to the Purchased Assets or that otherwise affect or govern the rates, services or operations of the Purchased Assets or their saleable output, or (ii) imposed on Buyer or on the Purchased Assets, in connection with any Required Regulatory Approval; (e) Liabilities associated with the Purchased Assets in respect of Taxes for which Buyer is liable pursuant to <u>Section 2.7</u> or <u>5.7</u>; or (f) for other Liabilities with respect to the ownership, possession, use or operation of the Purchased Assets first arising and to be performed on or after the Effective Time.

9.7.6.2 <u>Disclaimer of Liability of Buyer</u>. For the purposes of absolute clarity, except as otherwise specifically agreed elsewhere in this Agreement or in any Ancillary Agreement, except to the extent related to, caused by or resulting from the acts or omissions of the Buyer, its Affiliates and/or their Representatives and subject in all cases to the indemnification obligations of the Buyer pursuant to <u>Section 9.3</u> of this Agreement, Buyer will not be liable under this Agreement (a) for Liabilities relating to or arising from any Excluded Liability; (b) Liabilities arising from or relating to any breach or violation of Assumed Contracts, Transferable Permits and Transferable Environmental Permits by the Selling Parties prior to the Effective Time or default under such Assumed Contracts, Transferable Permits and Transferable Environmental Permits by the Selling Parties prior to the Effective Time; (c) Liabilities associated with the Purchased Assets in respect of Taxes for which the Selling Parties are liable pursuant to <u>Section 2.7</u> or <u>5.7</u>; or (d) for other Liabilities with respect to the ownership, possession, use or operation of the Purchased Assets first arising or to be performed prior to the Effective Time, exclusive of the Assumed Liabilities.

9.8 Exclusive Remedies. From and after the Effective Time, the sole and exclusive remedy for any breach or inaccuracy, or alleged breach or inaccuracy, of any representation or warranty in this Agreement or any covenant or agreement to be performed under this Agreement on or prior to the Effective Time, will be indemnification in accordance with this Article 9, as applicable. In furtherance of the foregoing, each Selling Party and Buyer waive, to the fullest extent permitted by Law, all other rights, claims, and causes of action (including any rights of contributions) that may be based upon, arise out of, or relate to this Agreement, or the negotiation, execution, or performance of this Agreement (including any tort or breach of contract claim or cause of action based upon, arising out of, or related to any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement), known or unknown, foreseen or unforeseen, which exist or may arise in the future, that it may have against the other arising under or based upon any Law (including any Environmental Law), common law, or otherwise. For the avoidance of doubt, indemnification in accordance with this Article 9 will not be the sole or exclusive remedy for post-Closing breaches of covenants that survive the Closing.

ARTICLE 10 MISCELLANEOUS

- 10.1 Entire Agreement; Amendment. This Agreement and the Ancillary Agreements (including their Exhibits and Schedules) constitute the complete and entire agreement between the Parties with respect to its subject matter and will supersede all previous oral and written negotiations and commitments and any other writings with respect to the subject matter of this Agreement, and there are no agreements, representations or warranties between the Parties other than the Confidentiality Agreement. This Agreement cannot be modified or amended except in writing duly executed by each Party.
- 10.2 Notices. All notices, requests, demands and other communications under this Agreement must be in writing and will be deemed to have been given (a) when delivered personally, (b) one Business Day after being sent by electronic mail, (c) one Business Day after being sent by nationally recognized overnight delivery service, (d) when sent by facsimile transmission, if the sending facsimile machine receives and prints a confirmation of receipt by the receiving facsimile machine, or (e) upon receipt after being deposited in the United States mail, certified and with proper postage prepaid, addressed as follows:

If to Buyer:

ReEnergy Rumford LLC 20 Century Hill Drive, Suite 203 Latham, NY 12110

Attn.: William H. Ralston, Esq. wralston@reenergyholdings.com

And:

ReEnergy Holdings LLC 20 Century Hill Drive, Suite 203

Latham, NY 12110

Attn.: William H. Ralston, Esq. wralston@reenergyholdings.com

With a courtesy copy to:

Brian A. Betancourt, Esq. Alston & Bird LLP 90 Park Avenue

New York, New York 10016 brian.betancourt@alston.com

If to Seller or RPC:

Rumford Paper Company 35 Hartford Street Rumford, Maine 04276 Attn.: Gerald A. LeClaire gerald.leclaire@newpagecorp.com

And:

NewPage Corporation 8540 Gander Creek Drive Miamisburg, OH 45342 Attn: General Counsel

doug.cooper@newpagecorp.com

With a courtesy copy to:

Bonnie L. Martinolich, Esq.

Preti Flaherty LLP One City Center Portland, Maine 04101 bmartinolich@preti.com

Any Party may change the address to which notices or other communications are to be directed to it by giving notice of the change to the other Party in the manner provided in this <u>Section 10.2</u>. Courtesy copies are for convenience only and failure to provide a courtesy copy does not alter the effectiveness of a notice or other communication given in accordance with this <u>Section 10.2</u>.

10.3 <u>Consents and Approvals</u>. Unless otherwise specifically indicated, wherever the consent or approval of either Party is required or permitted in this Agreement, the consent or approval must be in writing and may not be unreasonably withheld, delayed or conditioned.

- **10.4** <u>Counterparts</u>. This Agreement may be executed simultaneously in two or more counterparts (which may be by facsimile or electronic transmission), each of which will be deemed to be original, but all of which together will constitute one and the same instrument.
- **10.5** <u>Parties in Interest; Assignment.</u> This Agreement and all of the provisions of this Agreement will be binding upon, will inure to the benefit of, and will be enforceable by the Parties and their respective successors and permitted assigns. Neither this Agreement nor any of the rights, interests or obligations under this Agreement may be assigned by either Party without the prior consent of the other Party.
- **10.6** No Waiver. Except as otherwise expressly provided in this Agreement, the failure of any Party to enforce any provision of this Agreement on any occasion will not be construed as a waiver of that provision on any other occasion, nor to affect the validity of that or any other provision of this Agreement. No waiver of any breach of this Agreement will be deemed to be a waiver of any other or subsequent breach.
- 10.7 No Third Party Beneficiaries. This Agreement is for the sole benefit of the Parties and their permitted assigns, and nothing in this Agreement expressed or implied will give or be construed to give to any Person, other than the Parties and their permitted assigns, any legal or equitable rights under this Agreement, except as otherwise expressly provided in Article 9.
 - 10.8 Good Faith. The Parties covenant to act in good faith and in a commercially reasonable manner under this Agreement.
- **10.9** Governing Law. This Agreement will be construed in accordance with, and the legal relations among the Parties will be governed by, the Laws of the State of Maine as applicable to agreements executed and fully performed in the State of Maine.
- **10.10** Consent to Jurisdiction and Venue. Each Party consents to the exclusive jurisdiction and venue of any Federal Court in Maine or any state court located in Portland, Maine for adjudication of any suit, claim, action or other proceeding at law or in equity relating to this Agreement, or to any transaction contemplated by this Agreement. Each Party accepts, generally and unconditionally, the exclusive jurisdiction and venue of those courts and waives any objection as to venue, and any defense of *forum non conveniens*.
- 10.11 <u>Disclosures and Announcements</u>. Public announcements and releases of information concerning the transactions contemplated in this Agreement by Buyer or either Selling Party, including regulatory communications, will be subject to the cooperation and approval of both Parties in all essential respects.
- 10.12 Non-Recourse. No past, present or future director, officer, employee, incorporator, member, partner, stockholder, affiliate, agent, attorney or other Representative of either Selling Party or their Affiliates will have any liability for any obligations, liabilities or other Liabilities of either Selling Party under this Agreement or the Ancillary Agreements, of or for any claim based on, in respect of, or by reason of, the transactions contemplated by this Agreement or the Ancillary Agreements. No past, present or future director, officer, employee, incorporator, member, partner, stockholder, affiliate, agent, attorney or other Representative of Buyer or its Affiliates will have any liability for any obligations, liabilities or other Liabilities of Buyer under this Agreement or the Ancillary Agreements, of or for any claim based on, in respect of, or by reason of, the transactions contemplated by this Agreement or the Ancillary Agreements.

10.13 Exhibits. The various forms attached as Exhibits are believed by the Parties to be substantially completed and will be modified in writing by the Parties, acting reasonably and in good faith, as needed to reflect additional facts, circumstances and information discovered or developed after the Execution Date and prior to the Closing. Those Exhibits that are not attached to this Agreement on the Execution Date will be negotiated and completed by the Parties for attachment to this Agreement at or prior to the Closing.

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed by their respective duly authorized officers or representatives, as of the day and year first above written.

REENERGY RUMFORD LLC		RUMFORD COGENERATION COMPANY LIMITED PARTNERSHIP			
		By: Rumford	Cogeneration, Inc.		
By:	/s/ William H. Ralston				
Print Name:	William H. Ralston	Ву:	/s/ George F. Martin		
Title:	Chief Risk Officer	Print Name:	George F. Martin		
		Title:	President and Chief Executive Officer		
		RUMFORD PA	PER COMPANY		
		Ву:	/s/ George F. Martin		
		Print Name:	George F. Martin		
		Title:	President and Chief Executive Officer		

52

LIST OF EXHIBITS

1.1(A)	Form of Assignment and Assumption Agreement****	1.1(S)	Summary Terms for Shared Services Agreement ***
1.1(B)	Form of Bill of Sale***	1.1(T)	Summary Terms for Undivided Ownership Agreement ***
1.1(C)	Buyer Required Regulatory Approvals	2.1(1)	Real Property *
1.1(D)	Summary Terms for Energy Supply Agreement ***	2.1(2)	Personal Property **
1.1(E)	Summary Terms for Fuel Supply Agreement ***	2.1(3)(A)	Purchased Inventory **
1.1(F)	Indemnifiable Environmental Conditions	2.1(3)(B)	Excluded Inventory **
1.1(G)	Summary Terms of the Joint Operating Committee Agreement***	2.1(4)	Assumed Contracts
1.1(H)	Summary Terms for Joint Use Agreement ***	2.1(5)	Transferable Permits
1.1(I)	Summary Terms for Mortgage ***	2.1(6)	Transferable Environmental Permits
1.1(J)	Form of NewPage Guaranty****	2.1(8)	Undivided Ownership Assets
1.1(K)	Non-Solicitation Agreement	2.1(13)	Other Purchased Assets
1.1(L)	Permitted Liens*	2.2	Other Excluded Assets
1.1(M)	Form of Quitclaim Deed*	2.6.1	Summary Terms for Secured Note ***
1.1(N)	Summary Terms for Reciprocal Easement Agreement***	2.8	Allocation of the Purchase Price *
1.1(O)	Summary Terms for Recovery Boiler Agreement ***	5.10	Summary Terms for Interconnection Assignment Agreement *
1.1(P)	Form of ReEnergy Guaranty****	5.14.2(A)	Real Property Subject to Restricted Activity
1.1(R)	Seller Required Regulatory Approvals	5.14.2(B)	Real Property Subject to Restricted Activity
		5.15	RPC Separation Description **
		9.2	Title Policy Endorsements

^{*} To be provided after the Execution Date and at or prior to Closing.

^{**} To be provided at the Execution Date and updated at Closing.

^{***} Term Sheet to be provided at the Execution Date and developed into Ancillary Agreements to be executed at Closing.

^{****} To be negotiated by the Parties after the Execution Date and executed at Closing.

LIST OF SCHEDULES

3.3.2	Seller No Conflict
3.4	Seller Third Party Consents
3.5	Financial Records
3.7(A)	Exceptions to personal property title
3.7(B)	Personal property leases
3.8.1	Owned Real Property
3.8.3	Real Property Compliance with Laws and Permits
3.9(A)	Material Contracts
3.9(B)	Material Contracts Violations
3.10	Ordinary Course of Business Exceptions
3.11.2	Violations of Environmental Law
3.11.3	Exceptions to Environmental Permits
3.11.4	Environmental Claims
3.11.5	PRP/Releases
3.11.6	Hazardous Materials
3.12	Compliance with Laws
3.12.2	Permits and Compliance with Permits
3.13	Litigation
3.15.1	Exceptions to Intellectual Property Rights
3.15.2	Owned Intellectual Property
3.15.3	Licensed Intellectual Property
3.16	Insurance Policies
3.17	Labor Disturbances
4.3.2	No Conflict
4.4	Third Party Consents
4.9	Buyer Materials Adverse Effects

Buyer's Proposed Improvements

2.6.1

5.14.2.2

Inventory Adjustment

A copy of the Exhibits and Schedules will be furnished supplementally to the Commission upon request.

Rule 15d-14(d) Certification in Accordance with Section 302 of the Sarbanes-Oxley Act of 2002

I, George F. Martin, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of NewPage Holding Corporation and NewPage Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrants as of, and for, the periods presented in this report;
- 4. The registrants' other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrants and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrants, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrants' disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrants' internal control over financial reporting that occurred during the registrants' most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrants' internal control over financial reporting; and
- 5. The registrants' other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrants' auditors and the audit committee of the registrants' board of directors:
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrants' ability to record, process, summarize and report financial information; and

b)	Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrants'
	internal control over financial reporting.

Date: May 12, 2011

By: /s/ George F. Martin

George F. Martin President and Chief Executive Officer (principal executive officer)

Rule 15d-14(d) Certification in Accordance with Section 302 of the Sarbanes-Oxley Act of 2002

I, Ronald J. Arling, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of NewPage Holding Corporation and NewPage Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrants as of, and for, the periods presented in this report;
- 4. The registrants' other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrants and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrants, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrants' disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrants' internal control over financial reporting that occurred during the registrants' most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrants' internal control over financial reporting; and
- 5. The registrants' other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrants' auditors and the audit committee of the registrants' board of directors:
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrants' ability to record, process, summarize and report financial information; and

b)	Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrants'
	internal control over financial reporting.

Date: May 12, 2011

By: /s/ Ronald J. Arling

Ronald J. Arling Controller and Chief Accounting Officer (principal accounting officer)

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of NewPage Holding Corporation and NewPage Corporation (the "Companies") on Form 10-Q for the period ending March 31, 2011 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, George F. Martin, Chief Executive Officer of the Companies, certify, pursuant to 18 U.S.C. §1350, as adopted pursuant to §906 of the Sarbanes-Oxley Act of 2002, that to my knowledge:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial position and results of operations of the Companies.

/s/ George F. Martin

George F. Martin Chief Executive Officer May 12, 2011

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of NewPage Holding Corporation and NewPage Corporation (the "Companies") on Form 10-Q for the period ending March 31, 2011 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Ronald J. Arling, Chief Accounting Officer of the Companies, certify, pursuant to 18 U.S.C. §1350, as adopted pursuant to §906 of the Sarbanes-Oxley Act of 2002, that to my knowledge:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial position and results of operations of the Companies.

/s/ Ronald J. Arling

Ronald J. Arling Chief Accounting Officer May 12, 2011